RISK OF DIABETES MELLITUS ASSOCIATED WITH USE OF ORAL GLUCOCORTICOIDS IN PATIENTS WITH POLYMYALGIA RHEUMATICA AND GIANT CELL ARTERITIS

Lana Lai Yin Hui

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

> The University of Leeds Leeds Institute of Rheumatic & Musculoskeletal Medicine (LIRMM)

> > MAY 2020

"The candidate confirms that the work submitted is his/her own, except where work which has formed part of jointly-authored publications has been included. The contribution of the candidate and the other authors to this work has been explicitly indicated below. The candidate confirms that appropriate credit has been given within the thesis where reference has been made to the work of others."

Published Thesis Work

 Lana Lai YH, Emma H, Robert W, Sarah M. Association between glucocorticoid therapy and incidence of diabetes mellitus in polymyalgia rheumatica and giant cell arteritis: A systematic review and meta-analysis. Lana Yin Hui Lai, Emma Harris, Robert M West, Sarah Louise Mackie. RMD Open 2018;0:e000521

Thesis chapters: Systematic Review / Meta-Analysis (Chapter 4)

My contribution: Study conception, literature search and data collection, statistical analysis and drafting the manuscript

Contribution of other authors: SM and RW helped with the study conception, EH was a co-reviewer for the literature search and helped with data collection, RW provided guidance on the statistical analysis and all three co-authors played a crucial role in the critical review of the manuscript for content.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

@2020 The University of Leeds and Lana Lai Yin Hui

The right of Lana Lai Yin Hui to be identified as Author of this work has been asserted by Lana Lai Yin Hui in accordance with the Copyright, Designs and Patents Act 1988.

ACKNOWLEDGEMENTS

I wish to thank my amazing supervisors, Dr. Sarah Mackie, Professor Robert West and Dr. Mar Pujades Rodriguez for their tremendous support, mentorship and guidance throughout this PhD journey. The clinical and epidemiological expertise provided by Dr. Mackie and Dr. Pujades-Rodriguez were vital to the development of this project, and Professor West played a crucial role in helping me to overcome the technical and statistical challenges throughout the process.

I would also like to thank Dr. Emma Harris for her help as a co-reviewer for the systematic review and meta-analysis relating to the association between glucocorticoid therapy and incidence of diabetes mellitus in polymyalgia rheumatica (PMR) and giant cell arteritis (GCA). Special thanks go to Dr. Sarah Alderson who has added invaluable insight on how PMR and GCA patients are managed in primary care.

I would also like to express my utmost gratitude to my husband Joel who has been with me through the highs and the lows, constantly spurring me on throughout this journey and patiently taking care of our one year-old son, Lucas when I needed time to work on the thesis. I would also like to thank my parents for their support and constant encouragement to achieve the best of my abilities.

Last but not least, I would like to thank the University of Leeds for awarding me the Leeds International Research Scholarship and for giving me the opportunity to pursue my aspirations as a researcher.

ABSTRACT

Background: Oral glucocorticoid (GC) use can induce diabetes mellitus (DM). Patients with polymyalgia rheumatica (PMR) and giant cell arteritis (GCA) are both treated with GCs and can develop DM. However, it is not known whether the DM risk differs with prescribing choices such as GC dose or duration.

Objectives:

- (1) To quantify DM risk associated with dose and duration of oral GC in PMR/GCA patients within the first two years of diagnosis, using Clinical Practice Research Datalink (CPRD).
- (2) To explore prescribing patterns of oral GCs for PMR/GCA patients in primary care.
- (3) To compare CPRD-derived prescription data with extracted data from primary care prescription data.

Methods:

- (1) The impact of GC dose and duration on DM risk was explored using the extended time-varying Cox model. The additional risk associated with the timing of GC exposure was determined using the rolling cumulative dose and weighted cumulative exposure models. The risk was compared between patients on high GC dose versus low GC dose.
- (2) The total monthly oral GC dose was calculated and compared with clinical guidelines.
- (3) The CPRD-derived prescription data and my extracted data were compared.

Results:

- (1) When compared patients on high dose versus low dose regimens, DM risk was increased by 1.4-2.5-fold within the most recent ten months for PMR patients; and 1.6-2.8-fold within the most recent five months for patients with GCA (with or without PMR).
- (2) PMR prescribing was broadly in line with guidelines, but starting doses prescribed for GCA in primary care were lower than guidelines.
- (3) Data from the CPRD-derived prescriptions were similar to my extracted data.

Conclusion: Higher GC dose was associated with an elevated risk of new DM in PMR/GCA patients within the initial two years of diagnosis. Treatment strategies to reduce cumulative GC dose should be investigated.

TABLE OF CONTENTS

ACKNOWLEDGEMENTS	iii
ABSTRACT	iv
TABLE OF CONTENTS	v
LIST OF TABLES	. viii
LIST OF FIGURES	x
ABBREVIATIONS	xii
CHAPTER 1: INTRODUCTION	1
1.1. Rationale of the PhD Project	1
1.2. Study Aims and Objectives	2
1.3. Outline of Thesis	3
CHAPTER 2: POLYMYALGIA RHEUMATICA AND GIANT CELL ARTERITIS	5
2.1. Overview	5
 2.2. Polymyalgia Rheumatica (PMR)	6 6 7 10 10
 2.3. Giant Cell Arteritis (GCA) 2.3.1. History 2.3.2. Epidemiology 2.3.3. Diagnosis and Classification 2.3.4. Clinical Manifestations 2.3.5. Treatment 	13 13 13 14 17 17
CHAPTER 3: GLUCOCORTICOIDS	23
3.1. Overview	23
3.2. Pharmacokinetics	24
 3.3. Mechanism of Action 3.3.1. Genomic Mechanisms 3.3.2. Non-genomic Mechanisms 3.3.3. GC Effects on PMR and GCA 	25 25 26 27
3.4. Definition of Conventional Terms for Glucocorticoid Doses	29
3.5. Glucocorticoid-Induced Adverse Events (AEs)	31 32

CHAPTER 4: ASSOCIATION BETWEEN GLUCOCORTICOID THERAPY AND INCIDENCE OF DIABETES MELLITUS IN POLYMYALGIA RHEUMATICA AND 4.2. Methods and Materials40 4.2.6. Statistical Analysis45 5.3.4. Index of Multiple Deprivation (IMD)57 5.5.4. Factors of Interest61 5.6.2. Descriptive Analysis on Prescription Database and Prescribing Patterns ... 68 5.6.5. Rolling Cumulative Dose Model71 5.6.6. Weighted Cumulative Exposure (WCE) Model71 6.2. Patient Selection Process73 6.3. Baseline Characteristics75 6.4. Prescription Database76 6.4.1. Types of Oral GC Prescribed77 6.4.2. Missing Data for Key Variables77 6.4.5. Number of Tablets Per Day and Daily Dosing81

 6.5. GC Prescribing Patterns. 6.5.1. Total Monthly Oral GC Dose 6.5.2. High Dose (75th Percentile) Versus Low Dose (25th Percentile) Regimens . 6.5.3. Average Total Monthly Oral GC Dose 6.5.4. GC Prescribing Patterns in Primary Care vs. EULAR Guidelines 	81 82 84 85 87
6.6. Pattern of Distribution for DM Over Time	90
6.7. Survival Analysis 6.7.1. Kaplan Meier. 6.7.2. Extended Time-varying Cox Model. 6.7.3. Rolling Cumulative Dose Model 1 6.7.4. Weighted Cumulative Exposure (WCE) model 1 6.7.5. Comparison of Hazard Ratios Across the Extended Time-Varying Cox, Rolling Cumulative Dose and Weighted Cumulative Dose Analyses 1	94 97 00 03
CHAPTER 7: DISCUSSION AND CONCLUSION	00
71 Overview	00
	00
 7.2. Evaluation of Study Findings	09 09 14 17 21
7.3. Strengths and Limitations1	22
7.3.1. Strengths	22
7.0.2. Elimitations and Fotential Diases	20
7.4. Neconimentations	24
7.5. Future work1	34
7.6. Conclusions1	37
REFERENCES1	38
APPENDICES1	56
APPENDIX 1: 'MeSH' Search Terms1	56
APPENDIX 2: MOOSE Guidelines1	57
APPENDIX 3: ISAC Protocol1	58
APPENDIX 4: R Codes 1	59
Appendix 4.1. Cohort	59
Appendix 4.2. Covariates 1	62
Appendix 4.3. Prescription Database1	64
Appendix 4.4. Preparing Loops to Calculate Daily Doses	69
Appendix 4.5. Format Data For Survival Analysis	11
Appendix 4.0. Rapian-Ivielet Survival Analysis	1 Z
Appendix 4.7. Extended Time-varying Cox for Cumulative Dose	1 Z 72
Appendix 4.9. Weighted Cumulative Dose	74

LIST OF TABLES

Table 2.1: Diagnostic and classification criteria for PMR
Table 2.2: 1990 criteria for classification of giant-cell (temporal) arteritis16
Table 3.1: Literature on glucocorticoid-induced hyperglycemia and diabetes mellitus
selected by the EULAR task force35
Table 4.1: The Cochrane collaboration's tool for assessing risk of bias in RCTs43
Table 4.2: The Cochrane tool for assessing risk of bias in cohort study
Table 4.3: Studies included in the meta-analysis46
Table 4.4: Summary characteristics of individuals with PMR and/or GCA47
Table 4.5: Results from final multivariable model predicting risk of GC-induced DM49
Table 4.6: Risk of bias of randomized controlled trials50
Table 4.7: Risk of bias of observational studies50
Table 5.1: Description of CPRD data files obtained59
Table 5.2: Prednisolone-equivalent doses (PED) anti-inflammatory doses
Table 5.3: Frequencies of oral GC administration 62
Table 6.1: Characteristics of newly diagnosed patients, stratified by disease status75
Table 6.2: Types of oral GCs prescribed in primary care 77
Table 6.3: Comparison of missing data for key variables in the prescription database 78
Table 6.4: Distribution of dosing frequencies in the prescription database79
Table 6.5: Frequency and proportion of incident DM in PMR patients over 24 months 91
Table 6.6: Frequency and proportion of incident DM in patients with GCA
(with or without PMR) over 24 months93
Table 6.7: Extended time-varying Cox regression model for DM in PMR patients97
Table 6.8: Extended time-varying Cox regression model for DM in patients with GCA
(with or without PMR)98
Table 6.9: Hazard ratios (HRs) for DM in patients with high GC dose (75 th percentile)
versus low GC dose (25 th percentile) over 24 months using the extended
time-varying Cox analysis99
Table 6.10: Concordance index of rolling cumulative dose based on months following
the diagnosis of PMR and/or GCA100
Table 6.11: Hazard ratios (HRs) for DM in patients with high GC dose versus low GC
dose over 24 months using the rolling cumulative dose analysis

Table 6.12: Comparison of goodness of fit test across WCE models over 24 months
Table 6.13: Hazard ratios (HRs) for DM in patients with high GC dose versus low GC
dose over 24 months using the weighted cumulative dose analysis 104

LIST OF FIGURES

Figure 2.1:	Incidence of PMR for populations above 50 years of age7
Figure 2.2:	The 2010 BSR GC tapering regimen for treatment of PMR11
Figure 2.3:	The 2015 EULAR GC tapering regimen for treatment of PMR12
Figure 2.4:	Incidence of GCA for populations above 50 years of age14
Figure 2.5:	An example of GC tapering regimen for the treatment of GCA based on the
	2010 BSR guideline
Figure 2.6:	The 2018 EULAR algorithm for treatment of GCA19
Figure 3.1.	Genomic and non-genomic effects on the dose-response relationship of GC
Figure 3.2.	Mechanisms of glucocorticoid-induced hyperglycemia in the liver, skeletal
	muscle, adipose tissue, pancreatic beta cells and small intestine34
Figure 3.3.	The level of harm of long-term GC therapy in rheumatic diseases
Figure 4.1:	Flowchart of the selection process of studies41
Figure 4.2:	Proportion of PMR and GCA patients who developed new-onset DM with
	95% confidence intervals
Figure 6.1:	Flowchart of the patient selection process74
Figure 6.2:	Total monthly oral GC in PMR patients over 24 months
Figure 6.3:	Total monthly oral GC in GCA (with or without PMR) patients over 24
	months83
Figure 6.4:	Comparison of high GC dose (75 th percentile) versus low GC dose (25 th
	percentile) among PMR and/or GCA patients within the first two years of
	diagnosis
Figure 6.5:	Average total monthly oral GC prescribing pattern over 24 months for PMR
	patients, stratified by DM status85
Figure 6.6:	Average total monthly oral GC prescribing pattern over 24 months for
	patients with GCA (with or without PMR), stratified by DM status
Figure 6.7:	Comparison of monthly oral GC for PMR patients in primary care with
	EULAR guideline within the first year of diagnosis87
Figure 6.8:	Comparison of monthly oral GC for GCA (with or without PMR) patients in
	primary care with EULAR guideline within the first year of diagnosis88
Figure 6.9:	Percentage of first oral GC prescription without a quantifiable dosing
	instruction among patients with GCA (with or without $\ensuremath{PMR}\xspace)$ in primary care

Figure 6.10: Percentage of incident DM in PMR patients over 24 months90
Figure 6.11: Percentage of incident DM in GCA (with or without PMR) patients over 24
months92
Figure 6.12: Kaplan-Meier survival curves for time to DM diagnosis, stratified by
disease groups94
Figure 6.13: Kaplan-Meier survival curves for time to DM diagnosis, stratified by gender
Figure 6.14: Kaplan-Meier survival curves for time to DM diagnosis, stratified by age
group in quartiles96
Figure 6.15: Weight functions for best-fitting WCE models of oral GC cumulative dose
Figure 6.16: Comparison of hazard ratios across the three different statistical methods
in PMR patients on high GC dose versus low GC dose
Figure 6.17: Comparison of hazard ratios across the three different statistical methods
in patients with GCA (with or without PMR) on high GC dose versus low
GC dose107
Figure 7.1: Management and treatment algorithm for GC-induced DM in patients
without a pre-existing diagnosis of DM, adapted from JBDS-IP guideline 133

xi

ABBREVIATIONS

11β-HSD1	11β-hydroxysteroid dehydrogenase 1	
95% CI	95% confidence interval	
ACR	American College of Rheumatology	
ACTH	Adrenocorticotropic hormone	
AE	Adverse event	
BAFF	B-cell activating factor	
BD	Twice a day	
BIC	Bayesian Information Criterion	
BID	Twice a day	
BMI	Body mass index	
BNF	British National Formulary	
BSR	British Society for Rheumatology	
CBG	Capillary blood glucose	
CD4+	Cluster of differentiation 4+	
CDC	Centers for Disease Control and Prevention	
cGR	Cytosolic glucocorticoid receptors	
CHCC	Chapel Hill Consensus Conference	
CINAHL	Nursing and Allied Health Literature	
CPRD	Clinical Practice Research Datalink	
CRP	C-reactive protein	
CVD	Cardiovascular disease	
DM	Diabetes mellitus	
eGFR	Estimated glomerular filtration rate	
EGP	Endogenous glucose production	
EMA	European Medicines Agency	
EOD	Every other day	
ESR	Erythrocyte sedimentation rate	
EULAR	European League Against Rheumatism	
FDA	US Food and Drug Administration	
G6P	Glucose-6-phosphatase	
GC	Glucocorticoid	
GCA	Giant cell arteritis	
GLP-1	Glucagon-like peptide 1	
GIP	Glucose dependent insulinotropic polypeptide	
GP	General practitioner	
GDPR	General Data Protection Regulation	

GRE	Glucocorticoid response element		
HbA1c	Hemoglobin A1c		
HES	Hospital Episode Statistics		
HPA	Hypothalamic pituitary adrenal		
HR	Hazard ratio		
IA	Intra-articular		
ICD	International Classification of Diseases		
IL	Interleukin		
IM	Intramuscular		
IMD	Index of Multiple Deprivation		
IQR	Inter-quartile range		
ISAC	Independent Scientific Advisory Committee		
JBDS-IP	Joint British Diabetes Societies for Inpatient Care		
LSOA	Lower layer super output areas		
Mane	Morning		
MC	Mineralocorticoids		
MetS	Metabolic syndrome		
MHRA	Medicines and Healthcare products Regulatory Agency		
MOOSE	Meta-analysis of Observational Studies in Epidemiology		
MRI	Magnetic resonance imaging		
mRNA	Messenger ribonucleic acid		
MTX	Methotrexate		
NHS	National Health Service		
NIHR	National Institute for Health Research		
NLP	Natural language processing		
Noct	Night		
OD	Every day		
OM	Every morning		
ON	Every night		
ONS	Office for National Statistics		
OR	Odds ratio		
PED	Prednisolone-equivalent dose		
PEPCK	Phosphoenylpyruvate carboxykinase		
PET	Positron-emission tomography		
PICO	Population, Intervention, Comparator, Outcome		
PMR	Polymyalgia rheumatica		
PRN	When needed		
QDS	Four times a day		

QoL	Quality of life	
RA	Rheumatoid arthritis	
RCT	Randomized controlled trial	
RR	Relative risk	
SD	Standard deviation	
SES	Socioeconomic status	
SLE	Systemic lupus erythematosus	
Stat	Immediately	
TAB	Temporal artery biopsy	
TCZ	Tocilizumab	
TDS	Three times a day	
THIN	The Health Improvement Network	
TID	Three times a day	
TNF	Tumor necrosis factor	
Treg	T regulatory	
Th1	Type 1 helper T cell	
Th17	Type 17 helper T cell	
UK	United Kingdom	
USA	United States of America	
UTS	Up-to-standard	
WCE	Weighted cumulative exposure	
WHO	World Health Organization	

Four times a day

QID

CHAPTER 1: INTRODUCTION

1.1. Rationale of the PhD Project

Polymyalgia rheumatica (PMR) is a glucocorticoid-responsive condition causing bilateral shoulder and pelvic girdle pain and stiffness in older people [1, 2]. PMR is one of the commonest reasons for long-term glucocorticoid (GC) treatment in the community [3-5]. PMR is closely associated with giant cell arteritis (GCA), a common large vessel vasculitis affecting the over-50s, characterized by inflammation of the aorta, temporal arteries and other arteries supplying the arms, head and neck. Approximately one in five patients with PMR have GCA, and between 40-60% of GCA patients report PMR symptoms [6]. GCA is also treated with long-term GCs, but at higher doses [7]. The overall age-adjusted United Kingdom (UK) incidence rate of PMR is 8.4/10,000 person-years, and 1.1/10,000 person-years for GCA [8]. These incidence rates were calculated from Clinical Practice Research Datalink (CPRD), a primary care database consisting of longitudinal electronic medical records of patients registered with contributing general practices in the UK [9]. Both PMR and GCA have a female: male ratio of 2-3:1 and similar age distribution, with incidence peaking in the seventh and eighth decade of life [10]. GCs are usually prescribed at a dose sufficient to control initial symptoms, and are then tapered down over months with a typical average therapy duration of two years [11, 12]. Several studies have reported that the majority of patients with PMR and GCA experience GC-associated adverse effects [13, 14], with one of the most worrisome of these for both patients and rheumatologists being diabetes mellitus (DM) [15]. GC-induced DM is an abnormal increase in blood glucose concentration associated with GC use in patients with or without a previous history of DM [16]. Longer duration of treatment, higher GC dose and higher GC potency were reported to be associated with an increased risk of GC-induced DM [16-18]. The increase in blood glucose levels in response to GC therapy may cause an individual to become symptomatic with polyuria, weight loss, thirst and fatigue over time [19]. There is also a risk of progression to hyperosmolar hyperglycaemic state or diabetic ketoacidosis [20]. In the longer term, patients with GC-induced DM are at risk of microvascular (i.e. nephropathy, neuropathy and retinopathy) and macrovascular (i.e. cardiovascular) complications in the same way as other patients with DM. Novel adjunctive biologic therapies for GCA have been tested in clinical trials, including tocilizumab, but these biologic therapies are expensive and payers need to understand the health risks (including DM) associated with GC therapy in order to balance these against the additional costs to the payer of adding in a further drug with its associated costs. Despite the importance of this for patients, physicians and health authorities,

there is currently minimal information on what the absolute risk of DM is in these patient groups. This PhD thesis describes my work to quantify the risk of DM associated with oral GC use in PMR and/or GCA patients using the CPRD database.

1.2. Study Aims and Objectives

The overall aim was to determine what is the risk of developing new-onset DM in patients starting GC treatment for PMR and/or GCA. Within that overall aim, I systematically reviewed published literature, and also used a UK primary care research dataset to determine how that risk may vary with dose of GCs. I also wanted to investigate the time course of risk – how long after the GC therapy did the elevated risk appear to persist.

The primary objective was to model the relationship between DM risk and oral GC dose over time in newly diagnosed PMR and/or GCA patients within the initial two years of PMR/GCA diagnosis using CPRD.

The second objective was to explore the prescribing patterns of oral GCs for PMR patients, as well as GCA (with or without PMR) patients in real-world primary care settings, for which the observed prescribing patterns were compared to the prescribing recommendations in the European League Against Rheumatism (EULAR) clinical practice guidelines [21, 22].

The third objective was to compare CPRD-derived prescription data with the data that I extracted from primary care prescription data collected directly from registered primary care general practitioners (GPs) and provided by CPRD.

1.3. Outline of Thesis

The purpose of this PhD project is to develop a better understanding of the risk of DM associated with oral GC use in patients with newly diagnosed PMR and/or GCA in primary care. The risk of DM is explored and described through in-depth analysis of CPRD. CPRD is a primary care electronic medical records database consisting of clinical and prescription data linked to the Hospital Episode Statistics (HES), Office for National Statistics (ONS) mortality data and Index of multiple deprivation (IMD).

Chapter 2 provides background on the specific diseases of interest, PMR and GCA. This chapter will cover aspects such as the history, epidemiology, pathophysiology, diagnosis and classification, clinical manifestations, as well as treatment options.

Chapter 3 reviews different aspects of prednisolone, the most commonly used GC in the UK for the treatment of PMR and/or GCA. This chapter reviews the pharmacokinetic properties of prednisolone, mechanism of action, classification of conventional dosing and its relationship with both therapeutic and detrimental effects. Common GC-induced adverse events including DM will also be discussed in this chapter.

Chapter 4 is a systematic review of all published articles or conference abstracts that reported DM following exposure to oral GC therapy in patients with GCA and PMR. This chapter provides an estimate of the risk of DM in these patient populations using the random-effects model. It also provides the rationale for the importance of using large electronic health records data to better quantify the risk of DM associated with oral GC use.

Chapter 5 provides an overview of the research methods, including information regarding the prescription database and details on how the study population was constructed based on the inclusion and exclusion criteria for this study. This chapter will also address ethical considerations and use of different data sources in this study. The rationale behind use of the selected statistical methods in this project, which include the Kaplan-Meier survival analysis, extended time-varying Cox model, rolling cumulative dose model and the weighted cumulative exposure (WCE) model will also be discussed in detail.

Chapter 6 presents the baseline characteristics of PMR and/or GCA patients, stratified by disease status. Given the close overlap between PMR and GCA, there are some patients diagnosed with both PMR and GCA. Patients with both diseases are analyzed together with the GCA-only patients, as they are more likely to be receiving a similar treatment regime as those with GCA only. There are three major sections in this chapter. Firstly, the prescription data that I extracted from primary care data collected directly from registered primary care GPs across the UK will be compared with CPRD-derived prescription data managed by the CPRD team to assess agreement and discrepancies between the two datasets. A detailed analysis of the prescription database from my data extraction is also presented, covering aspects such as types of oral GC prescribed, distribution of GC dosing frequencies and duration of GC prescriptions. Secondly, the GC prescribing patterns in primary care are described and compared with EULAR guidelines. Thirdly, the risk of DM associated with oral GC dose within the initial two years of PMR/GCA diagnosis is determined using statistical methods described in the preceding chapter.

Chapter 7 discusses the findings, focusing on how the dose, duration and timing of GC treatment affects the risk of developing DM in patients with PMR and/or GCA, and how the results compare with other published literature. I also discuss the GC prescribing pattern in primary care, followed by some of the challenges associated with analyzing primary care prescription databases such as CPRD. The next sub-section includes a discussion on the strengths and limitations of the study, along with appropriate measures taken to minimize potential biases. I also discuss how my findings have implications for screening and management of DM in PMR and/or GCA patients. At the end of this chapter, the conclusions will bring together the main points from the preceding chapters to give an overview of the importance of this work, as well as outline future work.

CHAPTER 2: POLYMYALGIA RHEUMATICA AND GIANT CELL ARTERITIS

2.1. Overview

The content of this chapter will focus on the two central diseases of my thesis: polymyalgia rheumatica (PMR) and giant cell arteritis (GCA). PMR and GCA are two different but frequently overlapping inflammatory disorders that often occur in patients aged 50 years and older [23]. PMR typically presents acutely with morning stiffness and bilateral shoulder pain or pelvic girdle arching [24]. GCA typically presents with persistent localised headache, weight loss, fatigue, low-grade fever, and in more severe cases, vision loss [22]. Approximately 16-21% of PMR patients are reported to have GCA [25, 26], while 40-60% of patients diagnosed with GCA also have PMR [27, 6, 28]. While the current understanding of GCA is that it is a granulomatous autoimmune vasculitis affecting larger arteries and aorta, the pathology of PMR and its overlap with GCA is unclear because of a paucity of histological data [29]. Glucocorticoids (GCs) are the cornerstone of treatment for PMR and GCA. They are highly effective and patients usually see an improvement of symptoms within 24-48 hours upon initiation of therapy [30, 31]. Oral GC therapy is known to be associated with a risk of diabetes mellitus (DM), but the actual risk in PMR and GCA is unclear. Because of the strong overlap between the two diseases, I will be looking at the risk of DM in both PMR and GCA patients. PMR and GCA are both treated with GCs, but the starting dose is much higher in GCA as this higher dose is required to control GCA symptoms whereas lower dose is sufficient in PMR. Some patients have both PMR and GCA, they are usually treated with the higher starting dose, and so because there is a strong rationale to suspect that DM risk is related to the treatment rather than the underlying disease of PMR or GCA, in studies of GC-related complications, GCA patients are usually grouped together regardless of whether they also have a diagnosis of PMR. In addition, grouping them together will also add to the robustness of statistical analyses, as they are relatively rare and infrequently encountered in primary care, with a full-time GP possibly only seeing one new case every 1-2 years [32]. In this chapter, specific areas of discussion will include a brief history, epidemiology, diagnosis and classification, clinical manifestations and treatment of PMR and GCA respectively.

2.2. Polymyalgia Rheumatica (PMR)

2.2.1. History

Polymyalgia rheumatica (PMR) was first described in 1888 as "acute senile rheumatic gout" in five elderly patients [33], and until 1950s, there were case reports describing the disease under various names, such as "periarticular fibrositis", "peri-extra articular rheumatism", "periarthrosis humeroscapularis" and "pseudopolyarthrite rhizomielique". It was only in 1957 when the British physician Barber established the term "PMR" [34]. The concept of PMR was further developed by Healey, who described variant forms of PMR as "benign synovitis" [35].

Despite the establishment of the term "PMR", the pathogenesis of the disease remains unknown, and there is considerable uncertainty regarding diagnosis and outcomes of the disease. Initial features may be variable and difficult to distinguish as approximately one in five patients with PMR have GCA [25, 26], and approximately one in two GCA patients have polymyalgic symptoms [27, 6, 28].

2.2.2. Epidemiology

PMR is almost exclusively a disease that occurs in individuals over 50 years of age, with a prevalence that increases progressively with advancing age [8]. The peak incidence of PMR is reported to be between 70 and 80 years of age [26]. Women have approximately twice the risk of developing PMR as compared to men [36, 37, 8]. Two large, observational cohort studies in Denmark and the United States of America (US) and one case control study in Norway have reported that survival in PMR patients is similar to those in the general population [36, 38, 39]. The incidence of PMR varies according to geographical regions, with countries in the northern latitudes having much higher rates. A summary of incidence rates across various countries is shown in Figure 2.1.





Note: *USA – Population was of Scandinavian descent

2.2.3. Diagnosis and Classification

The diagnosis for PMR is a challenge as there are no established diagnostic criteria for PMR. Current guidelines recommend that the diagnosis of PMR should start with the evaluation of core inclusion and exclusion criteria, followed by assessment of the response to a standardized dose of GC [24, 21]. Core inclusion criteria include the following [24]:

- Bilateral shoulder and/or pelvic girdle pain
- Morning stiffness > 45 minutes
- Abrupt onset
- Age > 50 years
- Duration > 2 weeks
- Acute phase response (raised erythrocyte sedimentation rate (ESR)/ C-reactive protein (CRP))

Patients who present with the above-mentioned core criteria should also be screened and the following diseases - active cancer, active infection and active GCA should be excluded [24]. Next, patients should be started on oral prednisolone (15-20mg daily), and the clinical response assessed within 1 week. Patients are expected to have at least 70% global improvement by then, with elevated ESR and CRP levels normalizing within 3-4 weeks [24].

In 1979, Bird proposed the first classification criteria for PMR [45]. Two years later, Jones developed another set of classification criteria for his study [46]. This was followed by Chuang [47], who redefined a new clinical criterion for his study, which was further refined in 1984, when Healey [48] added the evaluation of steroid response to Chuang's existing criteria. A summary of the development of classification criteria over time is shown in Table 2.1.

Table 2.1: Diagnostic and classification criteria for PMR

Characteristics	Bird, 1979 [45]	Jones, 1981 [46]	Chuang et al, 1982 [47]	Healey, 1984 [48]	Dasgupta, 2012 (EULAR/ACR) [2]
Age onset (years)	> 65	> 50	> 50	> 50	*> 50
Onset duration	< 2 weeks	>/= 2 months	-	-	
Signs and symptoms	 Shoulder pain and/or stiffness bilaterally Depression and/or loss of weight Upper arms tenderness bilaterally 	 Shoulder or hip girdle pain Morning stiffness 	 Bilateral aching/stiffness ≥ 1 month involving ≥ 2 areas: neck or torso, shoulders or upper arms, hips or thighs 	 Pain in the neck, shoulders or pelvic girdle Marked morning stiffness 	 *Bilateral shoulder or pelvic girdle arching, or both Hip pain or limited range of motion (1 point)
Duration of morning stiffness (minutes)	> 60	-	> 30	> 60	> 45 minutes (2 points)
ESR (mm/hr)	> 40	> 30	> 40	> 40	*Elevated
CRP (μg/ml)	-	> 6	-	-	*Elevated
Response to steroids	NA	-	-	Rapid response to prednisolone (≤20mg/day)	-
Other criteria	-	 Absence of rheumatoid arthritis Absence of muscle disease 	 Exclusion of other causes (e.g. active rheumatoid arthritis, lupus erythematous or polymyositis, chronic infection, multiple myeloma and Parkinson's disease) 	 Absence of rheumatoid factor or antinuclear antibody 	 Absence of rheumatoid factor and/or anti- citrullinated protein antibodies (2 points) Absence of peripheral joint pain (1 point)
Number of criteria needed	≥ 3	All	All	All	Point-based (>/= 4)

Note: * = Core criteria for PMR

As shown in Table 2.1, the current American College of Rheumatology/European League Against Rheumatism (ACR/EULAR) guideline recommend the use of a pointsbased scoring algorithm for the classification of PMR [2]. A score of \geq 4 is categorized as PMR, with 68% sensitivity and 78% specificity; and by adding ultrasound, a score of \geq 5 had 66% sensitivity and 81% specificity. An additional two points were allocated to the ultrasound-based algorithm – patients with at least one shoulder with subdeltoid bursitis and/or biceps tenosynovitis and/or glenohumeral synovitis and at least one hip with synovitis and/or trochanteric burtitis will be given one point. Patients who have both shoulders with subdeltoid bursitis, biceps tenosynovitis or glenohumeral synovitis will be given another point.

Although the ACR/EULAR guideline proposed a classification criteria for PMR, they should not be used as diagnostic criteria [2]. It is essential to understand that the proposed criteria was designed as a research tool to distinguish PMR from other conditions and were not intended for the diagnosis of individual patients [49].

2.2.4. Clinical Manifestations

PMR primarily affects the muscles and joints of the shoulder, neck and hip girdles, with prominent bilateral pain and morning stiffness [50], of which the shoulders are affected in 95% of cases [47]. These symptoms usually develop over weeks to months [51], and typically are worst in the early hours of the morning or upon waking [52]. Approximately half of patients diagnosed with PMR may have distal manifestations such as hand swelling with pitting edema, peripheral arthritis and carpal tunnel syndrome [50, 43, 53, 54]. Systemic symptoms such as low-grade fever, fatigue, malaise and weight loss have also been reported between 30-50% of patients [47].

2.2.5. Treatment

GCs are the treatment of choice for PMR. In the absence of GCA, urgent GC therapy is not indicated before the clinical evaluation is complete. Patients diagnosed with PMR are typically started on 15mg of prednisolone daily and gradually tapered down over time. Initial doses of less or equal to 7.5mg/day and initial doses of greater than 30mg/day are not recommended [21]. Figures 2.2 and 2.3 show the recommended tapering regimen by the current British Society for Rheumatology (BSR) and EULAR guidelines respectively [24, 21]. For relapse cases, there is a slight difference in the recommendations between the BSR and EULAR guidelines. The EULAR guideline suggests that oral GC may be increased to the pre-relapse dose and decreased gradually within 4-8 weeks to the dose at which the relapse occurred [21]. The BSR guideline recommends GC doses to be increased to previous higher dose, and for those with clinical features of GCA, they should be treated as GCA, usually with oral prednisolone 40-60mg daily [24]. Single intramuscular of injection of methylprednisolone 120mg can also be used [24]. Both guidelines agreed that oral methotrexate may be considered for patients on GC who are at high risk of relapse or for those who did not respond to GC treatment. Due to the heterogeneity of disease course, approach to treatment is often tailored individually depending on disease severity, co-morbidities, laboratory markers, adverse effects and patient preference. GCs are usually needed for 1-3 years [55, 24]; although some may require small doses of GCs beyond this. The 2015 ACR/EULAR guideline recommends administering GC therapies for the minimum effective duration. In terms of monitoring, the BSR guideline recommends laboratory monitoring of full blood count, ESR/CRP, urea, electrolytes and glucose levels every 3 months, in addition to monitoring in improvement of clinical symptoms. The EULAR guideline recommends individualized dose tapering and regular monitoring of PMR patients, but unlike the BSR guideline, there is no fixed schedule.



Figure 2.2: The 2010 BSR GC tapering regimen for treatment of PMR [24]



Figure 2.3: The 2015 EULAR GC tapering regimen for treatment of PMR [21]

2.3. Giant Cell Arteritis (GCA)

2.3.1. History

GCA was first described clinically in 1890 under the name "arteritis of the aged" and was later histologically characterized by Horton in 1932, which led to the name "Horton's disease" [56]. It was also known as "arteritis temporalis" as it commonly affects the superficial temporal arteries, as well as the ophthalmic, posterior ciliary and vertebral arteries [57], while intracranial arteries are rarely associated with the vasculitic process [6]. These names were, however superseded and renamed as "giant cell arteritis" at the 1994 Chapel Hill Consensus Conference (CHCC) [58]. The presence of temporal artery inflammation was neither a required nor a sufficient feature for diagnosing giant cell arteritis, as not all patients with GCA have involvement of the temporal artery. In addition, occasional occurrence of temporal artery involvement in other forms of systemic vasculitis such as Wegener's granulomatosis or polyarteritis nodosa also exists. In the 2012 revised version of the CHCC nomenclature, GCA was defined as a large-vessel vasculitis, affecting the aorta and its large arterial branches, with a predilection for the branches of the carotid and vertebral arteries.

2.3.2. Epidemiology

GCA is the most common type of primary systemic vasculitis affecting individuals above 50 years of age, with women being two to four times more commonly affected as compared to men [59-62]. GCA has been reported worldwide in various geographical locations among various ethnic groups. There is however, striking variation in its incidence in the different populations and regions of the world. GCA is more common in Northern European countries and among immigrants of North European descent [63-65]. Figure 2.4 summarizes the annual incidence rates among those above 50 years of age, of which the highest annual incidences were reported in Scandinavian countries, including Finland [66], Norway [67], Iceland [68], Sweden [69] and Denmark [38], followed by Mediterranean countries such as Spain [70], France [71] and Italy [44]. GCA was relatively rare among Arabs [72] and Asians [73], as depicted in Figure 2.4.



Figure 2.4: Incidence of GCA for populations above 50 years of age. Data were obtained from references [66-68, 74, 69, 38, 75, 70, 64, 76-78, 71, 44, 79, 72, 73, 80]. *Note:* *USA – *Population was of Scandinavian descent*

2.3.3. Diagnosis and Classification

As with PMR, the diagnosis of GCA remains a challenge, as no definitive tests for GCA exist. Ultimately, the diagnosis of GCA is made based on the clinician's experience and expertise, often based, but not confined to a combination of clinical symptoms, clinical findings, laboratory results and diagnostic imaging [81-84]. The latest BSR guideline recommends that high-dose GCs should be immediately started in patients of which GCA is strongly suspected, based on the clinician's judgment that GCA is a likely explanation for the patient's symptoms than any other condition [85]. Since GCA is a medical emergency, patients with suspected GCA should be referred urgently and be evaluated by a specialist ideally on the same working day, or at least within 3 working days [85].

Temporal artery biopsy (TAB) is the gold standard for the diagnosis of GCA. A specimen is usually taken from either one side or both left and right temporal arteries as an outpatient procedure under local anesthesia. A segment of artery (about 2.0-2.5cm long) is removed and perfusion is provided using collaterals [86]. A positive biopsy result is highly suggestive of GCA, however, it may also be seen in other forms of systemic vasculitis such as Wegener's granulomatosis, polyarteritis nodosa and rheumatoid vasculitis [87]. A recent meta-analysis reported a 77% estimated sensitivity of TAB in GCA patients [88].

Over the last years, non-invasive modalities have emerged as alternatives, which allow the arteries to be visualized noninvasively and examined for signs of inflammation. The color-coded duplex ultrasonography is able to detect the presence of a dark hypoechoic circumferential wall thickening around the lumen (also known as the "halo"), as well as the existence of stenosis and occlusions, but it requires technical skills and a high level of experience [57]. High-resolution contrast-enhanced magnetic resonance imaging (MRI) is also available to evaluate the possible inflammation of vessel walls. Other non-invasive modalities such as the positron-emission tomography (PET), which uses radioactive isotopes, have also shown to be effective for imaging inflammatory processes in GCA [86]. The use of all these imaging techniques in GCA is however, limited at the present time because of their cost and availability.

In terms of laboratory parameters, the CRP and ESR are two of the most commonly used parameters as part of the assessment of GCA. A large US population-based study, using retrospective electronic medical records from 3001 patients reported CRP levels of > 2.45mg/dL, ESR of 47-107mm/hr and platelet counts of > 400,000/mL to be the strongest laboratory predictors of a positive TAB [89]. In addition to the elevated ESR, CRP and platelet counts, normochromic normocytic anemia and an increase in acute-phase proteins on serum protein electrophoresis were also indicators of systemic inflammation in GCA patients.

In 1990, the ACR published a set of classification criteria for GCA (Table 2.2). A patient is classified as having GCA if at least three of these five criteria are present. The results have been reported to be associated with a sensitivity of 93.5% and a specificity of 91.2% when compared to patients with other forms of vasculitis [90].

Characteristics	Definition
Age at disease onset \ge 50 years	Development of symptoms or findings beginning at 50 years or older
New headache	New onset of or new type of localized pain in the head
Temporal artery abnormality	Temporal artery tenderness to palpitation or decreased pulsation, unrelated to arteriosclerosis of cervical arteries
Elevated erythrocyte sedimentation rate (ESR)	ESR \ge 50 mm/h by the Westergren method
Abnormal artery biopsy	Biopsy specimen with artery showing vasculitis characterized by a predominance of mononuclear cell infiltration or granulomatous inflammation, usually with multinucleated giant cells

Table 2.2: 1990 criteria for classification of giant-cell (temporal) arteritis [90]

The 1990 ACR classification criteria are sometimes used in clinical practice for diagnosis of GCA, this is however, inappropriate because of a few reasons. Firstly, these criteria were developed for use in clinical research and designed for a relatively homogenous cohort with typical presentations in mind, which is often not the case in clinical practice. In fact, the criteria have been reported to perform inadequately as a diagnostic tool [91, 92]. Secondly, the criteria were developed in the 1990s, long before the invention of new diagnostic techniques such as the computerized tomography and MRI that may be used to facilitate the diagnosis of GCA if available. Thirdly, the initial assumption of the ACR criteria is that the patient has systemic vasculitis, and the criteria are meant to inform the clinician whether it is GCA or not. This is often not what a clinician has in mind - in clinical practice the starting point is often a collection of symptoms that may or may not be related to systemic vasculitis [93]. In addition, symptoms may be heterogeneous in nature and differ in terms of severity, making it difficult to apply the criteria. Over the past decade, the applicability of the 1990 ACR classification in routine clinical practice has been in the limelight, and while it remains critically important in the conduct of clinical trials and epidemiologic studies with welldefined patient populations; it is not appropriate as a diagnosis tool [49].

2.3.4. Clinical Manifestations

The latest BSR guideline for the management of GCA [85] lists the following as common symptoms in GCA patients:

- Headache
- Scalp hyperaethesia
- Jaw or tongue claudication
- Temporal artery tenderness, nodularity or reduced pulsation
- Polymyalgia rheumatica (pain and stiffness of shoulder and hip girdles)
- Fever, sweats or weight loss

Less commonly, patients may also have the following symptoms [85]:

- Carotidynia
- Audiovestibular symptoms
- Dry cough
- Indications of tongue or scalp ischaemia that may precede necrosis

2.3.5. Treatment

2.3.5.1. Glucocorticoids (GC)

Glucocorticoids (GC) are the first-line treatment for GCA. In most cases, the initial dose of prednisolone ranges from 40-60mg/day. For patients with more serious symptoms such as acute or intermittent visual loss, intravenous methylprednisolone 500-1000mg/day may be given for up to 3 consecutive days before commencing oral GC [85]. In cases where intravenous therapy is not immediately possible, then oral GC should be initiated as soon as possible. The latest BSR guideline for the treatment of GCA suggests that GC dose should be tapered to zero over 12-18 months, providing there is no return of GCA symptoms, signs or laboratory markers of inflammation; and in patients at high risk of GC toxicity or those who are receiving concomitant GCsparing therapy, a more rapid dose reduction may be implemented [85]. There is some flexibility in terms of the tapering regimen, of which emphasis was given to an individualized patient approach in the latest BSR guideline. Full assessment of the disease and co-morbidities, in addition to patient's personal priorities, should inform decisions on the tapering regimen, or if there is a need for additional treatments such GC-sparing therapies [85]. One example of a GC tapering regimen is shown in Figure 2.5. Alternative approaches may include reducing GC dose by 10mg/week in patients who are in remission above 20mg daily, and/or reducing the dose slower than stated in

Figure 2.5 in patients who are on or below 5mg daily [85]. The 2018 EULAR for the management of GCA also recommends a similar GC tapering approach to Figure 2.5, as shown in Figure 2.6. Initial doses of 40-60mg/day of GC are recommended for patients with active GCA, and if they present with GCA-related visual symptoms, then intravenous methylprednisolone 250-1000mg/day for 3 days may be considered, in addition to oral GC [22]. While the BSR guideline did not give specific recommendations on how the dose of GC should be tapered with the addition of tocilizumab (TCZ) or methotrexate (MTX) in patients with refractory or relapse GCA, the 2018 EULAR guideline included a treatment algorithm involving the use of TCZ [22]. The GC dose of patients with a major relapse may be increased to 40-60mg/day and TCZ (or MTX) may be initiated. For patients with minor relapse, GC should be tapered to the last effective dose, in addition to TCZ or MTX. With the addition of TCZ, the EULAR guideline recommends GC to be tapered to 0mg at 6 months to reduce the cumulative dose. It is however, not known if faster or conversely more prolonged GC withdrawal during TCZ therapy may lead to improved outcomes due to lack of evidence [22].



Figure 2.5: An example of GC tapering regimen for the treatment of GCA based on the 2010 BSR guideline [7]



Figure 2.6: The 2018 EULAR algorithm for treatment of GCA [22]

Prompt treatment is essential to prevent irreversible complications, especially relating to blindness and stroke [94, 7]. Patients with GCA usually show a rapid improvement of clinical symptoms within 24-48 hours of treatment [95]. Laboratory measures of disease activity such as ESR and CRP usually improve substantially within a few days of therapy initiation [95].

While on GC therapy, patients should be regularly monitored for treatment efficacy, as well as safety. Laboratory monitoring of inflammatory markers such as full blood count, CRP and ESR (or plasma viscosity if ESR is unavailable) is highly recommended before or immediately after the initiation of high-dose GC. Baseline laboratory tests of major organ system function, including renal and liver function tests, calcium and alkaline phosphatase should also be monitored [85]. Since GC use is reported to be associated with numerous adverse events [96, 97], the BSR guideline recommends patients to be evaluated for hypertension and hyperglycaemia within the first two weeks of GC therapy initiation, as well as consider appropriate bone protection such as calcium and vitamin D, with oral biphosphonate if not contraindicated.

2.3.5.2. Biologic Agents

Several studies looked at the efficacy of several biologics as adjuvant therapy. Tocilizumab (TCZ) is a biologic that was approved by the US Food and Drug Administration (FDA) and European Medicines Agency (EMA) for the treatment of GCA in 2017, based on the efficacy shown in two randomized clinical trials (RCT). One RCT [98] demonstrated increased rates of remission (85% vs. 20%) at 52 weeks, in addition to significantly lower mean cumulative GC dose in the TCZ group (43mg/kg vs. 110mg/kg). Another RCT [99] also demonstrated increased rates of remission (53-56% vs. 14-18%) at 52 weeks, with significant lower median cumulative GC dose among TCZ patients (1862mg vs. 3818mg). TCZ is also shown to be more cost-effective for patients with relapsing or refractory disease. When compared with GC alone, the costeffectiveness estimated was £24,977 per quality-adjusted life year gained, assuming that patients receive TCZ for one year at most, and this was within the range normally considered to be a cost-effective use of National Health Service (NHS) resources [100]. The latest 2019 BSR guideline recommends TCZ to be used in combination with a GC taper for patients with relapsing disease, or for those at high risk of developing GC toxicity [85].

Ustekinumab is a monoclonal antibody, and it is one of the newest agents whose use is being explored in the treatment of GCA. In an open-label study of 25 patients with refractory GCA, the mean GC dose was decreased from a median of 20mg/day to 5mg/day at week 52 and six patients (24%) discontinued GC therapy entirely [101]. A current RCT is being carried out among patients with refractory or relapsing GCA, and recruitment is expected to complete in 2023 [102].

Three tumor necrosis factor (TNF) antagonists – infliximab, adalimumab and etanercept were found to be ineffective in the treatment of GCA, based on the results of three RCTs. The infliximab study was terminated early, as the proportion of patients on infliximab without relapse at 22 weeks did not increase as compared to placebo (43% vs. 50%, p=0.65). The proportion of patients whose GC dosages were tapered to 10mg/day without relapse was also not higher than placebo (61% vs. 75%, p=0.31) [103]. For the adalimumab study, results showed that the addition of adalimumab to GC did not increase the number of patients in remission at six months, as compared to placebo (59% vs. 50%, p=0.46) [104]. The etanercept study was a small single-centre RCT, of which the results showed that 50% of etanercept-treated patients were able to control disease activity without GC at the end of 12 months, as compared to 22% in the placebo group. This was however, not statistically significant. Etanercept-treated

patients also had a lower cumulative GC dose at 1 year as compared to placebo patients (1.5g vs. 3g, p=0.03), though the limited number of patients (n=17) makes it impossible to draw definitive conclusions.

The role of abatacept, a biologic that blocks the activity of T-cells, has also been explored in a small RCT. Trial results demonstrated modest improvements in relapse free survival (48% vs. 31%, p=0.049) at 12 months and longer median duration of remission (9.9 months vs. 3.9 months, p=0.023) in abatacept-treated patients [105]. To date, there is however, insufficient evidence to support its use in the treatment of GCA.

2.3.5.3. Methotrexate (MTX)

Three RCTs comparing low dose MTX with placebo in patients with GCA treated with GCs had divergent conclusions – two of the trials showed no effect [106, 107], whereas one reported reduced relapse rates and lower GC doses with the addition of MTX [14]. A meta-analysis of the individual patient-level data of 161 patients from these RCTs [108] suggested that the use of MTX as adjunct therapy resulted in a lower relapse rate, the need for lower GC cumulative dose and a higher rate of GC-free remission, but there was no difference in adverse event rates. The latest BSR guideline [85] and 2018 EULAR guideline for the treatment of GCA [22] suggests that MTX might be considered for GCA, in combination with a GC taper, in patients at high risk of GC toxicity or those with relapse.

2.3.5.4. Aspirin

To date, there are no RCTs published on the use of aspirin in GCA. A Cochrane systematic review was done in 2014, looking at the safety and efficacy of aspirin as an adjunctive treatment in GCA, of which the authors concluded that there was insufficient evidence to justify the use in routine clinical practice [109].

2.3.5.5. Other Immunosuppressants

One small RCT showed that azathioprine of 150mg/day demonstrated a significant reduction in mean GC dose over 52 weeks, as compared to those in the placebo group (1.9mg/day vs. 4.2mg/day at the end of 52 weeks, p<0.05) [110]. The study, however only had 31 patients enrolled, of which 7 of the azathioprine-treated (44%) withdrew from the study due to azathioprine related side effects.

A few small studies have suggested that cyclophosphamide may be useful in GCA patients at high risk of GC-related adverse effects [111-113]. A systematic review identified 88 patients who received cyclophosphamide, of which 74 (84%) were responsive to cyclophosphamide and only 17 (19%) had a relapse. It is worth noting though, that all the patients were receiving a maintenance therapy with other immunosuppressive agents such as MTX, of which the true effect of cyclophosphamide would have been confounded by the effect of another drug.

The potential of cyclosporin A as adjunctive therapy in the treatment of GCA was also explored in two randomized open-label studies [114, 115], of which the authors reported that cyclosporin A did not demonstrate a significant steroid-sparing effect.

The potential of other immunosuppressants such as leflunomide [116], mycophenolate mofetil [117] and dapsone [118] as a GC-sparing agent in GCA have been explored, but there is insufficient evidence to support their use as they were all case series or retrospective studies. It is therefore, not routinely used in clinical practice due to the lack of evidence in terms of safety and efficacy.

While there is no known cure for PMR and GCA, these diseases can be treated and controlled. As mentioned in this chapter, GC is the cornerstone of therapy. I will be discussing the role of GC in detail in the next chapter.
CHAPTER 3: GLUCOCORTICOIDS

3.1. Overview

Cortisone, the first steroid, was first discovered in 1929 by a group of scientists who prepared extracts from the adrenal cortex that successfully controlled symptoms of adrenal insufficiency in patients with Addison's disease [119]. Cortisone was eventually developed into a chemical compound and was first used in 1949 for the treatment of rheumatoid arthritis [120]. Nearly seven decades later, the clinical use of steroids has expanded greatly, mainly due to their potent anti-inflammatory and immuno-modulating properties. The term "steroid" applied to a wide range of molecules with varying physiological effects, including mineralocorticoids (MCs) and glucocorticoids (GCs). MCs such as aldosterone play a critical role in the regulation of sodium and water transport [121]. Naturally-occurring GCs are part of the feedback mechanism that the body utilizes to reduce inflammation, while exogenous GCs are used to treat diseases caused by an overactive immune system, including, but not limited to respiratory diseases, autoimmune diseases and rheumatic diseases. GCs have been used for the treatment of both polymyalgia rheumatica (PMR) and giant cell arteritis (GCA) since the 1960's, and to date, they are still the cornerstone treatment for both these diseases. GCs are like a double-edged sword - while they are considered "the miracle or wonder drug" for most patients due to their effectiveness in the rapid amelioration of symptoms, they are also harmful, and clinicians often need to deal with the dichotomy that GCs can both treat and harm patients. One report published by the PMRGCAuk group (a registered charity in Great Britain, established by a group of patients and clinicians) adopted the metaphor of "fighting a dragon" when dealing with GC use in PMR and GCA [122]. A dragon has a fiery head, a long spiny back, and a sting in the tail. At the point of diagnosis, patients struggle with acute pain and discomfort, and while GCs deal with this crisis period, it is followed by the challenge of managing the balancing act between GC dose titration and re-emergence of acute symptoms, as well as having to deal with the short to medium term adverse effects [122]. Once they have tapered GCs to a minimum (the dragon's tail), they are faced with the long-term complications of GC therapy such as diabetes mellitus (DM). While this thesis will not be the ultimate solution for this daunting problem, it is hoped that the results would shed some light on the management of GC-induced DM. Since GCs are the main exposure in my thesis, this chapter will discuss in detail the pharmacokinetics and mechanism of action, followed by the definition of conventional terms for GC doses. The last section of this chapter will include a discussion on GC-induced DM.

3.2. Pharmacokinetics

The hypothalamic pituitary adrenal (HPA) axis is a complex set of interactions between the hypothalamus, pituitary gland and adrenal gland. The adrenal gland secretes steroid hormones essential for the regulation of stress response, blood pressure and volume, fluid and electrolyte balance, and inflammation. The HPA axis involves the stimulation of adrenocorticotropic hormone (ACTH) release from the pituitary by the hypothalamus. Elevated blood levels of ACTH in turn stimulate the zona fasciculata in the adrenal cortex to produce cortisol, a hormone essential for the utilization of carbohydrates, fat and protein, as well as maintaining energy homeostasis during stress [123].

Synthetic GCs mimic the effects of cortisol and are the first class of endogenous antiinflammatory mediators that have been successfully used for therapy purposes. The two most common GCs are prednisolone and prednisone. They were introduced into clinical practice in the 1960s, with prednisolone being primarily prescribed in the UK, and prednisone in the US. Prednisolone is an active steroid, while prednisone is a prodrug that is converted by the 11β -hydroxysteroid dehydrogenase 1 (11β -HSD1) to the active metabolite prednisolone in the liver after administration. Prednisolone is well absorbed from the gastrointestinal tract after oral administration and is known to moderately bind to the corticosteroid binding globulin (i.e. glycoprotein transcortin) and albumin in a non-linear manner [124-126]. Several pharmacokinetic studies have shown that the unbound fraction of prednisolone accounts for its biological effect, rather than the total serum drug concentration [127-129]. In other words, only the free drug that reaches the site of action and interacts with the receptor is biologically relevant; and since it exhibits non-linear binding properties, the dose-response relationship is also non-linear [130]. The plasma half-life after oral administration of prednisolone is between 3 to 4 hours, depending on the dose [131, 132].

3.3. Mechanism of Action

The host inflammatory response is a primary defence mechanism that is immediately activated in response to any infection or injury. There are two types of responses, namely the innate and adaptive immune systems [133]. The innate immune system is the first line of defence in acute inflammation, primarily composed of cells that serve as a barrier function and phagocytes recruited early to sites of pathogen invasion. It is relatively non-specific and provides a rapid reaction to infections or injuries. The adaptive immune system, on the other hand, consists of lymphocytes and antibodies that plays a major role in chronic inflammatory disease, provides a highly antigen-specific response, and develops much later in the host inflammatory response. GCs inhibit many of the initial events in an inflammatory response. They promote the resolution of inflammation by the following mechanisms: firstly, inhibiting vasodilation and therefore prevent the increase in blood flow; secondly, preventing increases in vascular permeability, thereby reducing exudate formation; and thirdly through the suppression of leukocyte emigration [134].

3.3.1. Genomic Mechanisms

A few studies have proposed two types of mechanisms that inhibit events of the inflammatory and immune response – genomic and non-genomic [135-137]. Genomic effects are mediated by cytosolic glucocorticoid receptors (cGR). These receptors are members of the steroid hormone receptor family, which is a superfamily of ligandinducible transcription factors. When administered, the lipophilic GCs will bind to the ligand-binding domain of the cGR with high affinity, which in turn results in activation of the receptor and disassociation of the receptor from the multiprotein complex [138]. Nuclear translocation of the GC-cGR complex occurs within 10-30 minutes of cell exposure to GC, which will then bind to DNA binding sites called the glucocorticoid response elements (GREs) [138]. The GC-cGR complex will then activate the transcription of anti-inflammatory factors (e.g. interleukin-10, annexin 1 and inhibitor of nuclear factor κB), as well as regulatory proteins [139]. This process, which is thought to be responsible for numerous adverse effects of GCs, is called transactivation [140]. Another mechanism of action is transrepression, which describes the ability of the GCcGR complex to inhibit the transactivation function of transcription factors such as activator protein 1 and nuclear factor κB [141, 142]. It has been suggested that the anti-inflammatory and immunosuppressive effects of GCs are induced by genomic transrepression, by which synthesis of proinflammatory mediators such as the interleukin-1 (IL-1, IL-2, tumor necrosis factor, interferon gamma and prostaglandins are suppressed [140, 143]. These genomic effects occur at any therapeutically relevant dosage, and are generally slow due to the time-consuming process of messenger ribonucleic acid (mRNA) transcription and translation. The degree of cytosolic receptor saturation is considered as a direct modulator of the intensity of therapeutic anti-inflammatory and immunosuppressive GC effects [144]. Some of the anti-inflammatory and immunosuppressive effects (e.g. rapid clinical response to administration of high dose intravenous GC), however, occur too rapidly to be explained by this genomic mode of action, therefore it is likely that non-genomic mechanisms are also involved [137, 145, 146].

3.3.2. Non-genomic Mechanisms

As outlined above, some GC responses do not fit the classical genomic model of GC action due to its rapid occurrence in response. Losel and Wehling defined non-genomic mechanisms as any action that does not affect gene expression initially or directly, but rather drives more rapid effects such as the activation of signaling cascades [146]. The following non-genomic effects of GCs have been studied and documented [147]:

- GCs exert rapid effects on intracellular calcium homeostasis and agonistinduced calcium mobilization
- GCs rapidly modulate skeletal and smooth muscle function
- GCs exert rapid effects on reactive oxygen species / reactive nitrogen species
- GCs exert rapid effects on inflammatory and apoptotic pathway

There are many factors that affect the clinical response of GCs, including the rate of absorption, concentration in the target tissues, affinity of GC to the GC receptors, rate of metabolism and rate of clearance, as well as the dose administered, as GC effects are strongly dose-dependent [148]. Figure 3.1 summarizes the occurrence of genomic and non-genomic effects in terms of a dose-response relationship [135, 136, 149, 144] that will be further discussed in Section 3.3.3.



Figure 3.1. Genomic and non-genomic effects on the dose-response relationship of GC *Note: Reproduced with permission from RightsLink / BMJ Publishing Group Ltd, permission granted on August 28, 2019 by email. F Buttgereit et al. Standardised nomenclature for glucocorticoid dosages and glucocorticoid treatment regimens: current questions and tentative answers in rheumatology. 2002: 61:718-722*

3.3.3. GC Effects on PMR and GCA

Endothelial cells play an active role in the regulation of inflammation and angiogenesis. The intercellular adhesion molecules on endothelial cells mediate leukocyte trafficking, migration, and development of inflammatory infiltrates [150]. Increased expression of endothelin 1 and endothelin B immunoreactivity has been reported to be in higher concentration in GCA patients compared to controls and may correlate with the degree of systemic inflammation [151]. In another study, Pirro et al. demonstrated that PMR was associated with a significant imbalance between endothelial injury and repair, which was dependent on the degree of systemic inflammation [152]. Levels of C-reactive protein (CRP) were elevated and associated with an increase in circulating endothelial microparticles, of which there was a significant reduction in levels following GC therapy [152].

When triggered by antigens, the immune response starts with activation of macrophages, followed by antigen recognition and activation of T cells. Interleukin-6 (IL-6) is a cytokine that plays an important role in the stimulation of antibody production in the presence of antigens. Two decades ago, the elevation of circulating IL-6 was reported in patients with GCA, and serum levels were reported to correlate with

27

disease activity [153, 154]. IL-6 functions as an important connector between injured vascular walls and immune cells, and is critically involved in promoting the differentiation of the Th17 lineage, which in turn produces a plethora of cytokines that regulate local and systemic inflammatory effects in GCA [155]. In addition, IL-6 also regulates the induction of anti-inflammatory T regulatory (Treg) cells [155]. Thus, IL-6 has two important roles in GCA - promoting pro-inflammatory T cell immunity and inhibiting the action of opposing anti-inflammatory T cells. In GCA, two distinct cluster of differentiation 4+ (CD4+) T effector cell subtypes have been identified as key regulators: type 17 helper T cell (Th17) and type 1 helper T cells (Th1) [156]. The Th17 pathway has been reported to be very responsive to GC treatment, as GCs rapidly reduce the Th17 effector cytokine production of IL-1, IL-6, IL-17 and IL-23 with simultaneous depletion of both circulating and tissue infiltrative Th17 cells [155, 157]. The rapid decline in these cytokines upon GC initiation is evident by the marked improvement of systemic inflammatory features in patients. Despite the effective reduction of the Th17 pathway, the Th1 cytokine identified in chronic vasculitis in GCA is associated with production of IL-2 and interferon-gamma, and is poorly susceptible to GCs [150]. It is possible that the persistence of Th1 cellular infiltrates despite prolonged GC treatment is responsible for the relapsing nature of GCA [155]. When compared with healthy controls, another study reported that patients with GCA had a decreased frequency of regulatory T cells and Th1, but had a significantly increased percentage of Th17 cells [158]. The initiation of GC treatment reduced the Th1 and Th17 cells, but not the regulatory T cells [158].

Adaptive immune alterations also occurred in PMR patients, mainly represented by the activation of Th17 cells mainly driven by increased IL-6 levels [159]. Patients with PMR have been reported to have higher IL-6, IL-1Ra and serum B-cell activating factor (BAFF) levels that are related with their clinical symptoms [160-162]. GCs act by suppressing IL-6 production rapidly but do not correct the underlying mechanism, resulting in increased IL-6 production following short-term withdrawal of GC, even after several months of treatment [163]. One study also reported that a decrease in the level of circulating IL-6 correlates with remission of clinical symptoms, and that high serum IL-6 receptor levels combined with low haemoglobin values resulted in a ten-fold increased risk of PMR relapse [164]. As with GCA, patients with PMR, when compared to healthy controls, had a decreased frequency of regulatory T cells and Th1, but had a significantly increased percentage of Th17 cells, of which was reduced by GC treatment [158]. In addition, PMR patients have been reported to have a decreased frequency of circulating B cells, of which rapid recovery is seen with the initiation of GC therapy [165].

3.4. Definition of Conventional Terms for Glucocorticoid Doses

In 2001, a panel of experts from the European League Against Rheumatism (EULAR) Standing Committee on International Clinical Studies including Therapeutic Trials gathered to discuss the management of GC therapy in rheumatic diseases [144]. One of the recommendations by the panel was on the relationship between clinical GC dosing and cellular GC actions.

Treatment with prednisolone-equivalent doses of less than 7.5mg per day is termed "low-dose" GC therapy, as this would result in a saturation of GC-receptor complex of approximately 50% [166, 167]. It is often used for maintenance therapy for rheumatic diseases such as rheumatoid arthritis (RA) and systemic lupus erythematosus (SLE) [168, 169], and is associated with relatively few adverse effects, as reported by Da Silva JA et al. The authors did a literature review of the adverse effects of low dose GC in rheumatic disease, which include the following subgroups: musculoskeletal, endocrine, cardiovascular, dermatological, ophthalmological, gastrointestinal, infectious as well as psychological and behavioural disturbances; in addition to an analysis of toxicity data from randomized controlled trials of GC in RA. Of all these adverse effects, they also looked at glucose intolerance and DM, of which the authors concluded that low-dose GC treatment did not increase the risk of DM [170].

"Medium dose" is defined as an administered dose of > 7.5mg, but \leq 30mg prednisolone equivalent a day with higher receptor saturation ranging above 50% but not completely saturated [166]. These doses are used in initial treatment for primary chronic rheumatic diseases such as acute gouty arthritis, juvenile idiopathic arthritis, PMR and SLE [171]. Adverse effects are considerable and dose dependent, which depends very much on the duration of treatment.

Doses of > 30mg and ≤ 100 mg prednisolone equivalent a day are considered as "high dose" because these doses significantly increase receptor saturation in a dose dependent manner resulting in an almost complete receptor saturation of 100% [166]. These doses are usually not administered as long-term therapy because of the occurrence of severe adverse effects, but only used as initial treatments for subacute rheumatic diseases, which includes systemic vasculitides such as GCA [171] and non-life threatening exacerbations of connective tissue diseases such as SLE [172].

Doses of > 100mg prednisolone equivalent a day are considered as "very high dose". At these doses, the cytosolic receptors are 100% saturated [166], and additional therapeutic benefit of very high doses could be obtained via non-specific, non-genomic effects. This is when physicochemical interactions with cellular membranes occurs and mediated by membrane bound receptor [135-137]. These doses are usually only given as initial doses for life threatening conditions or acute conditions such as vasculitis and SLE [173, 174]. They are not given for long-term treatment because of the severe adverse effects associated with these doses.

The panel of experts also proposed another category, called "pulse therapy", defined as administration of \geq 250mg prednisolone equivalent a day, usually given intravenously for less than 5 days. This term refers more to the very high dose given, rather than its intermittent characteristic in terms of time, for the following reasons: the first is that in clinical practice, doses of more than 250mg/day are usually only given as pulse therapy, thus these doses are exclusively given for a few days, but then reduced or stopped directly. Secondly, the non-genomic potencies of GCs generally come into play at these doses, which contributes to the success of very high doses and pulse therapy in acute exacerbations of immunological mediated diseases, such as SLE, vasculitis, RA and polymyositis [173-175]. The panel also suggested that pulse therapy results in termination of disease exacerbations in many cases with a relatively low incidence of adverse effects.

These proposals for nomenclature of GC treatment with its dose-response relationship reflects current and best knowledge available, and more importantly, it also reflects best practice in routine clinical care. Although GCs are widely used because of their immense therapeutic benefits as anti-inflammatory and immunosuppressive agents, many issues remained to be clarified, including their specific mechanism of actions, quantification of adverse events (AEs), as well as measures to counter these AEs associated with long-term GC use.

3.5. Glucocorticoid-Induced Adverse Events (AEs)

Chronic GC therapy is reported to be associated with various AEs, which includes DM, infections, fractures, gastrointestinal bleeding, hypertension, cataract, adrenal suppression, weight gain, myopathy and neuropsychiatric disorder [96, 176]. Despite nearly six decades of established use, the precise risks of these GC-related AEs have been difficult to quantify, mainly because of confounding by indication. Patients with more severe disease often have a greater burden of inflammation, therefore resulting in either a longer duration of treatment, or a higher dose of treatment, which usually leads to a greater risk of developing an AE. Of all the AEs, DM (14%) was ranked as the top most worrisome from a rheumatologists perspective, while it ranked 2nd (7%) (tied with cardiovascular diseases (CVD) (7%)), and after osteoporosis (9%) among patients [15].

Data from the general population in England and Wales showed that at any point in time, approximately 0.9% of the adult population was prescribed oral GCs, with the highest use (2.5%) by those aged between 70 to 79 years old [177]. GCs are relatively inexpensive, but their high usage in medical treatment, ranging from rheumatology to respiratory to endocrine, cancer, bowel diseases and allergies collectively incur significant financial burden to the healthcare system. Manson et al [178] did a cost analysis study, using the incidence rates of seven most commonly reported AEs (fractures, cataracts, DM, peptic ulcer, stroke, myocardial infarction and non-Hodgkin's lymphoma) in the general population of the UK to calculate the estimated total number of events caused by oral GCs. The cost per event or per year of treatment was then applied to estimate the annual economic burden attributable to oral GC-induced adverse events. The authors reported that the total cost for all seven AEs amounted to approximately £165 per oral GC treated patient per year. Diabetes mellitus incurred the second highest cost per patient, amounting to a total of £12.39 cost per treated patient or £2,519.86 cost per episode per year.

3.5.1. Glucocorticoid-Induced Diabetes Mellitus (DM)

Glucocorticoid-induced DM (GC-induced DM) or steroid-induced DM, is defined as an abnormal increase in blood glucose concentration during GC use in patients with or without a previous history of diabetes [16]. The American Diabetes Association's criteria for diagnosing DM is an 8 hour fasting blood glucose of \geq 7.0 mmol/L, 2 hour post 75g oral glucose tolerance test of \geq 11.1 mmol/L, HbA1c \geq 6.5% or a random plasma glucose of \geq 11.1 mmol/L in patients with symptoms of hyperglycemia [179].

A meta-analysis of studies evaluating the occurrence of DM in non-diabetic individuals treated with GCs reported that the rates of GC-induced DM and GC-induced hyperglycaemia were 18.6% and 32.3%, respectively [180]. A large nested case-control study in the UK using The Health Improvement Network (THIN) primary care database reported an adjusted odds ratio of 1.36 (95% confidence interval (CI): 1.10-1.69) for DM associated with three or more oral GC prescriptions versus no GC use [181]. Another large population-based cohort study using administrative databases in Canada reported that the risk of developing DM among the elderly was significantly higher in those initiated with an oral GC compared to the control group (adjusted rate ratio: 2.31; 95%CI: 2.11-2.54) [182]. Though none of these studies specifically involved PMR or GCA patients, there seemed to be a consensus that use of GCs is associated with an increased risk of developing DM.

The pathophysiology of GC-induced DM is complex and not completely understood, but there have been a few mechanisms proposed pertaining to how GC affects glucose homeostasis in the liver, skeletal muscle, adipose tissues, pancreatic beta cells and small intestines, as shown in Figure 3.2 [183].

Gluconeogenesis is a metabolic pathway that results in the generation of glucose from non-carbohydrate carbon substrates such as lactate, glycerol and glucogenic amino acids [184]. The two enzymes involved in the regulation of this pathway are phosphoenylpyruvate carboxykinase (PEPCK) [185] and glucose-6-phosphatase (G6P) [186]. Administration of GCs increase the gluconeogenesis process in the liver, as well as increase glucose production by limiting the metabolic actions of insulin. In adipose tissues, GCs promote lipid breakdown (i.e. lipolysis), resulting in an accumulation of fatty acids which interfere with glucose uptake [187]. GCs are also reported to increase the plasma concentrations of resistin, an adipose tissue-specific secretory factor and leptin, a hormone released from fat cells in adipose tissues that helps to regulate energy balance; while suppressing adiponectin, resulting in decreased insulin sensitivity and greater insulin resistance.

GCs also interfere with glucose homeostasis in the skeletal muscles by reducing protein synthesis and stimulating protein degradation (i.e. proteolysis), resulting in higher concentrations of amino acids and a reduction in muscle mass or progressive muscle atrophy, otherwise also clinically known as steroid-induced myopathy [188, 189]. GCs also contribute to a reduction in glucose uptake and an increase in the breakdown of glycogen to glucose (i.e. glycogenolysis) [190].

In addition, results from a randomized controlled trial of GC treatment in healthy individuals suggested that GC administration may dose-dependently impair insulinstimulated capillary recruitment, which is strongly related to insulin resistance, increased postprandial glucose levels and hypertension [191].

Incretins are a group of metabolic hormones that stimulate a decrease in blood glucose levels. The "incretin effect" is the increased stimulation of insulin secretion elicited by oral as compared with intravenous administration of glucose under similar plasma glucose levels. [192]. It is mediated by a gut hormone, glucagon-like peptide 1 (GLP-1) and glucose dependent insulinotropic polypeptide (GIP), a hormone produced by the small intestine that enhances the release of insulin following the intake of food [193-195]. The incretin effect is almost absent in individuals with type 2 DM [196, 197] and diminished in those with impaired glucose tolerance [198] and those who are obese [199]. The use of GCs is associated with a reduction in the incretin effect. It is however, unrelated to the secretion of incretin hormones but is related to insulin resistance and beta cell defects [200].

GCs have also been suggested to inhibit the production and secretion of insulin from pancreatic beta cells [201, 202]. It has been hypothesized that a reduction in both insulin release and biosynthesis [203] and the pro-apoptotic effect of prolonged GC use [204-206] contribute to beta cell failure over time. Another study also reported an increase in fasting and post-prandial glucagon levels after the administration of GCs, which further supports the hypothesis that GCs have a role in inducing beta cell dysfunction [207]. In addition, prolonged use of GCs may elevate the levels of plasma triglyceride and free fatty acids concentrations, leading to lipotoxicity, which indirectly induces beta cell failure [208, 209].





muscle, adipose tissue, pancreatic beta cells and small intestine.

Note: Reproduced with permission from RightsLink / Elsevier, permission granted on August 28, 2019 by email. Bonaventura A et al. Steroid-induced hyperglycemia: An underdiagnosed problem or clinical inertia? A narrative review. Diabetes Res Clin Prac. 2018: 139: 203-220

The EULAR task force recently published a paper on defining conditions where long-term GC treatment has an acceptably low level of harm to facilitate implementation of existing recommendations [97]. The authors also outlined a list of studies that were either published within 2010–2015, or were selected older outstanding landmark papers. The results were classified according to the four most worrisome AE of GC therapy as well as general literature on GC-induced AE, which include osteoporosis, DM, CVDs and infections. Since no specific information was given on the studies listed in this report, I conducted a review of relevant papers cited in the report [97] that were related to GC-induced hyperglycemia and/or DM and tabulated the results (Table 3.1) using the PICO (Population, Intervention, Comparator and Outcome) criteria. Effect size and statistical significance of the findings were included if the authors reported them.

1 st Author, Year	Population and study design	Treatment Groups (n)	Outcome
Burt, 2012 [210]	Cross-sectional study of subjects with inflammatory rheumatological disease	 Chronic prednisolone for > 6 months (60) Control: No GC for at least 6 months (58) 	 Significantly lower fasting glucose (net difference of 0.3mmol/L) for chronic prednisolone users Significantly higher post-glucose load glucose concentration (net difference of 1.2mmol/L) for chronic prednisolone users
Den Uyl D, 2012 [211]	Randomized controlled trial (RCT) of patients with early active RA	 Prednisolone 60mg/day (21) Prednisolone 30mg/day (20) 	 Incidence of type 2 DM increased by 17% from baseline (p<0.01) (evenly distributed across groups)
Fong, 2013 [212]	Audit of patients commencing high-dose steroid therapy in a tertiary referral hospital	 Prednisolone 25mg/day (49) Dexamethasone 4mg/day (18) Hydrocortisone 100mg/day (4) Combination (9) * minimum of 48 hours 	 Mean blood glucose was >/= 8mmol/L in 48% of pts & >/=10mmol/L in 14% of patients Among those with hyperglycemia, it developed within 48 hours in 94% of subjects

Table 3.1: Literature on glucocorticoid-induced hyperglycemia and diabetes mellitus selected by the EULAR task force

Petersons, 2013 [213]	Matched case control of patients with inflammatory rheumatologic disease	 Subjects initially not on GC, then 7-10 day course of oral prednisolone 6mg/day (9) Subjects on continuous long-term prednisolone (6.3 +/- 2.2mg/day) (12) 	 Prednisolone increased basal endogenous glucose production (EGP*) (p=0.05) & reduced insulin suppression of EGP (p=0.03), peripheral glucose disposal** (p=0.01), first phase (p=0.01) & second phase*** (p=0.02) insulin secretion Long-term prednisolone users had attenuated insulin suppression of EGP (p=0.03) & non-oxidative glucose disposal (p=0.02), whereas basal EGP, insulin secretion & adipose tissue areas were not significantly different
Raul Ariza- Andraca, 1998 [214]	Matched case control of patients with steroid-induced DM with rheumatic diseases	 Cases (27) Age & sex matched controls who were also on GC (27) 	 Cumulative prednisone dose (26.6 +/- 28g) was associated with the development of steroid-induced DM (OR=6.35, p<0.02) No significant differences in serum insulin levels
Rostom, 2013 [215]	Matched case control of RA patients	 Cases (120) Age & sex matched healthy controls (100) 	 6 metabolic syndrome definitions**** were used Frequency of metabolic syndrome (MetS) was significantly higher than control for all 6 definitions: ranging between 18-48.6% (p<0.05) GC use was a significant independent predictor of the presence of MetS in RA pts (OR=1.45, CI: 1.12-2.14, p=0.04)
Su, 2013 [216]	Cohort of Taiwanese citizens with and without RA	RA (4193)Without RA (596502)	 Relative risk (RR) for T2DM in RA vs. non RA patients a) Men: RR =1.68 (CI: 1.53-1.84) b) Women: RR = 1.46 (CI: 1.39-1.54) Absolute risk of T2DM in the cohort = 19%
Zeng, 2010 [217]	Cohort of Chinese female patients with SLE	Patients with SLE on GC: 146	 46/146 (31.5%) patients had hyperglycemia as defined by the World Health Organization (WHO) diagnostic criteria Age ≥35 years and high GC doses (defined as current prednisone doses of ≥ 25mg/day or mean monthly prednisone doses of ≥ 570mg/month) were risk factors for hyperglycemia

Note:

* Endogenous glucose production is the formation of glucose from substrates and is a physiological function that normally assists in self-regulation of blood glucose levels

** Peripheral glucose disposal is the disposal of glucose from the blood by the peripheral tissues such as skeletal muscles

*** After a meal, insulin is released within the beta cell, and the first phase of insulin secretion promotes peripheral utilization of the prandial nutrient load, suppresses hepatic glucose production and limits postprandial glucose elevation. This process begins within 2 minutes of nutrient indigestion and continues for 10-15 minutes. The second phase of prandial insulin secretion then follows, and is sustained until normoglycemia is restored. **** Joint Consensus 2009, National Cholesterol Education Programme 2004 and 2001, International Diabetes Federation, World Health Organisation and European Group for Study of Insulin Resistance The overall trend suggests that GC administration does increase the risk of hyperglycemia, and some cases, GC-induced DM. The results however, must be interpreted with caution for the following reasons. Firstly, most of the studies (especially those with observational designs) were associated with a high risk of bias, especially confounding by indication. The sample sizes for most of the studies were too small to draw any significant conclusions (although in many of the studies, the study authors reported statistically significant results), and in many of the studies, the effect sizes of the study outcomes were not reported. There is also a lot of heterogeneity across studies. The studies include various patient populations (e.g. patients from the European Union, UK, US, Taiwan, China) with various diseases (ranging from children with juvenile idiopathic arthritis to adults with RA) on various treatment regimens (wide range of GC doses and duration depending on indication and disease severity). In addition, the study designs also varied across the literature review - there were a mix of randomized controlled trials, cohorts and case-control designs. There is also a need to update this literature review as the papers selected by the task force was only until 2015, of which there have been an increased interest in the metabolic AEs of GCs in the recent years. It is also worth noting that none of these studies included patients with GCA or PMR, possibly because the diagnoses of GCA and PMR are more challenging, and also because of their lower incidence rate compared to the other rheumatic diseases, therefore there is still a huge research gap in these two disease domains.

As shown in Figure 3.3, the EULAR task force concluded that at doses of \leq 5mg/day, there was an acceptably low level of harm; and at >10mg/day, the risk of harm was elevated. At dosages between >5mg/day and \leq 10mg/day, uncertainty still exists and patient-specific characteristics need to be taken into consideration to estimate the individual risk of harm. In my opinion, the actual risk of GC-related harm may be hard to define, as there are many factors (such as age and lifestyle) that may influence the risk, even without GC exposure. GCs should still be used for treatment in clinical practice, given the known beneficial effects, but studies with more robust study designs are needed to quantify the risk association with GC use so that existing recommendations on GC therapy may be better implemented and practiced to optimize patient care.



Figure 3.3. The level of harm of long-term GC therapy in rheumatic diseases

Note: Reproduced with permission from RightsLink / BMJ Publishing Group Ltd, permission granted on August 28, 2019 by email. Strehl C et al. Defining conditions where long-term glucocorticoid treatment has an acceptably low level of harm to facilitate implementation of existing recommendations: viewpoints from an EULAR task force. Ann Rheum Dis 2016: 75 (6): 952-957

GCs have been used widely in the treatment of PMR and GCA for more than half a century. While their therapeutic efficacy is well established, there is still a huge research gap in terms of determining the magnitude of long-term harm in PMR/GCA. As mentioned in this chapter, DM is one of the most concerning AE associated with GC, and the next chapter will discuss in detail, a systematic review and meta-analysis of the risk of GC-induced DM among PMR and/or GCA patients based on published studies.

39

CHAPTER 4: ASSOCIATION BETWEEN GLUCOCORTICOID THERAPY AND INCIDENCE OF DIABETES MELLITUS IN POLYMYALGIA RHEUMATICA AND GIANT CELL ARTERITIS: A SYSTEMATIC REVIEW AND META-ANALYSIS

4.1. Overview

Long-standing DM is associated with complications such as chronic kidney disease, retinopathy, diabetic neuropathy etc. which often lead to a decline in quality of life (QoL), as well as marked shortening of life expectancy. This is also vital from a health economic perspective as it incurs increased additional healthcare utilization. Despite the importance of this for patients, clinicians and health authorities, there is currently minimal information on the absolute risk of GC-induced DM in patients with PMR and GCA. To address this clinically important question, I conducted a systematic review and meta-analysis of published studies to determine the risk of GC-induced DM in patients with PMR and/or GCA and assessed the potential risk factors associated with the development of DM in these patient groups.

4.2. Methods and Materials

4.2.1. Search Strategy

I searched for published studies or conference abstracts indexed in Pubmed, Ovid (Medline, Embase), Web of Science, Cochrane Library and Cumulative Index of Nursing and Allied Health Literature (CINAHL) from inception of each database to February 2017. The search included terms for patients with PMR and/or GCA who were prescribed oral GC therapy, as well as diabetes-related terms as some individuals eventually developed diabetes. The full search strategy is available in Appendix 1. I also manually screened reference lists of selected retrieved articles to identify further papers that may have been missed in the database search. I made every effort to include all available studies and conference abstracts (regardless of publication year), which included contacting the first authors by email if necessary. The selection process of identifying relevant studies is shown in Figure 4.1.



Figure 4.1: Flowchart of the selection process of studies

4.2.2. Eligibility Criteria

4.2.2.1. Inclusion Criteria

- All original research articles and conference abstracts that reported new onset diabetes following exposure to oral GC therapy in patients with PMR and/or GCA
- All randomized controlled trials (RCTs), cohort studies, cross-sectional studies and nested case-control studies

4.2.2.2. Exclusion Criteria

- All original research articles and conference abstracts that reported any preexisting diabetes or any pre-diabetic states (e.g. increased fasting glucose, glucose intolerance) or exacerbation of pre-existing diabetes (e.g. worsened diabetes, addition of insulin)
- Case series and case reports

4.2.3. Study Selection

A colleague and I independently carried out the initial screening of search results by title and abstract, using the abstract screener software "Abstrackr" [218] and Endnote X7.4 (1988-2015 Thomson Reuters). Study eligibility was determined independently and any disagreements were resolved by consensus with my supervisor.

4.2.4. Data Extraction

Data were extracted independently by my colleague and I using a standardized data collection Microsoft Access database. Any discrepancies in data extraction were resolved by consensus. For articles containing more than one study treatment group, I included the "GC-only" arm and excluded the groups where there was another study drug in addition to GCs. Information extracted included journal information, publication year, year/s of enrollment, study design, patient demographics (age, sex, body mass index (BMI), weight, height, medical history, family history, glucose level and hemoglobin A1C (HbA1c)), number of patients recruited, number of patients with DM, definition of DM, GC indication, treatment dose and duration, and duration of follow-up. Patient populations were classified as "PMR" or "GCA" based on their primary diagnosis as determined by the authors. Non-English articles were translated with the help of Google Translate and colleagues who were native speakers of the respective languages.

I used the Meta-analysis of Observational Studies in Epidemiology (MOOSE) reporting guidelines [219] as a checklist (Appendix 2) to ensure proper evaluation of quality and completeness of the observational studies.

The risk of bias was also assessed, using the Cochrane Collaboration's tools proposed by the Cochrane Group for RCTs [220] and for cohort studies [221]. My colleague and I did the risk of bias assessment independently, and discrepancies were resolved by consensus.

Domain	Support for judgment
SELECTION BIAS	
Random sequence generation	Describe the method used to generate the allocation sequence in sufficient detail to allow an assessment of whether it should produce comparable groups.
Allocation concealment	Describe the method used to conceal the allocation sequence in sufficient detail to determine whether intervention allocations could have been foreseen in advance of, or during, enrolment.
PERFORMANCE BIAS	Describe all measures used, if any, to blind
Blinding of participants and	study participants and personnel from
personnel	knowledge of which intervention a participant
	received. Provide any information relating to
DETECTION BIAS	Describe all measures used if any to blind
Blinding of outcome	outcome assessors from knowledge of which
	intervention a participant received. Provide any information relating to whether the intended blinding was effective.
ATTRITION BIAS	Describe the completeness of outcome data for
Incomplete outcome data	each main outcome, including attrition and exclusions from the analysis. State whether attrition and exclusions were reported, the numbers in each intervention group (compared with total randomized participants), reasons for attrition/exclusions where reported, and any re- inclusions in analyses performed by the review authors.
REPORTING BIAS	State how the possibility of selective outcome
Selective reporting	reporting was examined by the review authors, and what was found.

Table 4.1: The Cochrane collaboration's tool for assessing risk of bias in RCTs [220]

Table 4.2: The Cochrane	tool for assessing	risk of bias in	cohort study [221]
	U		

Domain	Support for judgment
Selection of population	Was selection of exposed and non-exposed cohorts drawn from the same population, presenting at same points of care over the same time frame?
Assessment of exposure	Can we be confident in the assessment of exposure? Was the exposure retrieved from secure records (e.g. surgical records, pharmacy records) or was it based on self-reporting or interviews?
Ascertainment of outcome not present at start of study	Can we be confident that the outcome of interest was not present at start of study?
Adjustment of prognostic variables	Did the study match exposed and unexposed for all variables that are associated with outcome of interest or did the statistical analysis adjust for these prognostic variables?
Assessment of prognostic variables	Can we be confident in the assessment of the presence or absence of prognostic factors? How was the prognostic data extracted? Was it based on review of charts and established databases with reproducibility demonstrated?
Assessment of outcome	Can we be confident in the assessment of outcome? Was it based on record linkage and documented medical records? How was the outcome of interest defined?
Assessment of Follow-up	Was the follow up of cohorts adequate? Were there any missing outcome data?

I used Cohen's kappa score to test for interrater reliability, where 0 represents the amount of agreement that can be expected from random chance, and 1 represents perfect agreement between the raters [222]. There were two different sets of Kappa scores – one for all RCTs, and the other for all observational studies. Kappa result can be interpreted as follows [222]:

- 0.00-0.20: None
- 0.21-0.39: Minimal
- 0.40-0.59: Weak
- 0.60-0.79: Moderate
- 0.80-0.90: Strong
- > 0.90: Almost perfect

4.2.6. Statistical Analysis

Baseline characteristics such as age and sex were summarised using descriptive statistics. The weighted mean and standard deviation were calculated by taking into account the sample size of each study for the following variables: age at baseline, proportion of females, cumulative dose, duration of GC use and duration of follow-up. I chose to use the weighted mean as this method takes into consideration the relative importance (based on sample size) of each study. The incidence proportion of new DM cases was modeled using binomial regression where the outcome variable was the number of newly diagnosed DM patients divided by the number of GC-treated patients at risk of developing DM. The candidate explanatory variables included were: mean age, year of enrolment, diagnosis (PMR or GCA), cumulative dose and treatment duration. Variables were selected a priori based on clinical knowledge for inclusion in the multivariable modelling rather than taking a purely data-driven approach. Results were presented as odds ratio (OR) and 95% confidence interval (CI). I carried out a meta-analysis using a random effects model, of which it is assumed that the observed estimates of effect can vary across studies because of real differences in the effect for each study, as well as sampling variability. Heterogeneity in the effect may be caused by differences in study population (e.g. age), interventions received (e.g. dose of GCs), and follow-up duration [223].

Heterogeneity was assessed using the I-square and tau statistics. I-square values less than 40% suggest that heterogeneity might not be important; values within the range of 30-60% suggest that there is moderate heterogeneity, while values above 75% represent considerable heterogeneity [224]. Statistical analysis was done with R (Version 3.3.1.) [225].

First Author, Publication Year	Country	Population	Study Design	Study Enrolment period
Von Knorring, 1979 [226]	Finland	PMR	Cohort	1967-1977
Godeau, 1982 [227]	France	GCA	Cohort	1966-1979
Chuang, 1982 [47]	USA	PMR	Cohort	1970-1979
Behn, 1983 [228]	UK	PMR	Cohort	1968-1980
Gouet, 1985 [229]	France	GCA	Cohort	1970-1984
Andersson, 1986 [230]	Sweden	GCA	Cohort	1968-1975
Delecoeuillerie, 1988 [231]	France	GCA	Cohort	1976-1986
Nesher, 1994 [232]	Israel	GCA	Cohort	1978-1992
Gabriel, 1997 [176]	USA	PMR	Cohort	1970-1991
Jover, 2001 [14]	Spain	GCA	RCT	1993-1997
Proven, 2003 [13]	USA	GCA	Cohort	1950-1991
Hutchings, 2007 [233]	UK	PMR	Cohort	2001-2003
Salvarani, 2007 [234]	Italy	PMR	RCT	2003
Cimmino, 2008 [235]	Italy	PMR	RCT	1998
Schmidt, 2008 [236]	Germany	GCA	Cohort	1997-2006
Dasgupta, 2009 [237]*	UK	PMR	Cohort	2001
Khalifa, 2009 [238]	Tunisia	GCA	Cohort	1986-2003
Martinez-Lado, 2011 [239]	Spain	GCA	Cohort	1992-2006
Mazzantini, 2012 [240]	Italy	PMR	Cohort	1997-2009
Dunstan, 2014 [79]	Australia	GCA	Cohort	1991-2011
Alba, 2014 [241]	Spain	GCA	Cohort	1995-2007
Seror, 2014 [242]	France	GCA	RCT	2006-2010
Muller, 2016 [243]	France	GCA	Cohort	2002-2008
Carbonella, 2016 [244]	Italy	GCA	Cohort	NA
Faurschou, 2017[245]	Denmark	GCA	Cohort	1997-2015

Table 4.3: Studies included in the meta-analysis

RCT=Randomized controlled trial PMR=Polymyalgia rheumatica GCA=Giant cell arteritis

* Conference abstract

My systematic literature search identified 25 eligible publications consisting of 24 journal articles and 1 conference abstract. Of the final 25 studies, 21 were cohort studies and four were RCTs. Nine studies reported on predominantly-PMR patients while 16 studies reported predominantly-GCA patients. Nine of the 16 studies I classified as GCA studies included some patients with polymyalgic symptoms (range 37.3%-62.3% of patients). Four of the nine studies I classified as PMR studies including some patients who also had been diagnosed with GCA (range 13.0%-26.7% of patients), but I decided to classify them as PMR studies because that was the primary diagnosis as determined by the authors.

Most of the studies were conducted in Europe (n=20), followed by USA (n=3), Australia (n=1) and Tunisia (n=1). A total of 19 studies started enrolment before the year 2000, the earliest commencing study enrolment in 1950. Details of the individual studies are shown in Table 4.1.

	Total (n=3743)	PMR (n=920)	GCA (n=2823)	
Demographics				
Age at baseline*, years	74.1 (3.6)	71.6 (3.1)	74.9 (3.7)	
% Female	67.8 (10.6)	71.0 (10.7)	66.7 (10.5)	
Glucocorticoids use**				
Cumulative dose, g	7.6 (4.2)	5.6 (3.3)	8.2 (4.5)	
Duration of GC use, years	2.4 (1.5)	2.1 (1.2)	2.5 (1.6)	
Follow-up				
Duration, vears	5.9 (4.1)	4.4 (3.3)	6.4 (4.4)	

Table 4.4: Summary characteristics of individuals with PMR and/or GCA

All data are presented as a weighted mean (standard deviation) across studies

*Age at diagnosis (n=11), age at study inclusion (n=3), age unspecified in study (n=11) **Doses are shown as oral prednisolone equivalent

The weighted mean age across the studies reviewed was 71.6 years for PMR and 74.9 years for GCA, with approximately two-thirds being female. Cumulative dose in the GCA group was higher than the PMR group, with longer treatment duration and followup as well (Table 4.2).





Note: The W (random) shows the percentage weights of individual studies in the random effects model

The incidence proportion of patients with PMR who developed new-onset DM was 6% (95% CI: 3%-9%) and the incidence proportion of patients with GCA who developed new-onset DM was 13% (95% CI: 9-17%) (Figure 4.2). Of all the newly diagnosed DM reported, 7 occurred during follow-up and 18 were unspecified.

Covariates	OR	95% CI	p-value
Mean age (years)	0.97	(0.89, 1.04)	0.3010
Year of enrolment	1.01	(0.99, 1.04)	0.1405
Cumulative dose (g)	1.03	(0.89, 1.16)	0.6723
Duration of GC use (years)	1.17	(0.98, 1.41)	0.0948
Diagnosis			
PMR (Reference group)	1		
GCA	2.37	(1.64, 3.52)	<0.001

Table 4.5: Results from final multivariable model predicting risk of GC-induced DM

Note: Adjusted for mean age, year of enrolment, cumulative dose, duration of GC and diagnosis of PMR or GCA

In my study, the risk of developing new-onset DM among GCA patients doubled compared to PMR patients, which is consistent with the results of the meta-regression (13% vs. 6% respectively). All other variables, including mean age, year of study enrolment, cumulative GC dose and duration of GC use were not statistically significant.

My attempt to identify predictors of DM in this meta-analysis is more exploratory in nature as there are some limitations in the multivariable modelling. With this limited dataset and large number of potential explanatory variables, there is a risk of overfitting and may limit generalisability of this model. Other limitations of the model include the assumption that there is a linear relationship between variables, but it is also possible that collinearity may exist among some of the predictor variables. I excluded follow-up duration in my analysis because there is a high possibility that follow-up duration was confounded with diagnosis since GCA patients were more likely to receive higher GC doses and for longer duration, thus tend to be monitored over a longer period of time.

Table 4.6: Risk of bias of randomized controlled tria	ls
---	----

	Random sequence generation (Selection bias)	Allocation concealment (Selection bias)	Blinding of participants & personnel (Performance bias)	Blinding of outcome assessment (Detection bias)	Incomplete outcome data (Attrition bias)	Selective reporting (Reporting bias)
Jover JA et al, 2001	L	U	L	U	L	L
Salvarani C et al, 2007	Ĺ	L	Ĺ	L	Ĺ	U
Cimmino MA et al, 2008	L	U	L	L	L	U
Seror R et al, 2014	L	L	L	L	Н	U

L	Low risk of bias
Н	High risk of bias
U	Unclear risk of bias

Most RCT studies scored low risk in most domains. In some cases, however, we were unable to assess the risk of bias due to insufficient information (Table 4.4).

	Similar population	Assessment of exposure	Outcome not present at start of study	Adjustment of prognostic variables	Assessment of prognostic variables	Assessment of outcome	Adequate follow-up
Von Knorring J et al	++	+	-			-	+
Chuang TY et al	+	++			++	-	+
Godeau P et al	-	-				-	
Behn AR et al	++	++				-	+
Gouet D et al	+	+		-		I	+
Andersson R et al	++	++		-		I	+
Delecoeuillerie G et al	++	++		-		I	+
Nesher G et al	++	++	-	-	+	I	+
Gabriel SE et al	++	++	+	+	++	++	+
Proven A et al	++	++	+	+	++	++	+
Hutchings A et al	++	++		+	+	+	
Schmidt WA et al	++	++	-	+	+	I	-
Dasgupta B et al	+	-	-			-	+
Khalifa M et al	+	+				-	
Martinez Lado L et al	+	++		+	++		+
Mazzantini M et al	++	++	++	+	+	++	+
Dunstan E et al	++	++	+	++	+	I	+
Alba MA et al	++	++	+	++		+	+
Carbonella A et al	+						-
Farschou M et al	++	++	+	-	++	++	++
Muller G et al	++	+			+		-

Table 4.7: Risk of bias of observational studies

++	Definitely yes (Low risk of bias)
+	Probably yes
-	Probably no
	Definitely no (High risk of bias)

The overall risk of bias was high for many of the observational studies, especially for domains relating to the outcome and prognostic variables (Table 4.5). DM was not precisely defined in the studies and there was a lack of uniformity on how it was measured. Of the 25 studies, only six included an a-priori definition of DM. Even with these six studies, various measurement methods were used, ranging from random blood sugar estimations to fasting plasma glucose levels to use of pharmacological interventions such as oral anti-diabetic drugs and insulin. In 13 articles (52%), adjustment and assessment of prognostic variables were not well accounted for. The Kappa score for RCTs was 0.779, and 0.804 for observational studies. Approximately 56% of the disagreements for observational studies were between pairs of (1) "probably yes" and "yes" or (2) "probably no" and "no". When combined the "probably yes" and "yes" as one criterion, and "probably no" and "no" as another criterion, the Kappa score was increased to 0.871.

4.4. Discussion

There is overwhelming epidemiological and pathophysiological evidence that GC therapy may cause DM [246-250]. My aim was to estimate the effect size in this particular population for the purpose of informing clinical decisions about care of patients with PMR and/or GCA and health economic analyses about the costeffectiveness of new therapies for PMR and/or GCA. In my meta-analysis of published literature, the estimated incidence proportion (cumulative incidence) of new-onset diabetes was 6% (95%CI: 3%-9%) for patients with PMR and 13% (95%CI: 9%-17%) for patients with GCA. These figures are plausible: they are slightly higher than current UK population rates for patients of this age and sex, of which the expected background incidence rate of DM over 4.4 years in PMR patients and 6.4 years in GCA patients (follow-up duration) has been reported to be 4.8% and 7.0%, respectively [251]. It should however, be interpreted with caution as the incidence proportion was reported in my study, and not the incidence rate. Therefore it would be difficult to compare to estimates from other studies, as the person-years at risk was not accounted for in my study. It should be noted that many of the studies I reviewed were conducted at a time when population incidence of DM was lower than it is now [251]. In addition, a few studies [252-256], including a recent meta-analysis [257] have shown that GCA patients had a lower prevalence of DM at the time of GCA diagnosis compared to ageand sex-matched controls, which may suggest that the magnitude of GC-induced DM to be greater than expected. Two cohort studies, using Clinical Practice Research Datalink (CPRD) and The Health Improvement Network (THIN) data [258, 259] found no difference in pre-existing DM between GCA patients and their age and sex matched comparators, while another 2 studies [260, 261] reported a higher prevalence of DM in the GCA cohort as compared to their age and sex matched comparators. Heterogeneity in study design was high. The study populations were diverse in terms of disease manifestation, situated at different geographical locations, and were also subjected to different treatment strategies. In addition, the studies included in this meta-analysis were done over a span of 40 years, during which clinical practice is likely to have changed. This heterogeneity was also reflected in high statistical heterogeneity identified by our meta-analysis as assessed by the I-squared and tau statistics. One particular difficulty was the lack of clarity and consistency regarding the definition of DM in the studies identified. It was also difficult to identify the onset of DM, as most studies (72%) did not specify the timing of DM occurrence during follow-up.

Since most of the studies reviewed did not have the primary aim of quantifying DM risk in PMR and/or GCA, the detail available in published reports was limited. For example, summary measures such as mean starting dose, mean treatment duration and mean cumulative dose cannot fully capture the pattern of GC dosing used for PMR and/or GCA, where the highest GC burden occurs during the initial stages of treatment [21, 7, 262]. A very recent Danish study [245] reported that the incidence risk ratio of newonset DM was 7.0 (95% CI: 5.2-9.3) in the GCA cohort during the first year of observation when compared to the general population. Beyond the first year, they reported that the incidence rates for DM were not significantly increased. In another large observational study of 5,011 GCA patients, the incidence risk ratios of DM was 1.4 (95% CI:1.2-1.7) as compared to matched non-GCA patients [263]. The median time for the occurrence of DM in the GCA group was 1 year, which supports the hypothesis that the risk of developing GC-induced DM may be highest within the first year of GC use. Other studies emphasized cumulative dose: a very recently published US study reported that the risk for new-onset DM rose by 5% with each 1000mg of GC exposure (in prednisolone-equivalent dose (PED)) in GCA patients [264]. It was difficult to explore these factors further in the meta-analysis due to limited information and heterogeneity in population, geographical location and treatment practices. The addition of my PhD project to the meta-analysis may help provide a better estimation of DM risk as it has a large sample size and designed with the primary aim of quantifying DM risk associated with GC use in PMR and/or GCA patients.

Because GCs are the mainstay of treatment for PMR and GCA, it was not possible to disentangle the effect of the disease from the effect of the treatment. It is however known that systemic inflammation itself can also induce a state of insulin resistance [265, 97] so it is plausible that the inflammatory disease itself (PMR or GCA) could have contributed to the risk of new-onset DM. In addition, some medications commonly prescribed to elderly patients may contribute to the risk of DM (e.g. thiazide diuretics, beta-blockers, niacins and statins). These were not reported by the studies identified, as their primary focus was not on DM.

Confounding by indication could not be excluded. For example for the observational studies, clinicians may have been less willing to prescribe higher GC doses to control disease activity in obese patients at high risk of DM.

The overall risk of bias was high for many of the observational studies, especially for domains relating to the outcome and prognostic variables. Therefore, results should be interpreted with caution. One of the potential next steps is to carry out a sensitivity analysis that includes only selected studies with a low risk of bias and determine if the results differ from the main analysis.

4.5. Conclusion

Findings from this study underline the importance of screening for GC-induced DM in patients with PMR and/or GCA in clinical practice [21, 7] and can also help inform dietary and lifestyle advice in patients commencing GC for PMR and/or GCA. As well as limitations inherent to the meta-analysis itself, there remains considerable uncertainty in our estimate of the absolute risk of DM in PMR and/or GCA, since most published studies were not conducted with this as the primary aim. Furthermore, there is virtually no direct evidence as to which patients are at the greatest risk of DM, which would inform decisions as to how treatment should be individualized. To address this issue, I will be discussing in detail the main focus of my PhD project in the next chapter, which is to quantify the incidence of type 2 DM associated with GC use in patients diagnosed with PMR and/or GCA using CPRD data.

CHAPTER 5: METHODS

5.1. Overview

This chapter seeks to tackle the aims set out in chapter one through the use of Clinical Practice Research Datalink (CPRD). There were three aims in this study. Firstly, to quantify diabetes mellitus (DM) risk associated with oral glucocorticoid (GC) use in primary care within the initial two years of polymyalgia rheumatica (PMR) and/or giant cell artertis (GCA) diagnosis. Secondly, to assess the prescribing pattern of oral GC use in patients with PMR and/or GCA in a primary care setting; and thirdly, to compare CPRD-derived data with data I extracted from primary care prescription data collected directly from registered primary care general practitioners (GPs). The first two chapters provide the basis on why it is crucial to quantify the risk of DM associated with oral GC use in patients with PMR and/or GCA. Patients with PMR and/or GCA are not only often prescribed with high GC doses, but also for a considerably long duration compared to other GC-treated diseases like asthma. Chapter three describes how GCs are like a double-edged sword - while GCs ameliorate symptoms, they also induce toxicities such as DM. Oral GCs are almost exclusively prescribed in PMR and/or GCA; therefore these patients are considered to be high risk of developing toxicities over Chapter three also describes in detail the pharmacokinetics time. and pharmacodynamics of GCs, including the relationship between dosing regimens and therapeutic or detrimental effects. Chapter four provides an estimate of the risk of DM in PMR and/or GCA patients based on published literature. This chapter also provides the rationale for the importance of using large electronic health records data such as CPRD to provide a better estimate of DM risk associated with GC use. The current chapter (chapter 5) will include a section on ethical considerations, followed by a brief discussion on the different data sources, which includes how data were collected and linked to provide the dataset from which the study population was derived from. The identification of study population and outcomes will also be discussed. Final sections will outline key steps in data preparation and the statistical analyses.

5.2. Ethical Considerations

5.2.1. Ethical Approval

CPRD is a UK government research service supported by both the Medicines and Healthcare products Regulatory Agency (MHRA) and the National Institute for Health Research (NIHR) to promote healthcare research through the use of electronic health records. All CPRD-related studies have to be approved by the Independent Scientific Advisory Committee (ISAC) for MHRA database research, of which the following aspects are of priority: a well-defined hypothesis; clear application of methodology, including consideration of possible bias and confounding; as well as compliance with all requirements to ensure patient confidentiality. All electronic patient records had been anonymized, therefore no patient identifiable data were available in the dataset other than the month and year of birth. Ethical approval for this study had been as part of my supervisor's (Dr Mar Pujades-Rodriguez) project approval (Approval number: 16_146; included as Appendix 3).

5.2.2. Legal Framework

In addition to ethical approval, to comply with the General Data Protection Regulation (GDPR) and ensure that participants' confidentiality and privacy were well protected, the linked CPRD database is stored securely on a university password protected drive, and has restricted access to named researchers. One of the practical steps I did was to familiarize myself with the relevant legal and ethical requirements before I started working on the dataset. I completed multiple ethics-related modules (e.g. Good Research Practice, Good Clinical Practice on Secondary Care, Research Data and Confidentiality and Ethics Reviewers Training) to ensure that I fully understood the guidelines and standards. I took appropriate steps to work within these frameworks (e.g. only publish data which is non-patient identifiable) to ensure data security and protect patient confidentiality.

5.3. Data Sources

5.3.1. Clinical Practice Research Datalink (CPRD)

Approximately 98% of the UK population is registered with a primary care general practitioner (GP) [266], who acts as the first point of contact for patients for any healthrelated issues. Patients may then be referred to other services such as hospital care as necessary. At each visit, patient data are routinely recorded electronically, and if they are registered with a CPRD-participating practice, then the information will be uploaded to the CPRD secure servers on a monthly basis unless the patients request to opt out of data sharing at their respective GPs. CPRD is one of the largest ongoing longitudinal databases with approximately 11.3 million patients (reported in 2013) from over 600 general practices across England, Wales, Scotland and Northern Ireland. This database mainly consisted of GPs from England, of which London had the highest number of patients (13.6%), followed by the North West region (11.1%), and the South East Coast (10.0%) [9]. Approximately 75% of English practices (as of 2013) were part of the CPRD linkage scheme and provided patient-level information. Patient-level data were then linked to other data sources such as Hospital Episode Statistics (HES) [267] and Office of National Statistics (ONS) mortality registry [268] and Index of Multiple Deprivation (IMD) [269].

Information recorded includes demographics, medical diagnoses, signs and symptoms, drug prescriptions, referrals, clinical and laboratory investigations and administrative dates (e.g. registration, de-registration and death). Read codes, a coded thesaurus of clinical terms used by the National Health Services (NHS) during the study period, were used to classify medical information, while product codes (based on the British National Formulary (BNF)) are assigned to prescribed medications [270]. In addition to these codes, there are also GP-uncoded notes, often entered as free text. These free texts are not available to researchers as they often contain identifiable information. Validation studies have reported a high positive predictive value of reported diagnoses and prescriptions in this database [271, 272].

The strengths of CPRD data include the following: It is one of the largest routinely collected health data that spanned primary care practices across UK and linked to the mortality registry, with a relatively long follow-up duration (median follow-up of 9.4 years for active patients and 5.1 years for the overall CPRD population)[9]. It is also reported to be broadly representative of the UK population in terms of age, sex and ethnicity [9].

CPRD is however, not without limitations. One of the key limitations is missing data -Information may be missing because it is not compulsory to have it recorded. For example, blood pressure is not routinely measured at every GP visit, unless there is an indication that it needs to be measured. Another limitation of CPRD is the incompleteness or delayed recording of information from secondary care as this information need to be manually entered into patient record. The variability in completeness of data across patients and across time would therefore require careful consideration to ensure that the study is not biased due to these limitations.

5.3.2. Hospital Episode Statistics (HES) [267]

Hospital Episode Statistics (HES) is a data warehouse for all inpatient admissions for NHS funded patients, which goes as far back as 1989. Clinical information about diagnoses has been linked to CPRD, of which data have been collated centrally and coded by administrators. Diagnostic data recorded in HES are coded using the International Classification of Diseases (ICD)-10 coding classification.

5.3.3. Death Registration Data by Office for National Statistics (ONS) [273]

The Office for National Statistics (ONS) mortality data contains information related to a person's death obtained from the death certificate for all deaths registered in England and Wales. It includes information on the underlying cause of death and date of death using ICD-9 (for information prior to 2000) and ICD-10 codes (for information after 2000).

5.3.4. Index of Multiple Deprivation (IMD) [274]

The IMD is a measure of relative deprivation for neighborhoods in England. It is used to compare small areas across England, identify the most deprived areas and explore the various domains of deprivation. The seven domains of deprivation are: income deprivation, employment deprivation, education, skills and training deprivation, health and disability, crime, barriers to housing and services, as well as living environment deprivation [275]. The small area data provided by CPRD is at the lower layer super output areas (LSOA) level which are typically built from 4-6 output areas with an average of 1,600 residents [276]. The relative deprivation of a particular area in England is determined by ranking each area from most deprived to least deprived. The IMD variable in CPRD has been divided into 20 groups to prevent disclosure of patient

location. The raw data of LSOA used to be made available to researchers in the past, but it is no longer available.

5.4. Data Linkage [273, 277]

NHS Digital is a trusted third party with the responsibility and authority to link identifiable data between CPRD, HES and ONS across England. Linkage is crucial to provide a more comprehensive and complete set of data needed for quality research. Linking ONS and CPRD creates a richer dataset that captures the date and the underlying cause of death of patients registered in the practices. The IMD is linked to CPRD data through the patient postcode, which serves as a useful proxy for socio-economic status.

5.5. Study Population

There are three levels of population hierarchy [278]. First, there is the "source population", which comprises all individuals who are registered in a GP practice in the UK. A subset of this population is the "database population", comprising all individuals included in the CPRD database. The smallest subset is called the "study population", of which individuals are selected from within the CPRD database using codes and algorithms based on the specific inclusion and exclusion criteria of the study. For the purpose of this PhD project, the term "study population" will subsequently be used.

Patient records for my study population were extracted from the CPRD database based on standardized diagnostic codes (Read codes); and for drug exposure, the BNF drug product or substance codes were used. Based on these diagnostic and prescription codes, relevant electronic health records were obtained. The data were then extracted as a number of separate data files, of which the information was linked to individual patients via a unique patient identifier – the patient ID. The unique patient ID is present in all files extracted and allows all the information to be linked to one specific patient. Table 5.1 provides details of the CPRD data files used in this project.
File Type	Data Description
Cohort	Demographic data including date of birth, gender, IMD, registration date
Clinical (1)	Clinical diagnosis of PMR and/or GCA, including diagnostic codes and dates
Clinical (2)	Clinical diagnoses of diabetes, including different types of DM codes and dates
Prescription (1)	GC drug prescriptions, including drug code, dates, strength, formulation
Additional	Contains information on smoking status and body mass index (BMI)

Table 5.1: Description of CPRD data files obtained

A clear set of pre-defined inclusion and exclusion criteria were then applied to identify the study population and the start of follow-up for each patient.

5.5.1. Inclusion Criteria

All patients who had:

- A first ever diagnosis of "polymyalgia rheumatica", "giant cell arteritis", "temporal arteritis" or "horton's disease" recorded in CPRD or HES between 1st January 1998 and 30th September 2015. I chose to only look at incident cases; therefore only patients with a first occurrence of PMR and/or GCA were included
- At least twelve months of CPRD-defined up-to-standard (UTS) data available prior to the start of follow-up to rule out any prior diagnosis of PMR or GCA (the "up-tostandard" date is a practice-based quality metric based on the continuity of recording and number of recorded deaths. The UTS date is calculated for each participating practice, corresponding to the latest data at which practices meet these minimum quality criteria. [9])
- At least one oral GC prescription during the duration of study follow-up

5.5.2. Exclusion Criteria

- Patients with pre-existing PMR and/or GCA (i.e. prevalent disease)
- Patients with a diagnosis of DM in CPRD or HES recorded before their PMR or GCA diagnosis
- Patients without any oral GC prescription during the duration of study follow-up
- Less than one year of up-to-standard registration in CPRD
- Under 18 years of age at the date of the first recorded PMR/GCA diagnosis
- Not eligible for data linkage due to one or more of the following reasons:

- Registered at the practice after the transfer of encrypted unique patient identifiers from NHS digital to the end-user
- No valid identifier for linkage (either NHS number or postcode)
- o Opted out or dissented from CPRD or the linkage scheme

5.5.3. Definition of Start and End of Study Follow-up

The study entry date was created, defined as the latest date of the following:

- 1st of January 1998
- Date of first recorded diagnosis of PMR and/or GCA
- UTS date plus one year
- Date of GP practice registration plus one year.
- At least 18 years of age

The end of follow-up date was determined, defined as the earliest of the following dates:

- End of data collection (date of last collection date for the practice or date of transfer of the patient out of the practice)
- Death or
- DM diagnosis

The date of death was identified from the primary care data CPRD or linked ONS mortality data, where available. For patients who had a recorded death in both CPRD and ONS, the date of death from ONS was selected as they were taken from the death certificate for all registered death.

For the majority of my analyses pertaining to DM risk and oral GC exposure, I looked at the initial two years of follow-up following diagnosis of PMR and/or GCA. When studying the overall GC prescribing pattern in primary care however, I looked at the starting dose, daily dose, cumulative dose and treatment duration relating to overall follow-up duration, and comparing the results from my study to other published literature.

5.5.4. Factors of Interest

5.5.4.1. Exposure

The main exposure for this PhD project was prescribed oral GC dose in primary care, and the first step was to prepare and clean the prescription data. There were 2 sets of prescription data – one being the CPRD derived data managed by the CPRD team; and the other being data that I extracted from primary care data collected directly from registered primary care GPs and linked to CPRD. This is preliminary work to compare the information derived from CPRD with my extracted data to assess the quality and completeness of CPRD derived data, which is often used by researchers in the analyses performed in many publications. All subsequent analyses were done using my extracted variables, by applying rules described in other studies and that others have also used in the past [12, 279]. These rules are detailed at the end of this section (pages 63 and 64).

The prescription database contains information on GCs in various dosage forms, which has been categorized based on their route of administration: oral, intramuscular (IM), intra-articular (IA), rectal, topical, inhaled, and nasal. For the purpose of this PhD project, I decided to only include oral prescriptions, as intravenous GC prescriptions are almost never prescribed by GPs for these diseases.

Information pertaining to the dose, frequency and duration of each oral GC prescription was then extracted from the following variables:

- "Product name" This variable documents the names of all GCs prescribed in primary care
- "Quantity prescribed" This variable documents the quantity of oral GC prescribed for each visit
- "Dosage instructions" This variable contains free text as documented by GPs in primary care. Most of the information relating to dose and frequency were extracted from this variable
- "Duration of treatment" This variable documents the duration of GC treatment prescribed

The various types of oral GCs with different anti-inflammatory potencies in the prescription database were identified from the product name, and the potencies of these prescribed GCs were standardized according to the most commonly used GC, prednisolone, to obtain the prednisolone-equivalent dose (PED), using the potency equivalence listed in Table 5.2.

Glucocorticoid (GC)	Prednisolone-equivalent Dose (PED)
Prednisolone	5mg
Betamethasone	750mcg
Deflazacort	6mg
Dexamethasone	750mcg
Hydrocortisone	20mg
Methylprednisolone	4mg
Prednisone	5mg
Triamcinolone	4mg
Budesonide	5mg
Cortisone	25mg

Table 5.2: Prednisolone-equivalent doses (PED) anti-inflammatory doses [280]

Next, information on the frequency of GC use was extracted from the "dosage instructions" variable. Common key phrases were identified to classify the data into the different categories as listed in Table 5.3.

Table 5.3: Frequencies of oral GC administration

Frequencies	Examples of abbreviations / terms used in CPRD
Once a day	OD, ON, OM, once a day, everyday, daily
Twice a day	BD, BID, 1 MANE 1 NOCT, twice daily, 1 every 12 hours
Three times a day	TID, TDS, three times a day
Four times a day	QID, QDS
Five times a day	5X daily
Six times a day	6X per day
Every other day	EOD, every second day, alternate days
Weekly	Once a week, weekly, on day 6, every Monday
Twice weekly	Every third day, every 3 days, twice weekly
Three times a week	Three times a week, three times per week
Tapering	reduce by, 6/5/4/3/2/1
When needed	PRN, as required, when necessary
Immediately	Stat, now, at once
As directed	As directed, as advised, as agreed
Unknown	-,, ~, null

In cases where only the number of tablets per dose was given (e.g. 2 tablets), but not frequency, then it was classified as "unknown". The new frequency variable was then screened and rechecked manually to ensure that there were no errors.

The duration of each prescription was determined from the "duration of treatment" variable. If the duration of treatment was mentioned in the "dosage instructions" variable but not given in the "duration of treatment" variable, then the information was transcribed into the "duration of treatment" variable for use in the analysis. In cases where the duration information was not given, the following approaches were used:

 Duration was determined by dividing the total quantity of GC tablets prescribed with the number of tablets per day. The number of tablets per day was determined by extracting information on the number of tablets per dose and frequency from the "dosage instructions" variable.

Formula: Duration (days) = Quantity prescribed / Number of tablets per day

 If the information in (1) was missing, then the duration was determined by calculating the difference in days between 2 consecutive prescriptions for each patient.

Formula: Duration (days) = Date of Prescription 2 – Date of Prescription 1 (sorted by individual patients and date)

The treatment stop date was not given. As most prescriptions in primary care are usually prescribed for a period between 28 to 90 days, therefore, all prescriptions with a duration of more than 90 days were capped at the 90 days limit. Any missing durations were imputed with the median duration of all prescriptions in the dataset.

The next step in the data cleaning process was to apply some assumptions based on clinical practice – for example, if no information were provided for the daily dosing frequency, the default would be once a day as GCs are usually administered once a day. Similarly, if no information were given for the number of tablets per dose, then the default would be one tablet per dose. All "mg" and "ml" related dosing instructions were also scrutinized to ensure that they corresponded to the correct number of tablets/ oral solutions.

Next the "number of tablets per day" variable was determined according to the following rules:

- Total quantity prescribed divided by the duration of each prescription
 Formula: Number of tablets per day = Quantity prescribed / Prescription duration
- If the information in (1) was missing, then the number of tablets per day was determined by multiplying the number of tablets per dose with the daily frequency.
 Formula: Number of tablets per day = Number of tablets per dose * Frequency

The daily dose for each patient was then calculated by multiplying the number of tablets per day with the conversion factor for the calculation of appropriate PED and product strength. For subsequent analyses, I prepared the dataset in an interval format, of which the daily GC dose for the initial two years of follow-up was determined using loops in R.

5.5.4.2. Outcomes

The primary outcome of this study was DM. Patients with type 1 DM, defined as those who had a recorded Read code of type 1 DM, or were not on any oral anti-diabetic medications but merely prescribed insulin, were excluded from the study. For all patients who had a Read code or ICD-10 of any type of DM (other than type 1), the earliest date of any diabetes-related medications prescribed or any type of DM diagnosis received was set as the index date. The study population was then categorized into 3 groups, those with DM, no DM and pre-existing DM.

For this PhD project, I studied the effect of GC cumulative dose on DM in the initial two years of follow-up for the following reasons: firstly, the average duration for the treatment of PMR and GCA was reported to be approximately 2 years in previous studies [12, 11], therefore any DM occurring after that might be less likely to be GC-related. While it is possible that DM may have a delayed occurrence, it is unlikely considering the short elimination half-life (approximately two to four hours) of GCs, so the drug is likely to be cleared out of the system within a day (approximately five half-lives). Secondly, other studies have reported that the onset of GC-related DM was likely to be within the first two years of GC treatment [245, 281].

5.5.4.3. Baseline Characteristics

The next step of data management was to determine the baseline characteristics of the study population. Age at baseline was calculated by subtracting the date of birth of each individual from the study entry date. The age at baseline of my study population was also the age at the time of PMR and/or GCA diagnosis, as my cohort only included newly diagnosed patients. The ethnicity variable was re-grouped into 2 major groups: white and others (which includes mixed ethnicity, Asian, black etc) and a missing category were added to identify patients with unknown ethnicity.

I also looked at various baseline DM risk factors for type 2 DM, including smoking, BMI, and family history of DM. BMI was calculated by dividing weight in kilograms by height in meters squared. Some individuals had missing values for heights on some visits, so the height recorded in any other visits was used for all the visits with missing height data, as height for adults is unlikely to change over time, provided that the patient was at least 18 years old at the time of the measurement. For BMI, any value recorded before or on the date of study entry and within 1 year prior to study entry was considered to be BMI at baseline. The average of all values was calculated and labeled as BMI at baseline. In addition, I also categorized BMI into 2 groups: those who were obese (BMI >/= 30kg/m^2) and those who were not obese (BMI < 30kg/m^2) or did not have any recorded BMI. The reason for including those who were not obese with those who did not have any recorded BMI was based on the assumption that those without any recorded BMI were more likely to have a normal BMI. It has been reported that underweight and overweight individuals were more likely to have their BMI recorded in primary care [282], thus contradicting the missing at random assumption. Another study reported that 97% of patients with a record of type 2 DM had a recent BMI recorded in CPRD [283], further supporting the possibility that those who did not have any recorded BMI were likely not obese as obesity is a known risk factor for type 2 DM. For smoking status, the smoking status recorded before or on the date of study entry and within 1 year prior to study entry was considered as baseline smoking status. For patients with multiple smoking statuses, the highest level of exposure was determined in the following order: current smoker, former smoker and non-smoker. All smoking status codes recorded at baseline and before the year prior to entry were checked and the highest level of exposure (ranked in the following order: current smoker, former smoker and non-smoker) was selected. Similarly, all smoking status codes recorded in the year after the date of entry were checked, and the highest level of exposure was chosen as post-baseline smoking status. Unlike BMI that is likely to vary over time, smoking status is less likely to change drastically over time; therefore for patients without any smoking information at baseline, the method of last observation carried forward (from pre-baseline) or backwards (from post-baseline) was used. For the rest of the study population without any smoking information, their smoking status was categorized as unknown. Since family history is a strong risk factor for DM, I also included this variable in my baseline characteristics. Lastly, any non-oral GC use during the study follow-up duration was also determined. Both family history of DM and non-oral GC use during follow-up are potential confounders that were adjusted for in later analyses.

5.5.5. Missing Data

Since information had already been collected, the quality of the study is reliant on the completeness of the data collected. Missing data are problematic for all observational research, particularly in routinely collected data [284]. It may result in selection bias if variables needed to define the study cohort are not missing at random, or if identifiers required for linkage are missing. The way on how missing data should be handled is challenging – restriction to those with only complete data may introduce bias, while multiple imputation may also not be appropriate because the pattern of missingness may not be missing at random. For example, approximately two thirds of BMI data were missing in my study; but BMI have been reported to be recorded more frequently in patients at risk of developing DM as compared to the general population [282]. This suggests that data may not be missing at random; therefore multiple imputation may not be appropriate in this case. In this case, BMI was only reported in the baseline characteristic table and not used in any of the other analyses, therefore I did not do any imputations.

For PMR patients with missing prescription durations, the median duration of all prescriptions for PMR patients was used. The same rule was applied for patients with GCA or with both PMR and GCA. Other studies have used a different approach - one study replaced any missing prescription duration with the average of that patient's duration for other prescriptions of the same drug with the same strength (if present); or the average duration of all other patients' prescriptions of the same drug with the same strength [12]. There are however, some limitations with this approach as the prescription duration of each patient may differ over time according to different circumstances. For example, some patients may be prescribed with longer duration of GCs at different times of the year as they travel for vacation. Using the average duration of all other patient's prescriptions of the same drug has its limitations as well, as some patients with limited mobility may have been prescribed with longer durations

to minimize the number of visits needed. Since there is no one best method to account for missing prescription durations, the median imputation was used, as it is less susceptible to any outliers that the other methods may present with. Missing data for dosing frequency and number of tablets per day have been discussed in detail in the preceding section (Section 5.5.4.1). I did consider doing a sensitivity analysis using complete case analysis, but it would be unlikely to obtain a large enough sample of cases with no missing data as the main reason for missing data in the prescription database was because all patients were managed in secondary care rather than primary care, of which information relating to secondary care were unavailable.

5.6. Statistical Analysis

5.6.1. Descriptive Analysis on Baseline Characteristics

Baseline characteristics of the study population were summarized using descriptive statistics – median and inter-quartile range (IQR), or mean and standard deviations (SD) for continuous variables, and frequency, percentage for categorical variables. For continuous variables with a skewed distribution (e.g. age and follow-up duration), I used the median and IQR; while for variables with a normal distribution (e.g. BMI), I used the mean and SD. Baseline characteristics were presented for patients with PMR only, GCA and PMR/GCA, as well as for the overall study population. I chose to analyze PMR patients as a stand-alone group, and categorize those with GCA and both PMR and GCA as another group. Patients presenting with both PMR and GCA usually receive the same high-dose GC regimen as patients with GCA only, with the rationale that they are considered to be at a similar risk of vascular complications. Therefore, the impact of GC treatment on the risk of DM would likely be comparable. In addition, grouping them together will also add to the robustness of statistical analyses, as the number of patients with both PMR and GCA was relatively small compared to those with PMR alone or GCA alone.

5.6.2. Descriptive Analysis on Prescription Database and Prescribing Patterns

Information from the prescription database was also presented using summary statistics, including the types of oral GC prescribed in primary care and dosing frequencies of oral GCs. The percentages of missing data for key variables needed for the calculation of GC dose were also determined and compared with the CPRD derived dataset. I also calculated the total duration of treatment, cumulative dose, first dose, monthly starting dose and average daily dose. The total duration of treatment was calculated as the sum of all prescription durations for each patient. The cumulative dose was calculated as the sum of quantity prescribed, multiplied by strength and PED for all prescriptions for each patient. The first dose was the total dose of the first prescription of each patient, while the monthly starting dose was calculated as the total patient. The average daily dose was calculated as the total duration of treatment for each patient.

I also looked at the prescribing pattern of oral GCs for PMR and/or GCA patients in primary care over the initial two years of PMR/GCA diagnosis, presented in the form of total monthly oral GC dose, high dose (75th percentile) versus low dose regimen (25th percentile) and average monthly oral GC dose. A comparison between real-world GC prescribing pattern in the initial twelve months of treatment and European League Against Rheumatism (EULAR) recommended guidelines was also made for these patient populations [22, 21].

5.6.3. Kaplan Meier Survival Curves

Survival analysis, also known as the time to event analysis, focuses on the distribution of survival time and models the time it takes for events to occur within a specified time interval. For my PhD, the "event" was newly diagnosed DM, and the time interval was specified to be within the first two years of PMR and/or GCA diagnosis. There are several methods to estimate a survival (or hazard) function, of which the Kaplan-Meier (KM) curve is one of the most used methods. For the KM curve, there are no assumptions about the baseline hazard distribution. Therefore, it imposes the least structure and is relatively easier to estimate and interpret. However, it is mostly descriptive in nature as it is difficult to incorporate predictors. The log rank test is often used to test the null hypothesis of no difference in survival between groups. In other words, it can be thought as a test of measuring whether the survival curves are identical (or overlapping) or not. It is conducted by assigning a "1" to the event of interest (e.g. DM) and a "0" to all competing risk events and censored observations. A

p-value of more than 0.05 indicates that the two survival curves are different and the null hypothesis should be rejected.

For my PhD, I used the survival Kaplan Meier methods and log rank test to compare the risk of DM in patients with PMR and/or GCA, stratified by underlying disease, gender and age group. Age group was categorised using tertiles, quartiles and quintiles.

5.6.4. Extended Time-varying Cox Model for Cumulative Dose

Survival analysis can also be extended to assess several risk factors or exposures simultaneously, in relation to survival time. One of the most popular techniques for this is the Cox proportional hazard regression. In a Cox proportional hazard regression model, the measure of effect is the hazard rate, which is the risk of failure (or the event of interest such as DM), given that the participant has survived up to a specific time [285]. The Cox proportional hazard model is also called a semi-parametric model, because there are no assumptions about the shape of the baseline hazard function, but there are, however, several important assumptions for appropriate use of the model. The assumptions include independence of survival times between each individual in the cohort, a linear association between the natural logarithm of the relative hazard and predictors, and that changes in predictors produce proportional changes in the hazard regardless of time.

One of the major limitations of the Cox proportional hazard model is that the hazard ratio is averaged over event times. In other words, when estimating the overall hazard ratio over the follow-up duration, the same weights are given to the very early hazard ratios that affect almost all individuals, as are given to very late hazard ratios affecting only the few individuals still at risk. However, the guidelines recommend starting at a high GC dose and tapering this dose down over time. DM risk associated with GC therapy is thought to be higher at higher GC doses. Therefore, use of the extended time-varying Cox is more appropriate as it allows incorporation of covariates that change over time into the model. It allows the comparison of the current GC dose of each patient who had the event to the GC doses of all others who were at risk at each event time [286].

The main objective of this PhD was to explore the association between GC dose exposure during the first two years of treatment and the risk of developing DM, of which the following variables were adjusted in the model: family history of DM and use of non-oral GC during the study duration. The goal was to use a parsimonious model with a minimum number of parameters; therefore only variables that were clear confounders were included. Age and BMI were not included in the model because while they may be associated with the risk of developing DM, they were unlikely to be associated with the use of GC for the treatment of PMR and/or GCA. In addition, slightly more than half of the patients did not have a recorded BMI, and while the use of multiple imputation could have been used to address the issue of missing data, it may not be appropriate for this study as studies have reported that BMI data were not likely to be missing at random in primary care [282, 283]. The use of complete case analysis would have resulted in a much reduced sample size.

Non-linearity of GC dose was tested using the ANOVA analysis of deviance test by comparing two models, one with and the other without the quadratic term. The best fitting model was selected.

I used two different approaches to examine the association of oral GC cumulative dose and DM using the extended time-varying Cox model. Firstly, the risk of DM was quantified based on each gram increase in cumulative GC dose within the initial two years of PMR/GCA diagnosis. Secondly, the risk of DM was quantified using survival percentiles [287], of which patients who were on high dose (using the upper quartile or 75th percentile) were compared with those who were on low dose (lower quartile or 25th percentile) within the initial two years of PMR/GCA diagnosis.

There were a few reasons why I chose to compare the risk of DM between those on high GC dose versus low GC dose. Firstly, since my study only includes newly diagnosed PMR and/or GCA patients, all of them would have been on GC therapy, thus I was unable to compare the risk with patients not on GC therapy. Secondly, the Interpretation of survival percentiles is easier and more intuitive, thus providing additional value and facilitates the translation of results from epidemiological studies into clinical practice. Another advantage is that it is well suited to evaluate situations in which the association of interest is changing over time. In other words, the association between varying GC doses with the risk of developing DM can be evaluated over time through focusing on the selected percentiles of observed distribution of events, without necessarily assuming a constant effect.

5.6.5. Rolling Cumulative Dose Model

While the extended time-varying Cox model allows the incorporation of GC cumulative dose to vary over time, it does not account for the timing of GC exposure. In other words, it does not indicate whether the more recent GC doses had a more significant effect on the risk of developing DM as compared to doses from a distant past. To explore whether the timing of GC exposure actually affects the risk of developing DM, I used the "rolling window calculations", of which rolling calculations of cumulative GC dose were applied to a fixed width subset of the data (also known as the window). The rolling cumulative dose method allowed visualization of how the impact of cumulative dose changes over time, and by varying the window (in this case, months following the diagnosis of PMR and/or GCA), I was able to determine the timing of the most recent GC exposure that had a largest impact on risk of DM. The best fitting model or optimal rolling window timeframe was determined by comparing the concordance index, of which the highest concordance index was chosen. The coefficient of that best fitting model was determined and applied to calculate the hazard ratios for patients with high GC dose versus low GC dose over the initial two years since PMR/GCA diagnosis.

5.6.6. Weighted Cumulative Exposure (WCE) Model

While the rolling cumulative dose analysis is a useful method to determine the risk based on recent cumulative doses, it has its limitation as the time-dependent hazard is fixed for each time window. To address this limitation, I used another statistical method, namely the WCE analysis that allows the assessment of risk based on the dose, duration and timing of treatment.

Breslow et al [288] and Thomas [289] first introduced and discussed the concept of weighted cumulative exposure (WCE) that combines information about dose, duration and timing of exposure into a summary measure. Vacek [290] then proposed a parametric modeling of the weight function in case-control studies, which was then further developed by Abrahamowicz et al [291], who used the parametric WCE framework within the Cox's proportional hazard model to refine the assessment of the associations between exposure and outcome. However, modeling strategies that impose a specific parametric form of the weight function in the absence of solid prior knowledge about its shape may lead to invalid results if the function is incorrectly specified [290]. With that in mind, Sylvestre et al [292] proposed the regression spline-based method for modeling the WCE as a time-dependent covariate in the Cox's proportional hazard regression analysis of cohort studies. Splines are flexible

mathematical functions defined by piecewise polynomials joined at points on the x-axis known as knots. Regression splines are particularly useful because they can be incorporated into any regression model that has a linear predictor, and they are also relatively simple but still have enough flexibility for most practical data [293].

A WCE analysis models the time-dependent hazard as a smooth flexible function of time. Taking the time-varying use of GC as an example, X_{GC} (t) is the dose of GC at time t. The joint effect of past doses of GC at the index date u is given by the time-varying WCE metric [292]:

$$WCE(u) = \sum_{t}^{u} w(u-t)X(t)$$

where $t \le u$ indexes times of GC exposure preceding u and where w(u-t) is the function assigning weights to past GC daily doses based on the time elapsed since the last dose was taken. In other words, at any point during the follow-up, the cumulative dose will be calculated as a weighted mean of the past doses, with higher weights assigned to more recent doses. The shape and the values of the weight function w(u-t) are unknown but they can be estimated from the data by the metric below:

$$w(u-t) = \sum_{j=1}^{m} \theta_j B_j(u-t)$$

where Bj, j=1,...,m, represent the m functions in the cubic spline basis and θj , j=1,...,m, represent the estimable coefficients of the linear combination of the basis splines. By using the flexible cubic B-splines, the model avoids the need to specify a-priori the analytical form or shape of the weight function. The WCE weight plots over time provide great clinical insight on the etiologically pertinent time window of past exposure and on the relative importance of doses taken at different time periods [294, 250, 291, 295]. This is particularly useful to help unravel the possibly complex mechanisms linking drug exposure to adverse events and help clinicians decide on how treatment regimens may be optimized to improve the risks versus benefits ratio in clinical practice.

The best-fitting WCE model was selected based on the lowest Bayesian Information Criterion (BIC) value. As with all the other analyses, the hazard ratio for the best fitting model was compared between those on high dose versus those on low dose over the initial two years since PMR/GCA diagnosis. All analyses were carried out in R Version 3.3.1. Results from my analyses will be presented in the next chapter.

CHAPTER 6: RESULTS

6.1. Overview

This chapter reports the results from the analyses of the inception cohort together with a brief description of the patient selection process, followed by the baseline characteristics of newly diagnosed patients, stratified by disease status. The next subsection will include a detailed description of the prescription database, followed by oral glucocorticoid (GC) prescribing patterns in primary care compared with European League Against Rheumatism (EULAR) guidelines. The final sub-section will report the quantification of diabetes mellitus (DM) risk associated with oral GC dose using the extended time-varying Cox, rolling cumulative dose and weighted cumulative exposure models.

6.2. Patient Selection Process

Figure 6.1 shows the flow of patients through the data extraction process. Approximately 29.9% of the patients were excluded due to patient related factors, of which 11.9% had pre-existing DM, including type 1 DM, type 2 DM, secondary DM, unspecified DM and steroid induced DM. As this was an inception cohort, all patients with a pre-existing diagnosis were excluded from the study, accounting for approximately 18.0%. Another 11.6% of patients were excluded due to treatment related factors, of which 3.8% never had any GC prescriptions. Approximately 4.6% had no recorded prescriptions during the study follow-up duration, while another 3.2% had no recorded oral GC prescriptions, thus they were all excluded from the study. A total of 23048 patients were included in the final cohort.



6.3. Baseline Characteristics

Patient Characteristics	All (n=23048)	PMR (n=19204)	GCA with or without PMR
			(n=3844)
Age at diagnosis of PMR/GCA, vears			
Median	74.1	74.3	73.2
(IQR)	(66.8-80.2)	(66.9-80.3)	(66.2-79.6)
Follow-up duration, years*		()	()
Median (IQR)	4.5 (2.0-7.8)	4.5 (2.1-7.8)	4.4 (1.9-8.1)
Gender, n (%)	· · · · ·	,	,
Females	15706 (68.0)	12908 (67.2)	2762 (71.9)
Ethnic group, n (%)		. ,	
White	20746 (90.0)	17211 (89.6)	3535 (92.0)
Others	401 (1.7)	297 (1.5)	104 (2.7)
Index of multiple deprivation (IMD),			
quartiles, n (%)**			
1 (Least deprived)	6079 (26.4)	5158 (26.9)	921 (24.0)
2	6266 (27.2)	5329 (27.7)	937 (24.4)
3	5836 (25.3)	4800 (25.0)	1036 (27.0)
4 (Most deprived)	4866 (21.1)	3917 (20.4)	949 (24.7)
Smoking, n (%)			
Current	2908 (12.6)	2263 (11.8)	645 (16.8)
Former	6259 (27.2)	5295 (27.6)	964 (25.1)
Never	7971 (34.6)	6704 (34.9)	1267 (33.0)
Body mass index (BMI), kg/m²,			
Mean (SD)	27.2 (5.3)	27.3 (5.3)	26.7 (5.4)
Family history of DM, n (%)	1360 (5.9)	1133 (5.9)	227 (5.9)
Non-oral GC use during follow-up*, n (%)	10259 (44.5)	8464 (44.1)	1795 (46.7)
Total oral GC cumulative dose over entire			
follow-up duration*, g,			
Median (IQR)	4.2 (2.0-8.0)	4.1 (2.0-7.6)	5.3 (2.0-10.4)
Ural GC cumulative dose within first two			
years of PMR/GCA diagnosis [*] , g,			
iviedian (IQR)	3.5 (1.8-5.3)	3.4 (1.8-5.1)	4.4 (1.8-7.0)

Table 6.1: Characteristics of newly diagnosed patients, stratified by disease status

Missing data (%): Ethnic group, 8.2%; BMI, 64.1%; Smoking, 25.6% *Follow up ceases on diagnosis of DM

**IMD categories were based on the overall distribution within the CPRD population

Table 6.1 describes the baseline characteristics of newly diagnosed patients, stratified by disease. PMR was the most common diagnosis (83%), followed by GCA (11%) and approximately 6% was diagnosed with a combination of both PMR and GCA. Patients with PMR and patients with GCA (with or without PMR) were similar in terms of age, sex, ethnicity and family history of DM.

Data on socio-economic status (see Table 6.1, index of multiple deprivation, IMD) suggest that PMR is under-represented in the most deprived quartile (as shown by the difference of 6.5% between the least deprived and most deprived quartiles), and similarly common in quartiles 1-3.

The percentage of current smokers was 5% higher in the GCA (with or without PMR) group as compared to the PMR group. The mean BMI at baseline in both groups were indicative of an overweight population, which was reflective of the general population in England. The 2017 Health Survey for England reported that the mean BMI among adults was 27.7kg/m² [296]. According to the World Health Organization (WHO) classification, overweight is defined as having a BMI between 25.0-29.9kg/m², and obesity is defined as \geq 30 kg/m². Of the 36% patients with a recorded BMI at baseline, approximately 63% had a BMI higher or equal to 25kg/m², suggesting that there is a possibility that overweight or obese patients were more likely to have a recorded BMI; while those with no recorded BMI may have had a normal BMI. Nearly half of the study population had some use of non-oral GC therapy during follow-up. The median total oral GC cumulative dose over the entire follow-up duration was approximately 4.2g (IQR 2.0-8.0g) in the overall cohort, and 3.5g (IQR 1.8-5.3g) over the initial two years since PMR/GCA diagnosis, suggesting that the extent of GC exposure is much higher in the first two years of follow-up. Total oral GC cumulative dose in patients with GCA (with or without PMR) was higher than in patients with PMR, as a higher dose is needed to treat symptoms and vascular complications of GCA.

6.4. Prescription Database

There were approximately 1 million prescriptions available between the start of study entry date and end of follow-up, of which 232,506 prescriptions (20.4%) were non-oral GC. These were not taken into account for the calculation of the exposure variable used in my analyses, as my primary objective was to study the effects of oral GC on the risk of DM. There were 68 different oral GC products in the prescription database, consisting eight types of GCs with different anti-inflammatory potency.

Types of Oral GC	Number of prescriptions, n (%)
Prednisolone*	648,481 (99.5)
Dexamethasone	1299 (0.2)
Hydrocortisone	1053 (0.2)
Budesonide	345 (0.1)
Betamethasone	291 (<0.1)
Prednisone*	265 (<0.1)
Deflazacort	150 (<0.1)
Methylprednisolone	7 (<0.1)
Total	651,891 (100.0)

Table 6.2: Types of oral GCs prescribed in primary care

*GCs generally recommended in treatment guidelines to treat PMR and/or GCA

Table 6.2 shows the eight types of GCs with different anti-inflammatory potency prescribed in primary care over the study duration. The two GCs generally recommended in treatment guidelines to treat PMR and GCA were prednisolone and prednisone, which accounted for approximately 99.5% of the entire prescription database used in my analyses.

6.4.2. Missing Data for Key Variables

There were four variables in the prescription database that contained information essential for data analyses – quantity of GC prescribed, name of product prescribed, prescription duration and "dosage instruction". These variables were listed separately as stand-alone variables in the database and extracted directly from primary care. The variable "dosage instruction" contained free text as documented by general practitioners (GPs) in primary care, of which most of the information relating to dose and frequency were recorded. In addition to these four variables, there were another two variables derived by CPRD, namely the "daily_frequency" (dosing frequency) and "daily_dose" (number of tablets per day). For this PhD, instead of using the two CPRD-derived variables listed above, I created another two variables (1) dosing frequency and (2) number of tablets per day from the information given in the "dosage instruction" variable. My extracted dosing frequency and number of tablets per day variables were then compared to the CPRD-derived frequency and number of tablets per day variables.

I first applied a rule-based approach for all common "dosage instruction" using the statistical software R. For example, the dosing frequency for all "OD" prescriptions was preset as "once-a-day". There were however, many typos and misspellings in the free text of the "dosage instruction" variable, of which manual screening was required for better accuracy. Of the initial 1 million prescriptions in the prescription database, there were 4182 unique "dosage instruction" that required manual screening. Information for daily dosing frequency and number of tablets per day was first extracted, and then rechecked on four separate occasions (January, March, June 2018 and August 2019) to ensure that the categories were properly coded. The extracted data was then contrasted and compared to the CPRD-derived data.

Key Variables	Extracted Data From Primary Care*	CPRD-derived**
	Missing da	ta, n (%)
Quantity prescribed	123 (0.0)	123 (0.0)
Product name	0 (0.0)	0 (0.0)
Duration	631,453 (96.9)	632,697 (97.1)
Dosing frequency	351,426 (53.9)	348,498 (53.5)
Tablets per day	376,782 (57.8)	358,246 (55.0)
Total	651,891	(100.0)

Table 6.3: Comparison of missing data for key variables in the prescription database

*Extracted data from primary care refers to the data I extracted from primary care data collected directly from registered primary care GPs

**CPRD derived data refers to the data that has been managed by the CPRD team

A summary of the distribution of missing data for the five important variables is shown in Table 6.3. The quantity of oral GC prescribed was a well-recorded variable, with only 0.02% missing data, and this was consistent with another two studies [297, 12] that reported a 99.3% and 99.9% of valid quantity value in drug exposure respectively. The product name (or GC name) was well documented, with no missing data as this variable was automatically extracted through scanning without the GP having to enter the information. This variable was crucial for determining the dosage strength and prednisolone-equivalent dose (PED) in the later stages of the analyses. The percentage of missing data in the "duration" variable was very high, as this was not a required field for GPs to fill out in the database. This finding was fairly consistent with above-mentioned study [297], which reported a missing value of 93%. I derived an additional 1244 duration values (0.2%) using the information recorded in the "dosage instruction" variable. In other words, if the duration was mentioned in the "dosage instruction" variable, but not given in the "duration" variable, then the information was extracted for use in subsequent analyses. Information on "daily dosing frequency" and "number of tablets per day" appeared to be more complete in the CPRD-derived dataset, because in the CPRD-derived dataset, when no information was provided on the number of tablets per day, one tablet was assumed, but only in some cases. There was however, no consistency and no obvious pattern on how this was determined in the CPRD-derived dataset. As above, a "once a day" dosing frequency is assumed for the CPRD-derived dataset if no information was available, though again, there was no consistency and no obvious pattern on how this was determined. For my extracted data, the number of tablets per day was assumed to be one if no information was available. Similarly, the daily dosing frequency was also assumed to be one if no other information was available.

6.4.3. Dosing Frequencies

Dosing Frequencies	Extracted Data From Primary Care*	CPRD-derived**	
	Number of Prescriptions, n (%)		
Once a day	283,231 (43.4)	297,366 (45.6)	
1.5 times per day	-	62 (0.0)	
1.67 times per day	-	2 (0.0)	
Twice a day	3,240 (0.5)	3,187 (0.5)	
2.5 times per day	-	1 (0.0)	
Three times a day	1,594 (0.2)	2,267 (0.3)	
3.5 times per day	-	2 (0.0)	
Four times a day	487 (0.1)	492 (0.1)	
Five times a day	5 (0.0)	8 (0.0)	
Six times a day	6 (0.0)	6 (0.0)	
Every other day	3,645 (0.6)	-	
Once weekly	109 (0.0)	-	
Twice weekly	60 (0.0)	-	
Three times a week	9 (0.0)	-	
Tapering	4,588 (0.7)	-	
When needed	3,483 (0.5)	-	
Stat/Immediately	8 (0.0)	-	
As directed	161,399 (24.8)	-	
Unknown	190,027 (29.2)	348,498 (53.5)	
Total	651,891 (100.0)		

Table 6.4: Distribution of dosing frequencies in the prescription database

*Extracted data from primary care refers to the data I extracted from primary care data collected directly from registered primary care GPs

**CPRD derived data refers to the data that has been managed by the CPRD team

Table 6.4 shows the distribution of dosing frequencies recorded in the prescription database. The three most common dosing frequencies were "once a day", "as directed" and "unknown", which accounted for 97.4% of the prescription database. About one fourth of the prescriptions were "as directed" prescriptions, which meant that the dose and/or dosing frequency were not specified, but patients were instructed to "take as directed" verbally. In some cases, the dose was specified, but not the dosing frequency, and vice versa. All prescriptions were categorized as "unknown" if they did

not contain any information on the specific dosing frequency, and no instruction on "take as directed" was indicated. The total number of prescriptions categorized as "once a day" was about 2.2% higher in the CPRD-derived dataset, as compared to the variables I created using the information collected from primary care data. This was partly because some prescriptions involving "every other day", "weekly" or "multiple times weekly" dosing were categorized as "once a day" in the CPRD-derived dataset. Another reason was due to the fact that if the dosing frequency were unknown, it would be categorized as "once a day", but only in some cases.

As mentioned in Section 6.4.2, I also applied certain assumptions that correlated with clinical practice in my subsequent analyses, which included the assumption that the dosing frequency would be once a day if any of the following directions were recorded in the primary care data: "tapering regimens", "when needed", "stat/immediately", "as directed", or "unknown". The application of this assumption resulted in 98.6% of the prescriptions having a "once a day" dosing frequency.

6.4.4. Duration of GC Prescriptions

As discussed in Section 6.4.2, the "duration" variable was not well recorded in primary care, with only approximately 3% of the prescription database having a recorded duration value. Despite the large number of missing data in the duration variable, there were other ways of deriving information on duration by using a combination of other variables in the database that was explained in detail in the previous chapter. After applying all the various approaches detailed in Chapter 5 (Section 5.5.4.1), approximately 1.7% of prescriptions still had a missing duration due to insufficient information. The median duration for prescriptions with a quantifiable duration for PMR, as well as GCA (with or without PMR) was 28 days. Therefore, for the 1.7% of prescriptions that had a missing duration, the duration was replaced with the median of 28 days. Approximately 4.9% of the prescriptions had a duration of more than 90 days, which is unlikely as most prescriptions in primary care were usually prescribed for a period between 28 to 90 days as GPs are strongly recommended by the National Health Service (NHS) to not prescribe medications for more than three months. To address this issue, all prescriptions that had a duration of more than 90 days were capped at the 90 days limit.

The median duration of treatment was approximately 1.7 years (IQR: 0.7-3.5 years) in the PMR group and 1.5 years (IQR: 0.4-3.5 years) for the GCA (with or without PMR) group.

6.4.5. Number of Tablets Per Day and Daily Dosing

Approximately 2.5% of the prescription data had a daily dose of less than 1mg, of which 0.8% had a daily dose of less than 0.5mg. On the other hand, approximately 0.6% had a dose of more than 120mg/day. One plausible explanation for these very high dose prescriptions was pulse therapy for GCA patients with threatened or established visual loss at diagnosis. This is possible but unlikely, as these were all oral prescriptions; therefore the maximum daily dose for these prescriptions was capped at 120mg.

The median average daily dose of GC prescribed for the entire follow-up duration in the practice was 6.6mg/day (IQR: 5.0-9.6mg/day) and 9.8mg/day (IQR: 6.5-17.3mg/day) for the PMR group and GCA (with or without PMR) group, respectively.

6.5. GC Prescribing Patterns

The results for this section present the oral GC prescribing patterns for two groups of patients – those with PMR, and those with GCA (with or without PMR) over the first two years since the diagnosis of PMR/GCA. The results are summarized in the forms of total monthly oral GC dose, high dose (75th percentile) versus low dose regimen (25th percentile) and average monthly oral GC dose. The prescribing pattern in primary care was also compared against EULAR guidelines for the different diseases respectively.

6.5.1. Total Monthly Oral GC Dose

Figure 6.2 presents the box plots of total monthly oral GC dose prescribed to PMR patients over 24 months. As expected, there was a decreasing trend in the total monthly oral GC dose prescribed over time. The initial total dose in the first month was 714mg in the upper quartile, compared to 263mg in the lower quartile. The range of total monthly oral GC doses was more varied in the first twelve months, as shown by the monthly box plots. It is also worth noting that after 18 months since diagnosis, half of the patients were no longer on GC treatment (as shown by the total GC dose of zero). The long upper whisker across the months indicates that the doses were more varied in the upper quartile as compared to the lower quartile. Outliers were present, of which there were some total monthly doses exceeding 3g in the initial stages of follow-up.



Figure 6.2: Total monthly oral GC in PMR patients over 24 months

Figure 6.3 presents the box plots of total monthly oral GC doses in patients with GCA (with or without PMR) over 24 months. There was a decrease in the total monthly oral GC doses prescribed over time. The initial total dose in the first month was higher compared to the PMR group, starting at approximately 1050mg in the upper quartile and 200mg in the lower quartile. The range of total monthly oral GC doses was more varied in the first 12 months, as suggested by the comparatively taller box in the boxplot. As with the PMR group, half of the patients were no longer on GC treatment after 18 months. The long upper whisker across the months indicates that the doses were more varied in the upper quartile as compared to the lower quartile. Outliers were also present in this patient population, of which there were some total monthly doses exceeding 3g in the initial stages of follow-up.



Figure 6.3: Total monthly oral GC in GCA (with or without PMR) patients over 24 months

6.5.2. High Dose (75th Percentile) Versus Low Dose (25th Percentile) Regimens

Figure 6.4 presents the prescribing trend of patients on high GC dose versus those on low GC dose, defined by the 75th percentile and 25th percentile respectively. Approximately 75% of patients with PMR were prescribed with less than 714mg/month (23.8mg/day) in the initial month of treatment, and one fourth of PMR patients stopped GC treatment by the sixth month. For patients with GCA (with or without PMR), 75% of them were prescribed with less than 1050mg/month (35mg/day) in the initial month of patients no longer on GC treatment after the third month. The purpose of Figure 6.4 is to describe the dose regimen in primary care practice, specifically looking at the range of patients who were on high dose and low dose GC. It is likely that patients on high dose Versus low dose GC users is important for risk assessment between the two groups.



Figure 6.4: Comparison of high GC dose (75th percentile) versus low GC dose (25th percentile) among PMR and/or GCA patients within the first two years of diagnosis

6.5.3. Average Total Monthly Oral GC Dose

Figure 6.5 presents the average total monthly oral GC dose over 24 months for PMR patients in primary care, stratified by DM status. Curves were fitted using the added smoothed conditional regression line. Patients who developed DM generally had higher GC doses throughout the course of treatment, starting with high doses, which were tapered over time. The difference in GC dose between patients who developed DM and those that did not develop DM was higher after six months of diagnosis. The highest average total monthly oral GC dose for DM patients was 520mg and 517mg in the non-DM group in the first month after diagnosis.



Figure 6.5: Average total monthly oral GC prescribing pattern over 24 months for PMR patients, stratified by DM status

Figure 6.6 presents the average total monthly oral GC dose over 24 months for patients with GCA (with or without PMR) in primary care, stratified by DM status. DM patients had a higher GC starting dose and showed a slower tapering trend after month nine, compared to the non-DM group. The highest average total monthly oral GC dose for DM patients was 730mg and 696mg in the non-DM group in the first month after diagnosis.



Figure 6.6: Average total monthly oral GC prescribing pattern over 24 months for patients with GCA (with or without PMR), stratified by DM status

Figures 6.5 and 6.6 are a depiction of the distribution of dosing for those with and without DM. These plots were however not used for interpretation, as there were some issues with it. For example, even though DM patients seemed to have a much higher dose in month 18 compared to non-DM patients, it does not really depict the actual dosing of all DM patients as only those who were still in the cohort were taken into consideration.

Figure 6.7 presents a comparison between real-world GC prescribing pattern among PMR patients in primary care and the EULAR guideline recommendations over the initial 12 months of treatment. The duration of 12 months was chosen, as specific guideline recommendations were only available up to 12 months. The median monthly starting dose for the entire PMR cohort was 500mg/month (or 16.7mg/day), which corresponds well to the recommended starting dose of 375-750mg/month (or 12.5-25mg/day by the 2015 EULAR guidelines [21]. The median first GC dose was 15.0mg (IQR: 10.0-20.0mg), while the cumulative dose over two years was 3.6g (IQR: 1.8-5.1g). Patients in primary care appeared to have a much steeper tapering regimen as compared to guidelines. For example, the median monthly dose was reduced to approximately 300mg/month (or 10mg/day) in the second month of treatment, suggesting almost a 7mg taper in GC dose within 4 weeks as compared to the recommended tapering of 2.5mg/day after 4 weeks of treatment. However, the prescribed median monthly dose in month 12 appeared to be approximately 300mg/month (or 1mg/day) higher than the recommended guideline dose of 1mg/day.



Figure 6.7: Comparison of monthly oral GC for PMR patients in primary care with EULAR guideline within the first year of diagnosis

Figure 6.8 presents a comparison real-world GC prescribing pattern among GCA (with or without PMR) patients in primary care and the EULAR guideline recommendations over the initial 12 months of treatment. Of the 3844 patients in this cohort, approximately 35% had both PMR and GCA. Of this 35% who had both diseases, 47% were diagnosed with PMR first, while another 53% had a diagnosis of GCA first. The median monthly starting dose for this cohort was 560mg (or 18.7mg/day), which was much lower than the recommended 1500mg (or 50mg/day) by the EULAR guideline [22]. The median first GC dose was 20.0mg (IQR: 15.0-35.7mg), while the cumulative dose over two years was 4.2g (IQR: 1.7-6.9g). I did some exploratory analysis by just looking at patients who had GCA as their first diagnosis. The prescribing pattern for this subset of patients did not differ from the cohort who had both GCA and a combination of both PMR and GCA, as shown in Figure 6.8. It is worth noting though, that approximately half of the initial prescriptions did not have a quantifiable dose (as shown in Figure 6.9), in which case the assumption of "one tablet per day" was applied to if no other information was available. When compared to the recommended tapering regimen by the EULAR guidelines, it appeared that GC dosing initiation in primary care for this patient population was much lower than the recommendations.



Figure 6.8: Comparison of monthly oral GC for GCA (with or without PMR) patients in primary care with EULAR guideline within the first year of diagnosis

Figure 6.9 presents the percentage of the initial oral GC prescription without a quantifiable dosing instruction among patients with GCA (with or without PMR) in primary care. As mentioned above, these were the prescriptions of which the dose and frequency were not specified; therefore the assumption of "one tablet per day" was applied if no other information was available.



Figure 6.9: Percentage of first oral GC prescription without a quantifiable dosing instruction among patients with GCA (with or without PMR) in primary care

6.6. Pattern of Distribution for DM Over Time

This section outlines two simple descriptive frequency plots (presented as Kaplan-Meier derivatives) on the incident rate of DM over 24 months. The purpose of these plots is to have better insight on the pattern of distribution of DM over time.

Figure 6.10 presents the percentage of incident DM in PMR patients over 24 months. The results suggest that incidence of DM is highest within the first month then decreases to an approximately steady rate. Table 6.5 shows the frequency and percentage of incident DM diagnosis by month since the start of disease, with a total of 832 (4.4%) patients developing DM within the first two years of follow-up.



Percentage of incident DM in 19204 PMR patients over 24 months

Figure 6.10: Percentage of incident DM in PMR patients over 24 months

Months	Incident DM (n)	Number at risk (n)	Percent (%)
1	75	19204	0.39
2	59	19129	0.31
3	59	19070	0.31
4	57	19011	0.30
5	31	18954	0.16
6	44	18923	0.23
7	40	18879	0.21
8	42	18839	0.22
9	42	18797	0.22
10	24	18755	0.13
11	33	18731	0.18
12	29	18698	0.16
13	26	18669	0.14
14	27	18643	0.15
15	28	18616	0.15
16	29	18588	0.16
17	30	18559	0.16
18	33	18529	0.18
19	25	18496	0.14
20	18	18471	0.10
21	20	18453	0.11
22	23	18433	0.12
23	19	18410	0.10
24	19	18391	0.10
Total	832		4.43

Table 6.5: Frequency and proportion of incident DM in PMR patients over 24 months

Figure 6.11 presents the percentage of DM in patients with GCA (with or without PMR) over 24 months. The results suggest that incidence of DM is highest within the first month, and then declines over time. Table 6.6 shows the frequency and percentage of incident DM diagnosis by month since the start of disease, with a total of 212 (5.7%) patients developing DM within the first two years of follow-up.



Figure 6.11: Percentage of incident DM in GCA (with or without PMR) patients over 24 months

Table 6.6: Frequency and proportion	of incident DM in	patients with	GCA	(with or
without PMR) over 24 months				

Months	Incident DM (n)	Number at risk (n)	Percent (%)
1	19	3844	0.50
2	25	3825	0.66
3	17	3800	0.45
4	21	3783	0.56
5	11	3762	0.29
6	10	3751	0.27
7	13	3741	0.35
8	4	3728	0.11
9	10	3724	0.27
10	9	3714	0.24
11	4	3705	0.11
12	9	3701	0.24
13	6	3692	0.16
14	2	3686	0.05
15	5	3684	0.14
16	3	3679	0.08
17	4	3676	0.11
18	3	3672	0.08
19	7	3669	0.19
20	8	3662	0.22
21	4	3654	0.11
22	6	3650	0.16
23	6	3644	0.16
24	6	3638	0.17
Total	212		5.68

6.7. Survival Analysis

6.7.1. Kaplan Meier

The results for this section are presented as Kaplan-Meier survival curves. Kaplan-Meier survival curves were used to assess time to DM diagnosis, with "event" being newly diagnosed DM. The probability of developing DM between groups was compared using the log-rank test.

6.7.1.1. DM-Free Time by Disease

Figure 6.12 presents the survival curves for time to DM diagnosis, stratified by disease group. The p-value from the log-rank test indicates that there was a significant difference in the probability of developing DM between disease groups.



Figure 6.12: Kaplan-Meier survival curves for time to DM diagnosis, stratified by disease groups
6.7.1.2. DM-Free Time by Gender

Figure 6.13 presents the survival curves for time to DM diagnosis for males and females. The p-value from the log-rank test indicates that there was no significant difference in the probability of developing DM between genders.



Figure 6.13: Kaplan-Meier survival curves for time to DM diagnosis, stratified by gender

Figure 6.14 presents the survival curves for time to DM diagnosis by age in quartiles. The p-value from the log-rank test indicates that there is no significant difference in the risk of developing DM between age groups. The entire cohort was also divided by tertiles and quintiles for age, and Kaplan-Meier survival curves were plotted, of which there was no significant statistical difference, with p-values at 0.99 and 0.29 respectively.



Figure 6.14: Kaplan-Meier survival curves for time to DM diagnosis, stratified by age group in quartiles

6.7.2. Extended Time-varying Cox Model

As detailed in Chapter 5, the use of the extended time-varying Cox model was a more appropriate statistical method to examine the association between GC cumulative dose and DM as it allows the incorporation of covariates that change over time into the model. For this section, I used two different approaches - firstly, the risk of DM was quantified based on each gram increase in cumulative GC dose within the initial two years of PMR/GCA diagnosis. Secondly, the risk of DM was compared between patients who were on high dose (75th percentile) versus those who were on low dose (25th percentile) within the initial two years of PMR/GCA diagnosis. The use of the "high dose versus low dose" comparison is a useful way to describe dose regimens and provides a more intuitive interpretation for clinical practice.

6.7.2.1. Extended Time-varying Cox Model for Cumulative Dose (per gram increase in risk)

Variables	Hazard Ratio*	95% Confidence Interval	p-value
Cumulative dose (per 10 g)	1.39	1.22, 1.60	<0.001
Cumulative dose squared (per 10 g)	1.20	1.03, 1.39	0.0184
Family history of DM	1.68	1.63, 1.74	<0.001
Non-oral GC use	1.07	1.05, 1.10	< 0.001

Table 6.7: Extended time-varying Cox regression model for DM in PMR patients

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

Results in Table 6.7 indicate that for every 10g increase in cumulative GC dose, there was a 39% increase in DM risk while adjusting for family history of DM and non-oral GC use during study follow-up among PMR patients. The results also suggest that there was a 68% increased risk of DM in patients with a family history of DM as compared with those without; while non-oral GC use during study follow-up was associated with a 7% increase in DM risk compared to those who were not prescribed any non-oral GC.

Variables	Hazard Ratio*	95% Confidence Interval	p-value
Cumulative dose (per gram)	1.02	1.02, 1.03	<0.001
Family history of DM	1.84	1.71, 1.98	<0.001
Non-oral GC use	1.30	1.24. 1.35	< 0.001

Table 6.8: Extended time-varying Cox regression model for DM in patients with GCA (with or without PMR)

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

The model in Table 6.8 indicates a 2% increase in DM risk with every gram increase in cumulative GC dose while adjusting for family history of DM and non-oral GC use during study follow-up among patients with GCA (with or without PMR). There was almost a 2-fold increased risk of DM in patients with a family history of DM compared with those without, while non-oral GC use during study follow-up was associated with a 30% increase in DM risk compared to those not prescribed non-oral GC.

6.7.2.2. Extended Time-varying Cox Model for Cumulative Dose (High dose versus low dose)

For all subsequent analyses, I will be comparing the hazard ratios between those on high dose versus those on low dose GC therapy. Patients on high GC dose were those in the upper quartile (i.e. 75th percentile) and those on low GC dose were in the lower quartile (i.e. 25th percentile). The difference in aggregated dose between the lower quartile and upper quartile for each month since PMR and/or GCA diagnosis was calculated, multiplied with the coefficient calculated from the extended time-varying Cox model, and then exponentiated to obtain the hazard ratio. Hazard ratios were calculated for the first 24 months since the diagnosis of PMR and/or GCA.

Table 6.9: Hazard ratios (HRs) for DM in patients with high GC dose (75th percentile) versus low GC dose (25th percentile) over 24 months using the extended time-varying Cox analysis

Monthe since DMD	Hazard Ratios (HRs)		
and/or GCA Diagnosis	PMR	GCA	
1	1.02		
2	1.02	1.02	
2	1.05	1.04	
3	1.05	1.05	
	1.00	1.00	
5	1.07	1.07	
6	1.09	1.08	
1	1.10	1.09	
8	1.11	1.10	
9	1.12	1.11	
10	1.13	1.11	
11	1.14	1.12	
12	1.15	1.12	
13	1.16	1.13	
14	1.17	1.13	
15	1.18	1.14	
16	1.19	1.14	
17	1.20	1.15	
18	1.20	1.15	
19	1.21	1.15	
20	1.22	1.16	
21	1.22	1.16	
22	1.23	1.16	
23	1.24	1.17	
24	1.24	1.17	

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

Results in Table 6.9 indicate that as GC cumulative dose increases, the risk of developing DM also increases over cumulative months following the diagnosis of PMR and/or GCA. The results suggest that for PMR patients, those who were on high doses had a 2-24% increased risk of developing DM over 24 months as compared to those on low dose. The magnitude of risk was narrower in the GCA (with or without PMR) group, of which the risk ranged from 2-17% over 24 months since the start of diagnosis.

6.7.3. Rolling Cumulative Dose Model

The risk of DM from the extended time-varying Cox model is generally proportionate to the amount of cumulative GC doses over time, but as shown in Figures 6.7 and 6.8, GC doses are usually tapered over time, therefore is very likely that the risk of DM will peak at a certain time point during treatment, and then decreases. This is where determination of the timing of GC exposure is crucial. One of the methods that may be used to address this issue is the use of the rolling cumulative dose model.

Table 6.10: Concordance index of rolling cumulative dose based on months following the diagnosis of PMR and/or GCA

Rolling Months Since	Concordance Index (Standard Error)	
Diagnosis of PMR	PMR	GCA
and/or GCA		(with or without PMR)
1	0.555 (0.002)	0.574 (0.004)
2	0.556 (0.002)*	0.576 (0.004)*
3	0.555 (0.002)	0.575 (0.004)
4	0.554 (0.002)	0.574 (0.004)
5	0.553 (0.002)	0.572 (0.004)
6	0.551 (0.002)	0.570 (0.004)
7	0.550 (0.002)	0.569 (0.004)
8	0.548 (0.002)	0.568 (0.004)
9	0.547 (0.002)	0.566 (0.004)
10	0.545 (0.002)	0.565 (0.004)
11	0.544 (0.002)	0.564 (0.004)
12	0.542 (0.002)	0.563 (0.004)
13	0.541 (0.002)	0.562 (0.004)
14	0.540 (0.002)	0.561 (0.004)
15	0.539 (0.002)	0.560 (0.004)
16	0.539 (0.002)	0.560 (0.004)
17	0.538 (0.002)	0.557 (0.004)
18	0.537 (0.002)	0.555 (0.004)
19	0.537 (0.002)	0.553 (0.004)
20	0.537 (0.002)	0.551 (0.004)
21	0.536 (0.002)	0.551 (0.004)
22	0.536 (0.002)	0.551 (0.004)
23	0.536 (0.002)	0.551 (0.004)
24	0.536 (0.002)	0.551 (0.004)

* Best fitting model

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

Table 6.10 shows a comparison of the concordance index for all the models, of which the highest index was considered as the best fitting model. The results indicate that the most recent two months of GC exposure had the highest impact in both groups. The coefficients of the models in the second month for both groups were determined and applied to calculate the hazard ratios in patients with high GC dose versus low GC dose over the first two years since diagnosis, as shown in Table 6.11 below.

Monthe since DMD	Hazard Ratios (HRs)		
and/or GCA Diagnosis	PMR	GCA (with or without PMR)	
1	1.22	1.34	
2	1.26	1.52	
3	1.37	1.55*	
4	1.38*	1.48	
5	1.38*	1.43	
6	1.38*	1.38	
7	1.37	1.34	
8	1.34	1.30	
9	1.32	1.27	
10	1.29	1.25	
11	1.27	1.23	
12	1.25	1.21	
13	1.24	1.20	
14	1.22	1.19	
15	1.21	1.17	
16	1.20	1.15	
17	1.19	1.14	
18	1.18	1.13	
19	1.17	1.12	
20	1.16	1.12	
21	1.15	1.11	
22	1.14	1.11	
23	1.13	1.10	
24	1.12	1.09	

Table 6.11: Hazard ratios (HRs) for DM in patients with high GC dose versus low GC dose over 24 months using the rolling cumulative dose analysis

*Highest DM risk

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

The hazard ratios for rolling cumulative doses over two most recent consecutive months are listed in Table 6.11. For the PMR group, DM risk was highest (38% increased risk) between four to six months after the initiation of GC therapy in patients on high GC dose as compared to those on low GC dose. The risk subsequently decreased over time. For the GCA (with or without PMR) group, the risk was highest three months after the initiation of GC therapy, as indicated by a 55% increased risk in patients on high GC dose as compared to those on low GC dose. As with the PMR group, the risk of developing DM in this group also decreased over time.

In terms of confounders, PMR patients with a family history of DM were found to have a 68% (HR: 1.68, 95%CI: 1.63-1.74) increased risk in developing DM during the most recent two months of GC exposure compared to those without a positive family history. Use of non-oral GC use during study follow-up was also associated with a 7% increased risk (HR: 1.07; 95% CI: 1.05-1.09) compared to non-users. Similarly for the GCA (with or without PMR) group, a positive family history of DM and use of non-oral GC during study follow-up were associated with an increased risk of 84% (HR: 1.84; 95% CI: 1.71-1.98) and 29% (HR: 1.29; 95% CI: 1.24-1.35) respectively.

6.7.4. Weighted Cumulative Exposure (WCE) model

While the rolling cumulative dose analysis is a useful method to determine the risk based on recent cumulative doses, it has its limitation as the time-dependent hazard is fixed for each time window. To address this limitation, I used another statistical method, namely the WCE analysis that allows cumulative dose to be calculated as a weighted mean of past doses, with higher weights assigned to more recent doses.

Most Recent Months	Bayesian Information Criterion (BIC)	
To DM Diagnosis	PMR	GCA (with or without PMR)
1	16211.05	3455.236
2	16183.40	3449.437
3	16120.64	3403.200
4	16105.22	3395.599
5	16097.93	3393.847*
6	16093.24	3395.278
7	16089.28	3397.116
8	16086.20	3398.771
9	16084.40	3399.954
10	16083.96*	3400.760
11	16084.59	3401.487
12	16085.90	3402.204
13	16087.46	3402.790
14	16088.93	3403.233
15	16090.10	3403.555
16	16090.89	3403.781
17	16091.37	3403.949
18	16091.64	3404.077
19	16091.77	3404.170
20	16091.82	3404.265
21	16091.81	3404.265
22	16091.77	3404.273
23	16091.72	3404.262
24	16091.67	3404.234

Table 6.12: Comparison of goodness of fit test across WCE models over 24 months

* Best fitting model

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

Table 6.12 shows the BIC results for various WCE models for the most recent months to the diagnosis of DM. For the PMR group, doses taken in the most recent 10 months had the highest impact on the risk of developing for DM; while for those with GCA (with or without PMR), the WCE weight function indicates that GC doses in the most recent 5 months to DM diagnosis had the highest impact.

Months since PMP	Hazard Ratios (HRs)		
and/or GCA Diagnosis	PMR	GCA (with or without PMR)	
1	1.15	0.91	
2	1.39	1.67	
3	1.69	2.73	
4	2.00	2.84*	
5	2.25	2.38	
6	2.43	2.10	
7	2.48*	1.91	
8	2.43	1.77	
9	2.30	1.66	
10	2.16	1.58	
11	2.03	1.52	
12	1.92	1.46	
13	1.83	1.41	
14	1.75	1.38	
15	1.68	1.36	
16	1.62	1.32	
17	1.57	1.29	
18	1.53	1.26	
19	1.49	1.25	
20	1.46	1.22	
21	1.43	1.21	
22	1.40	1.20	
23	1.37	1.19	
24	1.34	1.17	

Table 6.13: Hazard ratios (HRs) for DM in patients with high GC dose versus low GC dose over 24 months using the weighted cumulative dose analysis

* Best fitting model

Note: Adjusted for cumulative dose, family history of DM and non-oral GC use

To explore how the risk changes each month since the diagnosis of PMR and/or GCA, the hazard ratios were calculated and compared in patients on high GC dose versus those in the low dose group, as shown in Table 6.13. For the PMR group, the risk was highest (2.5-fold) 7 months after the diagnosis of PMR among those on a high GC dose as compared to those on a low GC dose. For the GCA (with or without PMR) group, the risk was highest (2.8-fold) 4 months after the diagnosis of GCA and/or PMR.

Figure 6.15 displays weights estimated by the best-fitting WCE models against time elapsed since GC dosing. The x-axis represents the time of which GC dose was taken, starting at month 0 until month 10 for PMR, and month 5 for those with GCA (with or without PMR). Plots indicate how the impact of past GC exposure on the development of DM decreased with increasing time elapsed since the start of follow-up. The higher weight function in the most recent time period suggests that more recent treatment has a greater impact on the risk of DM. The estimated weights across the disease groups suggested that past exposure becomes irrelevant after 10 months of GC initiation for PMR patients and 5 months for patients with GCA (with or without PMR). Interestingly, there seemed to be a delay in the onset of DM diagnosis in relation to current dose, with the highest impact only occurring at month 3 (for the PMR group) and month 2 (for the GCA (with or without PMR) group). One of the plausible reasons is that there was a delay in the recording of DM diagnosis at primary care since it would take at least two measurements of glucose test, assessment of symptoms and HbA1c testing to confirm the diagnosis of DM, for which time is needed.









In terms of confounders, PMR patients with a family history of DM were found to have a 2.3-fold increased risk of developing DM during the most recent ten months of GC exposure as compared to those without a positive family history. Use of non-oral GC was negatively associated with the risk of DM (adjusted HR=0.41). The results for the GCA (with or without PMR) group were similar. Family history was associated with a 2fold increased risk of developing DM during the most recent five months of GC exposure, while use of non-oral GC during study follow-up was negatively associated with the risk of DM (adjusted HR=0.37).

6.7.5. Comparison of Hazard Ratios Across the Extended Time-Varying Cox, Rolling Cumulative Dose and Weighted Cumulative Dose Analyses

Figures 6.16 and 6.17 summarize the risk of developing DM between patients on high dose versus low dose over 24 months using three different statistical methods. All three methods reflect hazard ratios that indicate an increase in risk of developing DM over time, especially within the first year of follow-up. Of these three methods, results from the WCE are the most reliable, taking into account the dose, duration and recency of treatment. The result suggest that oral GC dose taken up to 10 months ago may result up to a 2.5-fold increased risk in PMR patients, while for those with GCA (with or without PMR), use of oral GC taken up to 5 months ago may increase the risk of developing DM up to 2.8-fold.



Figure 6.16: Comparison of hazard ratios across the three different statistical methods in PMR patients on high GC dose versus low GC dose





The results for patients with PMR were more reliable as they were more likely to be treated in primary care, even in the initial stages of their disease. Patients with GCA (with or without PMR), on the other hand, were more likely to be managed by a consultant in secondary care for initial treatment before referring back to primary care for further management, therefore there is a strong possibility that GC doses captured in primary care during the initial months of treatment were largely underestimated.

In the next chapter, I will discuss the results of my findings and its impact in clinical practice.

CHAPTER 7: DISCUSSION AND CONCLUSION

7.1. Overview

This study has primarily examined the risk of diabetes mellitus (DM) associated with oral glucocorticoid (GC) use in patients with polymyalgia rheumatica (PMR) and/or giant cell arteritis (GCA) in a primary care setting. One of the issues was that all patients were exposed to GCs, therefore the risk of DM was compared between those on a high dose regimen with those on a low dose regimen. This is the first study to explore the impact of dosage, duration and timing of oral GC use on the risk of DM in PMR and GCA patients through the use of routinely collected primary care data with a large sample size using the rolling cumulative dose and weighted cumulative exposure models. All models indicated that there was a higher risk of developing DM for the higher dose regimen compared to the lower dose regimen. This is also the first study to compare trends of oral GC prescribing patterns in primary care with current clinical practice guidelines. While there have been published literatures on starting doses, average daily doses and cumulative doses, there are no published literature on the tapering regimen of oral GC doses in real-world practice as compared to guidelines. In addition, this is also the first study to compare Clinical Practice Research Datalink (CPRD) derived data (i.e. data managed by the CPRD team) with primary care data collected directly from registered primary care general practitioners to assess the completeness of CPRD derived data, which is often used by researchers in the analyses performed in many publications.

This chapter will cover the following areas: firstly, an evaluation of the results presented in Chapter 6, including a discussion on the risk of DM associated with GC use and how the study findings relate to other published literature. This sub-section will also include a discussion on GC prescribing pattern in primary care, followed by some of the challenges associated with analyzing primary care prescription databases, as well as discrepancies identified between the CPRD-derived GC data versus the original primary care data. Secondly, I will be discussing the strengths and limitations of this study, along with the appropriate steps taken to minimize the various biases. As the use of GCs in the treatment of PMR and GCA is inevitable, the third sub-section sets out considered recommendations for monitoring and treating those who actually develop DM after GC treatment. This chapter will end with an outline of future work and a summary of the thesis.

7.2.1. Risk of DM associated with Oral GC Cumulative Dose

There have been published literatures exploring the association between oral cumulative GC therapy and DM risk in patients with PMR and/or GCA patients, and while these studies explored the impact of dosage and duration of GC use on risk of DM, none of them considered the recency of GC exposure. It is insufficient to only consider the time-dependent aspect of GC exposure, as recency of exposure might be just as crucial. For my PhD, I used three different statistical methods, including the extended time-varying Cox model, the rolling cumulative dose model and weighted cumulative exposure (WCE) model, to quantify the risk of DM associated with oral cumulative GC use. Of the three methods used, the novel rolling cumulative dose and WCE models have added insights into the dose and temporal relationship between oral cumulative GC use in primary care and risk of developing DM in patients with PMR and/or GCA. The WCE model depicts the use of GCs in clinical practice best as it allows GC exposure to be represented as the weighted sum of past doses, allowing the most recent doses to be given the most weight using cubic splines.

7.2.1.1. Risk of DM associated with Oral GC Cumulative Dose in PMR Patients

Results from the rolling cumulative dose and WCE analysis suggest that PMR patients on high dose GC therapy have up to 1.4-2.5-fold increase in DM risk compared to those who were on low dose therapy. A typical high GC dose (75th percentile) in the initial month of treatment was 24mg/day and a typical low dose (25th percentile) was 9mg/day in my dataset of PMR patients, following which doses were tapered down over time. To date, there are relatively very few publications on the incidence of DM in patients with PMR compared to patients with GCA. Results from my meta-analysis in Chapter 4 showed that the incidence of DM associated with GC use was only reported by 9 studies up till 2017, all of which had small sample sizes of less than 250 patients. The incidence proportion of GC-induced DM based on these studies was 6%, which was 1.2% higher compared to the UK general population of similar gender and age group [11]. A more recent study from an academic medical centre in North America mainly consisting of northern European ancestry PMR patients reported that there was no difference in the rate of GC-induced DM as compared to their age and sex-matched comparators without PMR [298]. The sample size for that study was however, relatively small (359 patients) as compared to my current study (19,204 patients).

Results from the WCE analysis also indicated that the risk of developing DM was only significant within the most recent ten months of treatment, with the highest risk being at month seven. Results from the rolling cumulative dose support this finding, with the highest risk occurring between four to six months after GC initiation. One of the unique findings of this study is that the GC effect on DM lasts for several months but becomes negligible (diminishes to almost zero) after approximately one year.

7.2.1.2. Risk of DM associated with Oral GC Cumulative Dose in Patients with GCA (with or without PMR)

Results pertaining to the risk of DM associated with oral GC cumulative dose in patients with GCA (with or without PMR) ought to be interpreted with caution, as it is very likely that the initial doses of GC therapy in this study are an underestimate of what is actually being prescribed in a real-world setting. This is because GCA patients are often initially managed in secondary care as the disease itself is considered to be a medical emergency [299, 31], of which untreated cases may lead to permanent blindness; thus early and rapid treatment with high GC doses is believed to be crucial to control inflammatory symptoms and prevent ischaemic manifestations. Since CPRD is a primary care database, any initial (likely higher) doses of GCs prescribed in secondary care were not available. As a result, it is likely that the risk of DM is underestimated. There is however, a small possibility that the risk might be overestimated because patients may have had a relatively low oral GC dose but a large non-oral GC dose. Since "non-oral GCs" was adjusted as a binary variable, the actual magnitude of DM risk may have been disrupted. This is however, unlikely, as most non-oral GCs are unlikely to be given long term (i.e. intravenous GCs are usually only given in the initial stages of treatment of GCA for less than a week). Another factor to consider is the bioavailability of the drug, which is often dependent on the mode of administration. Systemic GCs given intravenously will have a 100% bioavailability, but as mentioned, they are usually only given for a short duration, therefore unlikely to have a large impact on the estimation of DM risk. Another type of non-oral GC that may have been commonly used in this group of elderly patients is inhaled GCs for the treatment of respiratory diseases such as asthma. Inhaled GCs have a very low bioavailability (16-28%) [300]; therefore it is unlikely that the use of inhaled GCs would result in a higher DM risk even if given for a long duration. This hypothesis is supported by one large nested case-control population study in the UK, of which the authors reported that non-oral GCs (including inhalers, topical preparations, eye drops and injections), had no association with the development of incident DM [181]. An issue of interest is whether the addition of these non-oral GCs (especially the high dose systemic GCs with excellent bioavailability) actually alters the magnitude of DM risk when given concurrently with oral GCs. This is however, beyond the scope of this PhD, as prescribed medications in secondary care were not linked to CPRD. Despite this limitation, results from my analyses still showed an increased risk of DM by up to 1.6-2.8-fold in patients on high GC dose versus low GC dose. A typical high dose (75th percentile) and low dose (25th percentile) in the initial month of treatment for this patient population was 35mg/day and 7mg/day respectively.

I was unable to do a direct comparison with other studies as the methods used to quantify DM risk associated with cumulative GC dose varied across studies. The pattern however, was consistent with published literatures that looked at the effect of oral cumulative GC dose on DM risk. One large nested case-control study reported that newly diagnosed GCA patients exposed to prednisolone 30mg/day had almost a 5-fold increased risk (Adjusted OR=4.7; 95%CI: 2.8-7.8) of developing DM compared to those with a low average daily prednisolone dose of 5mg/day [281]. The study also reported a trend of increasing risk with increasing cumulative prednisolone dose, with a 2-fold increased risk of DM observed in the highest cumulative dose category (≥20g) compared to the lowest dose category (≤3g) [281]. One population-controlled Danish study looking at 1849 vasculitis patients, of which 91% of the patients were diagnosed with first-time GCA, reported that median cumulative prednisolone-equivalent dose (PED) dose of \geq 5.6 grams was associated with a hazard ratio for DM of 1.6 (95% CI: 1.02-2.5), and a 30% increased DM risk per 10mg increase in average daily oral prednisolone equivalent dose (HR: 1.3; 95% CI: 1.01-1.80), all within the first year [245]. Another CPRD cohort study also reported that newly diagnosed GCA patients had a 1.4-fold increased risk of developing DM compared with non-vasculitis patients, though the authors did not report any information on the total cumulative dose or duration of GC use in this patient population [258].

One retrospective cohort study using US claims data reported that the risk of newonset DM rose by 5% (HR=1.05; 96%CI: 1.03-1.07) with each gram of oral GC exposure in patients with newly diagnosed GCA [264]. Results from the extended timevarying Cox regression model in my study indicated a 2% (HR=1.02; 95%CI: 1.02-1.03) increase in DM risk with each gram of oral GC exposure in these patients. It is important to note though, that the authors in this study had full access to all GC prescriptions from the claims data, as indicated by the median initial oral GC dose of 40mg/day; while the median initial oral GC dose in my study was only 20mg/day. Results from my WCE analysis suggest that oral GC dose taken in the most recent five months of treatment had a significant impact on DM risk, with the highest impact being at month four. The rolling cumulative dose model also showed a similar pattern, with the highest impact of DM risk being at month three. This pattern is consistent with other published studies for GCA patients. One large nested case-control study reported that majority of DM cases associated with GC use occurred within 2 years following initiation, with over 40% of DM cases developing during the first year [281]. Another Danish study reported that the incidence risk ratio of new-onset DM was 7.0 (95% CI: 5.2-9.3) in their GCA cohort during the first year of observation when compared to the general population. Beyond the first year, they reported that the incidence rates for DM were not significantly increased [245]. In another large observational study of 5,011 GCA patients, the incidence risk ratio of DM was 1.4 (95% CI: 1.2-1.7) as compared to matched non-GCA patients [263], of which the median time for the occurrence of DM in the GCA group was one year from the start of treatment. While all these published literatures support the hypothesis that the risk of developing DM is highest within the first year of GC use, my study findings have added on insights on the specific timing when the risk is highest, which is likely to be within the first six months of therapy initiation.

Among the confounders considered, family history of DM was associated with nearly a 2-fold higher DM risk across groups when adjusting for cumulative dose and non-oral GC use during follow-up using the extended time-varying Cox regression, rolling cumulative and WCE models. The result is consistent with published studies that examined the association between family history of diabetes with type 2 DM, of which it was reported that those with a first-degree relative having DM had a more than two-fold excess risk [301-304], especially among those of European ancestry [305]. Results for non-oral GC use during study follow-up were inconclusive from my study. The extended time-varying Cox model and rolling cumulative model suggested an increase in risk: 7% for PMR patients and 30% for patients with GCA (with or without PMR), but the WCE model suggested a negative association (HR: 0.41 & 0.37) for the PMR group, and GCA (with or without PMR) group respectively. One of the possible reasons for this discrepancy is related to the recency of oral GC exposure. The hazard ratios for the rolling cumulative model was calculated based on the coefficient of two most recent consecutive months since diagnosis for both groups, while the WCE model was calculated based on the most recent 10 months for PMR and 5 months for GCA (with or without PMR). There is a possibility that the risk is higher in the initial months since PMR/GCA diagnosis, when given concurrently with higher oral GC doses but declines over time. This however, was not explored in detail, as use of non-oral GC during study follow-up was only accounted for as a binary (yes/no) variable in my study. Non-oral GCs (including inhalers, topical preparations, eye drops and injections), as stand-alone prescriptions, have been reported to have no association with the development of incident DM in one large nested case-control population study in the UK, but the authors did not evaluate the risk of DM association with these treatments when given concurrently with oral GC therapy [181].

One issue of interest from the results (Table 6.1 in the results chapter) was that PMR appeared to have a lower prevalence in the most deprived index of multiple deprivation (IMD) quartile (quartile 4), but similarly common across the first three quartiles. This is an indicator that PMR may be under-represented in the quarter of the population living in areas with greater levels of socio-economic deprivation. One of the possible reasons is that PMR patients in my study population were incident cases; therefore it is possible that those with better socio-economic status (SES) were presented to their general practitioners (GPs) in the initial stages of their disease because of better access to health care and better health awareness. People in the community with lower SES have been reported to have poorer physical and mental health and poorer access to health care [306]. One study investigating the association of PMR with SES in primary care reported that no association was found between PMR and SES [307]. This study however, included all PMR patients (not just incident cases), and involved only eight GPs in North Staffordshire. Since there were only eight practices included, there may not have been a sufficient range of SES for the effect to be apparent. Therefore, the results from that study may not apply to my study population.

7.2.2. GC Prescribing Pattern in Primary Care

Relatively little is known about real-world oral GC prescribing patterns of PMR and/or GCA in primary care. While there have been a few surveys and studies reporting the starting dose, average daily dose, cumulative dose and duration of GC use, none of these studies described the pattern of GC use over time.

7.2.2.1. GC Prescribing Pattern in PMR Patients

PMR patients are usually diagnosed and managed in primary care if there are no complications. One study consisting of 183 PMR patients in the UK reported that 83% of these patients were managed exclusively in primary care without reference to specialist opinion [308]. Another recent survey of UK GPs reported that of the 1249 respondents, only 6.4% (n=80) would refer all PMR patients to secondary care as a matter of routine care [309]. These studies support the idea that in the UK, PMR patients are generally managed in primary care; which was also reflected in my study findings as the prescribing pattern of oral GC corresponded fairly well to the 2015 European League Against Rheumatism (EULAR) guidelines suggesting that GC was mostly initiated in primary care for these patients [22].

The median monthly oral starting GC dose for PMR patients in my study was 500mg/month (or 16.7mg/day), which corresponds well to the recommended starting dose of 12.5 - 25mg/day by the 2015 EULAR guidelines [21]. The median first GC dose was 15.0mg (IQR: 10.0-20.0mg), which was in agreement with another CPRD study that also reported a median first GC dose of 15.0mg (IQR: 8.0-21.0mg) [12]. Helliwell et al, in their qualitative study exploring the management of PMR among UK GPs, reported that 56.4% (n=704) of study respondents would initially treat PMR with either 15mg or 20mg of prednisolone [310]. Their survey finding was consistent with the results from my study.

The median average daily dose of GC prescribed for the entire follow-up duration in the practice was 6.6mg/day (IQR: 5.0-9.6mg/day); while the total cumulative dose for the entire follow-up duration in the practice of approximately 4.5 years was 4.1g (IQR: 2.0-7.6g). These findings were similar to the above-mentioned CPRD study of incident PMR patients, of which the authors reported a median average daily dose of 6.0mg (IQR: 4.0-9.0mg) and a median total dose of 4.0g (IQR: 2.0-8.0g) [12]. In addition, I also looked at the cumulative dose for the first two years of follow-up. The cumulative dose over two years was 3.6g (IQR: 1.8-5.1g), which was approximately 88% of the

total cumulative dose for the entire follow-up duration in the practice, suggesting that the extent of GC use was highest within the first two years of follow-up. The median duration of treatment in my study population was approximately 1.7 years (IQR: 0.7-3.5 years). The above-mentioned CPRD study also reported similar findings, with the median duration for total GC treatment time of 1.9 years (IQR 1.0-4.0 years) [12], while another study that looked at PMR patients in three UK GPs reported a median treatment duration of 1.4 years (IQR: 0.8-2.4 years) [308].

When compared to the 2015 EULAR guidelines, patients in primary care appeared to have a much steeper tapering regimen. GC doses were halved to approximately 8.1mg/day (from the initial 16.7mg/day) after merely three months of treatment. GC doses were tapered off rapidly until the ninth month, of which doses were then tapered more slowly than the EULAR guidelines suggest. This slower tapering pattern at the end of the first year may be due to a few reasons. Firstly, it is possible that those were the minimum effective doses to keep patients in disease remission, and that the dose was held at a particular level for a subset of patients in order to keep their symptoms under control. Secondly, patients or clinicians may have been reluctant to stop the treatment for fear that symptoms may restart, and while they may be concerned with adverse events associated with high GC dose, they may have the perception that the risk of adverse events associated with low GC dose is very low. Therefore, they are not in a hurry to stop treatment beyond a certain level of low GC dose. Thirdly, it is also possible that some of these patients eventually developed a relapse or flare in disease, requiring a significant dose increase. At the group level this would increase the average monthly dose. A few studies reported that the relapse rates were associated with higher initial GC dose and faster GC tapering rates [311, 312]. Because of the concern of over-rapid tapering of GC doses leading to increase in relapses, Kirwan et al suggested a new regimen that minimizes the risk of relapse. They recommended a starting dose of 15mg daily for six weeks then 12.5mg daily for six weeks, then a 10mg maintenance dose for one year, with a 1mg/per month taper thereafter if needed [313]. The authors also reported a 20% lower rate of relapse at two years using this new regimen, as compared to another cohort that used a more rapid dose reduction regimen in line with current British Society for Rheumatology (BSR) recommendations [233]. The issue of how GCs should be tapered in the safest and most effective way with minimal relapses however cannot be determined in this study and is beyond the scope of this PhD.

GCA patients, on the other hand, are more likely to be managed in secondary care in the initial stages of their disease, as they are usually referred for specialist review and diagnostic confirmation, as recommended by guideline [7]. Though the actual rate of referral is unknown, one survey showed that 35.6% of 1251 UK GPs would refer patients with suspected GCA to secondary care without initiating GCs [309]. The response of this survey seemed to reflect my study findings - of which 39.5% of first oral GC prescription did not have any information on dosing and frequency, suggesting that these patients may have been managed in secondary care. The survey also reported that 49.7% would initiate GCs before referring the patients out to secondary care, of which the most common initiating dose was 60mg of prednisolone [309]. One of the potential challenges with this group of patients is that, the first GC prescription may be available in primary care, but subsequent prescriptions will not be available, as they will be seen in secondary care. The first prescriptions for these patients are also likely to be shorter in duration, as the main purpose of these prescriptions was to keep their symptoms under control while waiting to be seen by specialists in secondary care. It is therefore not surprising that the prescribing pattern among patients with GCA (with or without PMR) seemed to be much lower than recommended guideline doses in my study. The median monthly starting dose was 560mg/month (or 18.7mg/day), and the median first GC dose was 20.0mg (IQR: 15.0-35.7mg) in my study. The results are consistent with another study that looked at a subgroup of UK patients in primary care, of which the authors reported a median GC starting daily dose of 20mg/day (IQR: 10-40mg/day) [314]. The median average daily dose of GC prescribed for the entire follow-up duration in the practice was 9.8mg/day (IQR: 6.5-17.3mg/day); while the median cumulative dose for the entire follow-up in the GP practices duration of approximately 4.4 years was 5.3g (IQR: 2.0-7.6g). The median cumulative dose from my study was comparable to another CPRD study involving only incident cases of GCA, of which the authors reported a median cumulative dose of 5.1g for patients up to more than two years of follow-up [77].

The cumulative dose over two years was 4.2g (IQR: 1.7-6.9g), which was approximately 79% of the total cumulative dose over the entire follow-up duration in the practice, suggesting that the extent of GC use was highest within the first two years of follow-up, as with the PMR group. The median duration of treatment in my study population was approximately 1.5 years (IQR: 0.4-3.5 years).

When compared to the 2018 EULAR guidelines, patients in primary care appeared to have a much lower treatment regimen, though the results do indicate a tapering trend over time in accordance with the guidelines. In my study, 57.1% of first oral GC prescriptions did not have a quantifiable dosing instruction, for these prescriptions, the assumption of "one tablet per day" was applied if no other information was available. This would have resulted in a substantial underestimation in dose as nearly two-thirds of the prescriptions were 1mg and 5mg prednisolone tablets, whereas the expected initial dose would be between 40-60mg. Another important observation is that since patients were unlikely to be prescribed with 1mg or 5mg tablets if they were to be on high dose GCs (i.e. 40-60mg), therefore there is a possibility that the "first" oral GC prescription recorded in CPRD may not truly be the "first" prescription, but more likely to be the "first prescription, but more likely to be the "first prescription upon discharged from secondary care".

7.2.3. Challenges Associated With Primary Care Prescription Databases

There were many challenges associated with data cleaning and analyzing complex primary care prescription databases such as CPRD; and while some were related to the complexity and variability in abbreviations or prescribing terminologies, some were simply related to transcribing errors such as misspellings. I have summarized here some of the major challenges encountered during the database cleaning process:

7.2.3.1. Misspellings

It was fairly common to come across misspellings in the free text section of the prescription database, as with other types of clinical text [315, 316]. Most of them were typos. Some of the misspellings were insertions of unnecessary alphabets, such as "1 ddaily" or "1 daily6" or "1 dailly", and some were deletions of alphabets such as "1 dail" or "1 dail;", while some others were merely straightforward typos such as "as directde" or "as direced" etc. There were also examples where the spacing between words was omitted, such as "1aday" instead of "1 a day". Most of these misspellings were either keywords of dosing or frequency, e.g. "1 mnae" instead of "1 mane"; or "notce" instead of "nocte" which was a challenge when using computer-coding system, as the system failed to recognize the misspelled keyword, resulting in that particular prescription being excluded or miscategorized. Another major misspelling issue was associated with the letter "O" and the number "0". There were multiple prescriptions in the database where "OM" was written as "0M", of which it was unrecognized by the computer coding system. To address this issue, manual screening was done to appropriately extract relevant prescribing data. The complexity of this problem is

reflected in the instruction "as directed", as there were 44 different typos and misspellings for this one specific instruction.

7.2.3.2. Ambiguity in interpretation

There were multiple cases where the prescription text may be interpreted in more than one way; and to address this problem, clinical judgment played a crucial role. For example, a prescription of "3 per day" may be interpreted as either 3 tablets taken once, or 1 tablet taken 3 times. Since GCs tend to be taken in a single daily dose, therefore the first intuition would be to consider it as 3 tablets taken once. In this case, the pattern of administration within a day may not be important for data analyses, as the goal was to look at the average dose per day and cumulative dose; therefore in this case, it did not matter which assumption was used to extract the information. There were, however cases where the assumption mattered as different interpretations would result in major differences for the cumulative dose and average dose per day. One typical example is a prescription of "1/2 alt days". This may be interpreted as "1, then 2 tablets taken alternately each day" or "half a tablet taken on alternate days". The interpretation of the prescription in this case would make a huge difference (1.5 tablets per day versus only 0.25 tablets per day). The first intuition was to assume that the first assumption of "1, then 2 tablets taken alternately each day" to be more likely, as it would make more sense for the prescriber to write "0.5 EOD" instead of "1/2 alt days" if the patient were really meant to take "half a tablet on alternate days". There were also similar prescription styles of "2/1 alt days' and "2/3 alt days" etc, of which the first assumption would again makes more clinical sense, as it would not make sense to write "2/1" instead of "2". In addition, it would not make sense to request patients to divide tiny prednisolone tablets into 3 portions, and to take 2 of the 3 portions on alternate days. To further confirm that the first assumption was more likely of the two, 5 random sets of the "quantity of tablets prescribed" were selected and the total duration of treatment was calculated (by dividing the "quantity of tablets prescribed" with "number of tablets per day") and compared with the dates of the next consecutive visit. For example, if the prescription was "1/2 alt days" and the patient was prescribed with 84 tablets; then the 2 possibilities would be either 84/1.5 = 56 days or 84/0.25 = 336days. This would then be compared with the dates of the next consecutive visit for the patient, and if the next scheduled appointment was to be 2 months later, then this would correlate with the first assumption. All 5 random samples supported the first assumption, therefore it was concluded that the interpretation of this prescription style would be to take "x, then y number of tablets alternately each day" unless specified otherwise.

7.2.3.3. Varying dosages and tapering regimens

Another major challenge was the variation in prescribing patterns, especially for cases where medications required administration at different specific time, e.g. "take one tablet at 10am" or "take one at teatime". The specificity of these instructions resulted in this type of prescription not being captured by the computer coding system, therefore, the only option was to manually screen for them. There were also some prescriptions that required tapering in doses, which was also difficult to capture using the computer coding system, as the doses and dosing frequency were not fixed.

7.2.3.4. Non-specific instructions

As discussed in the results chapter, approximately half of the prescriptions had a "as directed" or no GC dosing instructions. Each GP has a practice formulary with a clinical drug database to facilitate electronic prescribing. For most medications with a fixed dosing regimen (e.g. anti-hypertensive medications, anti-diabetic medications or antibiotics), a default instruction is usually pre-set to minimize prescribing errors. These default instructions can be changed by the prescriber if patients were to take the medication differently to the default. The challenge with treatment regimens that require tapering such as GCs is that there is no fixed dosing regimen, therefore the default instruction in the clinical drug database is often written, "as directed". GPs are encouraged to change these "as directed" prescriptions into specific instructions, but this remains a challenge for the following reasons. Firstly, the dose may have been specified by a hospital consultant if the patients is under secondary care, so more often than not, patients are told to adjust their medication as directed by the specialist. Therefore the "as directed" may mean "reduce the dose by 1mg if blood test is normal" and the patient is expected to make the necessary adjustment once the result is known; or in other cases, the "as directed" may refer to "reduce the dose by 1mg if the pain is gone or when symptoms have resolved". Secondly, when a prescription is reissued, the easiest and fastest way for a prescriber to do so is by clicking on the previous prescription for that patient and enter reissue. Although the prescriber may change the instructions if needed, it is often presumed that the patient will continue taking the way they have been previously been instructed on. Thirdly, there is a possibility that GPs record the instructions as "free text" in the clinical notes (which is not accessible to researchers), and instructed the patient verbally on the specified dosing instructions; and with GCs being one of the most common medication that requires tapering, the pharmacist may not query the GP on specific instructions upon receiving the prescription.

Data cleaning of the prescription database will remain a challenge because of its size and complexity. Not only does it require adequate computer hardware and software, it also requires experience in data management and expertise in computer programming with clinical input. An effective data cleaning approach that fulfils the following requirements would be helpful to promote efficiency and minimize errors [317]: firstly, it should be able to detect and remove all major errors and inconsistencies, especially during the integration of multiple sources. Secondly, the approach should be supported by tools that require minimal manual inspection and programming effort. Thirdly, the algorithm for data cleaning should be specified in a declarative way and be usable across the various disease domains as well as for query processing. One potential solution is the incorporation of an automatic misspelling detection and correction system with the rule-based natural language processing (NLP) method. While rulebased approaches may also be implemented using statistical softwares such as R or Stata, they are only applicable if the extent of misspellings and typos is already known. This is however, a challenge, as the range of possibilities for any misspelling and typo errors are unknown unless screened manually, as discussed above. One study that applied their spell checker to three different types of free-text data: clinical notes, allergy entries and medication orders reported a detection performance of up to 94.4% and correction accuracy of up to 88.2% [315]. Another study incorporated a few spellchecking algorithms for NLP in vaccine safety and reported a sensitivity of 74% and specificity of 100% [318]. The following spell-checking algorithms were proposed in their study: firstly, the "metaphone" algorithm which was used to search for similar sounding words. Secondly, the "header" algorithm that looked for words with the same first four characters. Thirdly, the "N-gram" algorithm that allowed mid-string searches. Fourthly, the "transposition" algorithm that search for words where any two characters are switched. Fifthly, the "deletion" algorithm that search for word matches by sequentially inserting a wildcard character in the misspelled word to simulate a character deletion. Next, the "insertion" algorithm that searched for word matches by sequentially deleting a character in the misspelled word to simulate a character insertion, and lastly the "substitution" algorithm that searched for word matches by simulating character substitution. Results from these studies showed promising results, but it will still be some years before the use of automated misspellings checkers in NLP can fully be implemented in primary care research due to strict governance and access restrictions on the use of free text.

7.2.4. Comparison Between CPRD-derived Data and Primary Care Data

There were 2 discrepancies found between CPRD-derived data and the data I extracted from primary care for the "number of tablets per day" variable. Information for this variable was mainly extracted from the "dosage instructions" variable, and any instructions that involved "X mg" (e.g. 40mg daily) was coded as X number of tablet (e.g. 40 tablets) in the CPRD-derived data, without taking into consideration the tablet strength. In this example, the total number of tablets should have been 8, after taking into consideration that it was a 5mg prednisolone tablet, and not 40. This misclassification would have led to an overestimation of the daily dose, as the number of tablets per day will eventually be multiplied with the tablet strength and PED to obtain the final daily dose. The other discrepancy was related to prescriptions that involved "alternate days" dosing (e.g. 1/2 alt days"). This may be interpreted as "1, then 2 tablets taken alternately each day" or "half a tablet taken on alternate days", of which the CPRD-derived data opted for the latter. Both interpretations are plausible, but the first interpretation was clinically more likely. Details on how these instructions were interpreted in this study have been presented in Section 7.2.3.2. While the first type of misclassification would have resulted in an overestimation of the daily dose, this type of misclassification would have resulted in an underestimation of the daily dose for the CPRD-derived data, and the balancing out of these two effects might make any errors in the calculation of daily dose harder to spot in the group-level data.

Another problem with the CPRD-derived data was inconsistency in the application of the assumption of "once per day" dosing if no specific instruction was given. The assumption was only applied in some cases and there was no clear pattern on the criteria of which the assumption was applied to. Since the CPRD-derived prescription data was mostly managed using pre-defined computer algorithms and dictionaries with minimum human manual screening, it is likely that misspellings may have contributed to this inconsistency especially given that misspelling errors were common in the prescription database, as discussed in section 7.2.3.1.

While a few discrepancies have been identified between CPRD-derived data and the data I extracted from primary care, they are unlikely to have any significant impact on the analyses. Even if there were to be some misclassifications in the number of tablets (e.g. 40 tablets instead of 8 of the 5mg tablets), most researchers would eventually apply a cap in the maximum daily dose based on clinical judgment; therefore the impact of this overestimation in daily dose would have been mitigated. Alternate day dosing medications are relatively uncommon – it is usually only used when a drug is

being tapered off (e.g. GC), and even so, the duration of a tapered dose regimen is relatively short; therefore any misclassification in alternate day dosing prescriptions would be unlikely to have a major impact on the results, especially if the study has a large sample size. In addition, alternate-day GC therapy in GCA patients has been found to be associated with high rates of flares compared with daily treatment (70% versus 20%), therefore it is not recommended for use in clinical practice [319].

Results from my study indicate that CPRD-derived data is reliable to be used for analyses. Although there were some misclassifications in some important variables (e.g. number of tablets per day and frequency), the impact of these misclassifications was likely minimal. The findings from my study relating to GC starting dose, average daily dose, cumulative dose and treatment duration were all comparable to other published literatures, suggesting that there is no major difference between the CPRDderived data and my extracted data, which was, comparatively, a lot more labor intensive due to some involvement of manual screening of the free text in the "dosage instructions" variable.

7.3. Strengths and Limitations

7.3.1. Strengths

This is to date the largest population-based longitudinal study specifically examining the incidence of DM in PMR and/or GCA patients. The novel rolling cumulative dose and WCE methods used in this study has allowed the quantification of DM risk associated with dosage, duration and timing of oral GC exposure to be determined, which is crucial to facilitate decision making in clinical practice. To date, only conventional methods (e.g. cumulative dose, current dose) have been used to determine the risk of DM associated with GC use. There are limitations with the use of conventional methods as imperfect assumptions are often made with regards to the timing of GC use in relation to the outcome. For example, use of "current dose" models take into consideration only the existing dose taken, but ignore the impact of past doses, while "recent dose" models only take into consideration the most recent dose (e.g. past 30 days) but ignore all doses taken outside that defined time frame. Another typical example is the use of "cumulative dose" models, of which the model assumes that doses taken years ago have the same impact as doses taken recently. In contrast, the rolling cumulative dose model and WCE allow the assessment of risk based on the dose, duration and recency of GC dosing.

The use of CPRD as one of the largest routinely collected electronic health data that spanned primary care practices across UK and linked to the mortality registry is an important strength of this study. In addition, the data is also contemporary and up-todate (generated after 1/1/1998); therefore results from this study provide a better idea of risks associated with current clinical practice. This would have been very different if data from the early or mid-nineties were used, as the diagnoses and treatment could have been very different then. Another important strength is the representativeness of this database. When compared with the 2011 UK census, CPRD patients were reported to be broadly representative of the UK population in terms of age, sex and ethnicity [9]. Not only is it a generally good representation of the general population in the UK, it also has a relatively large sample size with long-term follow-up data, which is crucial for study of relatively rare diseases such as PMR and GCA to allow a more efficient estimation of association between exposure and outcome. The electronic aspect of prescribed medications not only decreases the rate of possible data entry errors, but also allows a more efficient way of maintaining patient's confidentiality and privacy.

CPRD has been widely used in research for decades, and a number of methods have been used to assess validity, including internal and external validations [320]. A systematic review of validation and validity of diagnoses of different disease domains in CPRD (e.g. musculoskeletal system, respiratory system, circulatory system etc), reported a high estimate of validity, though the quality of reporting was inadequate to permit a clear interpretation [320].

The retrospective aspect of this study allows access to information readily available; therefore the study would only need a relatively short period of data construction and data cleaning, as compared to prospective studies that requires information to be collected as follow-up is carried out.

7.3.2. Limitations and Potential Biases

This section will discuss a few limitations and potential biases that were considered within the data sources and related to the design of this study. These potential biases were addressed carefully as they may influence the validity of the estimates of the true relationship between the study exposure and outcome measure.

123

7.3.2.1.1. New User/ Inception Cohort Study Design

One of the issues relating to the inception cohort or new user study design and the exclusion of patients without GC prescriptions (i.e. because PMR and GCA diagnoses would be unlikely) was that there were no "controls" without GCs. The presence of a control group would have allowed a more robust conclusion to be made - that any change observed in the active GC treatment group is more likely due to the treatment being studied, rather than to other factors. The use of the inception cohort design, however, was appropriate for my diseases of interest, as patients with PMR and/or GCA would almost be exclusively treated with GCs upon diagnosis. Therefore, the comparison between a high dose regimen and a low dose regimen would be a good depiction of real-world clinical practice. One alternative is to use the case-control study design, but there are challenges associated with the use of this study design. One of the biggest challenges is to identify controls from the general population as a comparator. These controls are likely to be different to patients diagnosed with PMR and/or GCA, and would also be less likely monitored for DM by the GPs. Another limitation is the lack of general population sample to allow comparison of those getting type 2 DM anyway. To date, there is only one study that looked at the incidence of type 2 DM in primary care between 1991 and 2000 using CPRD [321], of which a comparison may be inappropriate as the time frame of my study was between 1998 and 2016. Even if a comparison were possible, it would also be extremely difficult to clinically differentiate between the diagnosis of "steroid-induced DM" and "non-steroid induced DM". It is therefore possible that some of the incident type 2 DM cases recorded in primary care were actually steroid-induced DM.

Another limitation is the reduced sample size because only those who were newly diagnosed with PMR and/or GCA and had at least one GC prescription during the study period can be included. However, the final sample size of my study population (n=23,048) was still large compared to other studies.

7.3.2.1.2. Observational Study Design

One of the key limitations relating to the use of an observational study design is the inability to deduce a cause-and-effect relationship between the exposure and outcome. While my data do show a fairly convincing association between higher GC doses and DM risk, I am unable to prove conclusively that the association is causal. The presence of unmeasured confounders cannot be ruled out, which I will be addressing in Section

7.3.2.4. The use of an observational study design however, is the most practical and feasible for determining the association between GC dose and DM risk for the following reasons. Firstly, both PMR and GCA, although not uncommon, are still relatively rare compared to other common diseases like coronary heart disease or asthma. It would therefore be unpractical to use a randomized controlled trial (RCT) study design, as patient recruitment would likely take a long time. Secondly, DM is not an acute disease; therefore time is needed for the event to occur, of which it would be very time-consuming and expensive to conduct using the RCT study design. Thirdly, using routinely collected primary care data such as CPRD has many advantages, which include a large sample size, long-term follow-up data and provides information on "real-world" use and practice.

7.3.2.2. Misclassification Bias

7.3.2.2.1. Misclassification Bias of Diagnosis

Misclassification bias is another common issue of concern in routinely collected health data. The criteria for classification of PMR/GCA by the British Society for Rheumatology (BSR) [7, 24] were not available in the dataset for diagnosis confirmation; therefore the diagnosis of PMR and GCA were solely defined based on recorded physician diagnosis in primary care.

Diagnosis of PMR and GCA is a challenge in clinical practice. For PMR, overlap in symptoms with other co-morbidities, presence of multiple co-morbidities and the lack of an established diagnostic test have been identified as contributors to diagnostic uncertainty [310]. For GCA, in addition to its relative rarity compared to other musculoskeletal diseases, primary care GPs are often faced with the frequently non-specific nature of early GCA symptoms, as well as its high prevalence of similar symptoms in the general population [32, 322], which often leads to a delay to diagnosis or misdiagnosis [322-324]. An additional complexity of primary care data is that the absence of a Read code for the disease is interpreted as an absence of the disease itself, resulting in a higher positive predictive value [320], but lower sensitivity. One of the possible reasons may be due to variations between GPs in coding diagnoses in the patient's electronic health record, or if the information was entered as free text, then it may be missed out on. There is also a possibility that patients with mild disease or with atypical presentations might be less likely to be diagnosed [260].

To date, there has only been one validation study for the diagnosis of PMR, which reported a wide variation in the diagnosis of PMR among general practitioners and hospital specialists [325]. It is worth noting though, that the study was done in the early 1990s and only represented a small subset (Norwich) of the UK population. The diagnosis of GCA in CPRD has been previously validated by comparing electronic CPRD records with typical symptoms and clinical response to GCs, and the authors reported a validity of 91% [8]. To minimize the likelihood of PMR and GCA misclassification in my study, only those with at least one GC prescription were included in the study. Since both PMR and GCA are inflammatory diseases with clinical symptoms, it is therefore unlikely that patients would be left untreated if they truly had the disease. Furthermore, the median age of my study population was in the mid-seventies, with nearly two thirds of them being female, which was consistent with other published studies [260, 258, 11, 298].

7.3.2.2.2. Misclassification Bias of Outcome

As with PMR and GCA, the diagnosis of DM was also primarily defined based on recorded physician diagnosis in primary care. The parameters for criteria classification by World Health Organization (WHO) [326] were largely missing, as specific markers (e.g. HbA1c, fasting plasma glucose) are not routinely collected in primary care settings.

Another major challenge pertaining to the outcome is its definition. The outcome of interest for this PhD was GC-induced DM. There is however, no appropriate measure to validate this, thus there is a possibility of misclassification of outcome. One of the precautionary steps I did was to only take into account cases of new-onset DM occurring within the initial two years of follow-up. Non-steroid induced type 2 DM is well documented as a progressive disease, primarily characterized by a progressive decline in beta-cell function and worsening of insulin resistance over time [327]. Various studies have reported that the development of non-steroid induced type 2 DM generally occurs between 29 months to 10 years in those with baseline impaired fasting glucose reading, depending on risk factors [328-332]. Furthermore, the highest incidence of type 2 DM is reported to be between 40 and 69 years of age in the National Diabetes Audit, UK [333]; and between 45 and 64 years of age by the Centers for Disease Control and Prevention (CDC) [334] in the US, suggesting that those who were at risk of developing DM is likely to have had developed DM in middle age or earlier; while the median age for my study population was 74 years old. It is therefore reasonable to assume that the development of new-onset DM in my study population was likely to be a consequence of GC therapy (for at least over 50% of the patients), and not a phenomenon pathogenically linked with the traditional risk factors for DM. One could argue that there might be a subgroup of predisposed patients who progressed from latent to manifest DM exacerbated by high-dose or prolonged GC therapy, while this is plausible, it is unlikely to involve the whole study population considering the large sample size and magnitude of effect in DM risk. Another possible misclassification relating to the outcome was that patients may have had undiagnosed DM, but when prescribed with GCs, DM testing is likely to have been done as part of routine screening due to its known risk. As a result, DM was discovered, and although not due to GC treatment, it may have been considered as related to GC use in my study just because routine screening revealed the condition. While cases as such may be possible, it is unlikely to involve the whole study population and would be likely to apply equally to the low-dose and high-dose patient groups.

The quality of recording of DM in CPRD has been evaluated in a cross sectional study [335], of which the effect of miscoding and misclassification on incidence estimates was evaluated. Results from the study showed that the incidence of diabetes did not increase and was consistent with other reports [336, 251] if only the diagnoses codes were used. In my study, only the diagnoses codes were used. In addition, only practices that met the quality marker (up-to-standard) date were included.

7.3.2.2.3. Misclassification Bias of Exposure

The possibility of misclassifying the main exposure (i.e. oral GC dose) also cannot be ruled out. Prescription records in CPRD merely indicate that the drug has been prescribed and dispensed, but it does not provide any information on how much of the drug was actually taken by patients. It is possible that patients had their prescriptions filled, but never taken, or in some cases taken initially and then discontinued, or taken as needed or intermittently. My analyses showed that approximately half of the prescription database for oral GCs had a "as directed" or "unknown" instruction. Hence, I assumed that one tablet was taken per day to calculate the daily dose if no other information was available. The application of this conservative assumption may have resulted in an underestimation in DM risk associated with the dose prescribed. In any case, the resulting misclassification of exposure is likely to be non-differential, resulting in a bias toward the null – which is of less concern, given that the results indicated statistical significant associations with clinically important estimates.

Another limitation of my study relating to the main exposure is the inability to account for medications prescribed by specialists in secondary care. This includes the non-oral GCs (i.e. intravenous, intra-arterial or intramuscular GCs) that would have been likely used in my patient population, especially during initial treatment or during disease flares. In certain circumstances, use of non-oral GCs may have been recorded, but it would only represent a very small sample, which does not give a clear reflection on the actual total use of non-oral GC use at the GP level. However, in practice, oral GCs remain the mainstay of treatment, and any other mode of administration is likely to be for a short duration; and while it may affect the magnitude of the risk, it is unlikely to alter the overall conclusion.

Another possible limitation relating to GC exposure was the use of monthly dose regimens in the assessment of DM risk. The monthly dose regimens were used to make the analyses more tractable, but the use of weekly or daily dose regimens may reveal more details on the actual risk of developing DM.

7.3.2.3. Time-related Bias

Time-related bias, such as immortal-time bias, defined as a period of cohort follow-up or observation time during which the outcome of interest cannot occur [337], is another common issue in pharmacoepidemiology studies. To prevent the introduction of timerelated bias, I used a time-dependent definition of exposure that properly classifies the appropriate duration of person-time of study follow-up as "exposed" or "unexposed", according to the date of start and end of drug use. In other words, the start of follow-up was unrelated to the prescribed GC medication. The same set of inclusion and exclusion criteria were applied to the entire study population. Another challenge often associated with treatment effects is misclassification of duration of use as treatment use and dose often vary with time. This issue was addressed by using the extended time-varying Cox model, rolling cumulative model and the weighted cumulative dose analysis, of which oral GC dose was allowed to vary over time.

7.3.2.4. Unmeasured Confounding

Unmeasured confounding is defined as confounding associated with variables not included in the data under study, leading to confounding bias [338]. This is particularly prominent in routinely collected data, as some variables needed for the analysis may not be routinely collected in clinical practice. One particular type of unmeasured confounding is confounding by indication, which refers to a determinant of the outcome

parameter that is present in people at perceived high risk or poor prognosis and is an indication for the intervention [339]. This means that differences in treatment may partly originate from the differences in indication for medical intervention such as the presence of risk factors for particular health outcomes. It is however, not a major concern in this study, as the selection of GC use is unlikely to be related to DM risk. There is however, still a possibility that clinicians may have been less willing to prescribe higher GC doses, or use a more rapid tapering regimen to control disease activity in obese patients who are at higher risk of developing DM. The reverse could also be possible - clinicians may be more likely to give higher GC doses to overweight or obese patients on the basis that higher body weight corresponds with higher drug clearance, and therefore an increased dose is required. This is known as reverse causation, of which a predisposition to DM (or a DM diagnosis) may have resulted in a need for higher GC doses. This is however, very unlikely as GC clearance is correlated to lean body weight rather than adipose weight as adipose tissue has little metabolic activity [340]. Normal-weight patients have a total body weight consisting of lean and adipose body weight in an approximate 4:1 ratio, while the lean: adipose weight ratio in obese patients has been reported to be 3:2 [341]. This indicates that obese patients have a lower lean weight, of which an increased GC dose is not required, as a higher drug clearance is unlikely. Another possibility is the issue of disease severity, which is particularly difficult to measure precisely in routinely collected health data. Because GCs are the mainstay of treatment for PMR and GCA, it was not possible to disentangle the effect of the disease from the effect of the treatment. It is however known that systemic inflammation itself can also induce a state of insulin resistance [265, 97] so it is plausible that the inflammatory disease itself (PMR or GCA) could have contributed to the risk of DM. In addition, some medications commonly prescribed to elderly patients may also contribute to the risk of DM (e.g. thiazide diuretics, betablockers, niacins and statins).

7.4. Recommendations

It is indisputable that patients with PMR and/or GCA would need to be treated with GC despite the increased risk of developing DM in the initial stages of their treatment. The challenge is that they are not just treated with GC, but often initiated on high starting doses as well, with longer duration of treatment compared to patients with other systemic inflammatory conditions [342]. While the GC doses are tapered gradually over time, these patients are often still susceptible to the risk of developing GC-induced DM, as reported in this study. Therefore, good DM management in this patient population is essential as long standing DM is often associated with microvascular and macrovascular complications. There is however, little evidence and guidance on how patients with GC-induced DM should be managed, especially among the elderly, which represents my patient population.

A multidisciplinary EULAR task force on GC therapy has developed a few recommendations for the management of medium to high dose GC therapy in rheumatic diseases. "Medium to high dose" GC therapy was defined as > 7.5mg but \leq 100mg PED daily [342], which fits well into the range of treatment doses that PMR and/or GCA patients usually receive in the initial stages of treatment. Although the recommendations were not specific for the management of GC-induced DM in PMR and/or GCA patients, some of the general approaches were still applicable and may be adopted by GPs in primary care settings. The recommendations are as follow [342]:

- 1. Explain to patients and their family and/or carers the aim of medium/high dose GC treatment and the potential risk associated with such therapy.
- 2. Discuss measures to mitigate any risks, including diet and regular exercise.
- Provide an accessible resource to promote best practice in the management of patients using medium/high dose GCs to GPs, such as having a website on the benefits and risks of GC treatment, advising how to manage inter-current illnesses and acute situations.
- 4. Before starting medium/high dose GC treatment, any co-morbidities predisposing to adverse events such as DM should be given consideration. Patients with these co-morbidities require tight control (i.e. more intensive monitoring and adjusting medication, if needed) to manage the risk/benefit ratio. Glucose monitoring before start of therapy and during therapy is also advised.
- 5. Select the appropriate starting dose to achieve therapeutic response, taking into account the risk of under-treatment.
- 6. Keep the requirement for continuing GC treatment under constant review, and titrate the dose against therapeutic response, risk of under-treatment and development of adverse events such as DM.
- 7. All patients should have appropriate monitoring for clinically significant adverse events, including DM.

The UK Joint British Diabetes Societies for Inpatient Care (JBDS-IP) has put together a set of guideline on the management of hyperglycemia and steroid (glucocorticoid) therapy [343], which again, although non-specific to PMR and/or GCA patients, still applicable to the management of GC-induced DM in these patient populations. Although the guideline includes management for in-patient care, it also includes management for GC-induced DM in general; therefore it is still applicable to my patient population. In addition, capillary blood glucose (CBG) testing is accessible to patients in the community and self-monitoring may be done at home.

The guideline recommends a Hemoglobin A1c (HbA1c) baseline level to be taken prior to the commencement of GC therapy. For patients without a pre-existing diagnosis of DM, the CBG testing should be done at least once daily, preferably prior to lunch or evening meals, or alternatively 1-2 hours post lunch or evening meal. If the initial CBG is less than 12mmol/L, then the once daily testing is adequate. If a subsequent CBG is found to be greater than 12mmol/L, then the frequency of testing should be increased to four times daily before meals and before bed. In cases where the CBG is consistently greater than 12mmol/L (i.e. on two occasions over 24 hours), then patients should be started on treatment.

The guideline also recommended a few treatment options for DM patients who are taking once daily GC. One of the treatment options is to use a sulphonylurea such as gliclazide. Gliclazide is a short acting sulphonylurea, usually taken once daily to promote insulin release from the pancreatic beta cell. It can be titrated to a maximum of 320mg daily, with a maximum of 240mg in the morning. Gliclazide is relatively cheap compared to other anti-diabetic medications, but the risk of hypoglycemia associated with this medication may be problematic for older patients, especially for PMR and/or GCA patients who are mostly above 70 years of age. Therefore, older patients prescribed with gliclazide should be counselled on the common signs and symptoms of hypoglycemia and actions to take if it does occur.

Another oral treatment option is the use of a biguanide such as metformin that works by decreasing glucose production and increasing insulin sensitivity. It can be titrated to a maximum of 1g twice a day. Benefits of metformin include its low cost, favourable weight profile and low risk of developing hypoglycemia, which makes it a safer option to use during GC tapering, especially for patients without a previous history of DM who may be less likely to have their blood glucose tested consistently. Although considered as one of the safest first-line therapy in DM, use of metformin in the elderly should still be monitored. One of the main concerns with the use of metformin is an increased risk of lactic acidosis, especially in the elderly; though the American Diabetes Association in its consensus report has clarified that despite early concerns, the evidence for an increased risk is minimal [344]. It should however, be used with caution in patients with reduced renal function. It should not be used in those with an estimated glomerular filtration rate (eGFR) of less than 30mL/min, and for those with an eGFR between 30-60mL/min, the dose should be reduced [345, 346].

One of the non-oral treatment options is the use of the basal human insulin (i.e. Humulin I, Insuman Basal or Insulatard). Morning administration of these basal insulins may be beneficial as they closely fit the glucose excursion induced by a single dose of oral GC in the morning. The recommended starting dose is 10 units of insulin with a daily dose increase between 10-20%, titrated to the blood glucose level, although dose increments of up to 40% have been shown to be required in some patients [347]. As with all insulins, one of the major concerns is the risk of hypoglycemia. Given the heterogeneity of the older PMR and/or GCA patient population, the risk of hypoglycemia must be carefully considered before using an insulin regimen to achieve an aggressive target for hyperglycemia control [344]. The management and treatment algorithm for patients without a pre-existing diagnosis of DM has been summarized in Figure 7.1, of which the information was adapted from the JBDS-IP – Management of hyperglycemia and steroid (glucocorticoid) therapy guideline [343].



Figure 7.1: Management and treatment algorithm for GC-induced DM in patients without a pre-existing diagnosis of DM, adapted from JBDS-IP guideline [343].

Treatment of DM in PMR and/or GCA patients will always remain a challenge, especially with this group of patients being older in age and more susceptible to the adverse effects of anti-diabetic medications due to age-related changes in the pharmacokinetics and pharmacodynamics affecting drug disposition. This is often aggravated by polypharmacy and complex regimens associated with multiple comorbidities. Shared decision making has been advocated as the most appropriate approach to improving the quality of these important medical decisions [348, 349]. Key aspects of this shared decision include the establishment of an ongoing partnership between the patient and provider, information exchange, deliberation on choices and finally, making the decisions [350]. One key element of improving communication and involving patients in the decision making process is to find congruence between patient goals and clinicians goals [344]. Clinicians should first educate patients on the significance of risk factors and then discuss the possible harms and benefits of each intervention to reduce these risk factors. Equally important is to discuss treatment options and the possible adverse events that accompany each treatment options. The treatment doses should always be balanced against the need to use the lowest effective dose for the shortest period of time to avoid adverse events [6].

7.5. Future Work

One area of interest would be to identify and stratify patients who are at high risk and explore their DM risk associated with GC use. One potential group of patients who are considered to be at high risk is those with relapsed or refractory PMR/GCA. These are the patients with the highest unmet need and the hardest to manage. It is likely that current treatment is sub-optimal in these patients, thus leading to frequent relapse. They are also likely to have a pre-existing high cumulative GC burden with greater concomitant medication usage and greater burden of co-morbidities [351, 281, 352]. For GCA patients, tocilizumab has recently been made available as an alternative drug, but the cost of it is high, and use of this drug is likely to only be in a very small selected patient population based on eligibility [100]. Another group of patients who are at high risk of developing DM are PMR/GCA patients with co-morbidities, especially those with hypoalbuminemia or liver disease. Hypoalbuminemia is quite common at presentation of PMR/GCA as it is a feature of the acute phase response. GCs are metabolized primarily in the liver; therefore the presence of liver disease prolongs the half-life of GCs. This would result in patients having a reduced drug clearance, and possibly an increased risk of GC-related toxicity such as DM. GCs are known for their high protein binding property, with non-linear protein bound attributes. Since only the unbound form is the biologically active compound, only the measurement for the free fraction of GC in the plasma is relevant. Patients with hypoalbuminemia will have a higher free fraction of circulating GC in the plasma; therefore they may be at higher risk of experiencing GC-related toxicities such as DM. It is possible that what is most detrimental in regard to DM is not just the higher dose, but the combination of higher dose with active uncontrolled inflammation.

The reversibility of GC-induced DM in PMR/GCA patients is also an area of potential interest. The effect of GC has been reported in some case reports as being transient and reversible, of which the effect of GC doses on endocrine metabolism is expected to return to baseline as GC doses are reduced over time [20, 353]. This issue is however, not well studied and remains a challenge in observational studies, as it is likely to be confounded by many factors, including but not limited to treatment dose, duration, lifestyle changes, co-morbidities, as well as use of other medications.

The impact of BMI on the risk of type 2 DM is a well-investigated topic, with many studies reporting that higher BMI was associated with a higher risk of developing type 2 DM [354-356], though it is unknown if higher BMI has any impact on the development of steroid-induced DM. BMI was not accounted for in my study for reasons mentioned in Section 5.6.4, but it is definitely important to consider the role of BMI and obesity in the development of steroid-induced DM, especially the timing and duration of obesity, as well as pre-diabetic status. Though prolonged duration of obesity (or higher BMI) has been reported to cause additional metabolic changes, leading to the development of hyperglycemia and diabetes [357, 358], there is still a huge research gap on whether this hypothesis applies to steroid-induced DM.

As discussed in section 7.2.2.1, the issue of how GCs should be tapered in the safest and most effective way with minimal relapses is of great importance. Since relapse rates have been reported to be associated with higher initial GC dose and faster GC tapering rates, one of the possible areas of interest is to compare the risk of DM in patients with a tapering dose regimen versus a fixed dose regimen. This can be done by comparing the tapering dose regimen from my study population with Kirwan's proposed fixed dose regimen - starting dose of 15mg daily for six weeks then 12.5mg daily for six weeks, then a 10mg maintenance dose for one year, with a 1mg/per month taper thereafter if needed [313]. Another approach would be to do a RCT of tapering dose regimen versus a fixed dose regimen. This approach would however, require a significant amount of time for patient recruitment, as well as a significant amount of financial resources to implement the trial. In addition, patients with PMR and/or GCA may develop flares throughout their course of disease, potentially more so for patients on low dose regimens. Therefore an additional challenge would be to incorporate a protocol on how these patients should be managed if flares occur, which would add complexity to the design protocol.

As discussed in section 7.2.4, there is a substantial deviation in the prescribing pattern of oral GC in patients with GCA (with or without PMR) compared to EULAR guidelines. One of the possible next steps is to impute the initial GC doses using a data driven algorithm with clinical input and reevaluate the risk of DM associated with oral GC cumulative dose.

As discussed in the section 7.3.2.3.3, a weekly dose regimen could be explored instead of using the monthly dose regimen. The WCE method could also be applied to quantify the risk of other GC-related toxicities such as osteoporosis, cataract, adrenal suppression etc.

The last issue of interest is related to the current GP prescribing software that does not seem to deal well with gradual GC taper regimens as often used in PMR and/or GCA. Even if recorded, the complete regimen is not available as only the dose would be recorded (i.e. 5 4 3 2 1), but not the duration for each of the specific doses. This might be an area that needs to be improved by the GP software developers.

7.6. Conclusions

Although GCs have been used extensively for decades in the treatment of PMR and GCA, the risk of developing DM has not been quantified in relation to the dose, duration and timing of oral GC cumulative dose exposure. Applications of the rolling cumulative dose and WCE models have provided great clinical insight on the etiologically pertinent time window of past exposure and on the relative importance of GC doses taken at different time periods. My study finding demonstrates that patients with PMR and/or GCA have a substantially increased risk of developing DM within the first few months of treatment initiation. Implementation of a screening program for DM should be considered in these patient groups. This study has also provided insights on the prescribing pattern of oral GC in primary care as compared to guidelines. Understanding the pattern of use in a real-world clinical setting is crucial, as patients with PMR and/or GCA often require long-term treatment with GC. Optimum doses of GC are needed to control disease symptoms and prevent potential relapses, yet the doses should also be as low as possible and prescribed for the shortest duration possible to minimize the occurrence of GC-induced DM. Since treatment with GC is absolutely essential in these patient groups, effective prevention and efficient management of side effects such as DM associated with GC therapy is crucial to reduce morbidity and increase quality of life in these patients.

REFERENCES

- 1. Kermani TA, Warrington KJ: Polymyalgia rheumatica. *The Lancet* 2013, 381:63–72.
- 2. Dasgupta B, Cimmino MA, Maradit-Kremers H, et al: 2012 provisional classification criteria for polymyalgia rheumatica: a European League Against Rheumatism/American College of Rheumatology collaborative initiative. *Ann Rheum Dis* 2012, 71:484–492.
- 3. Fardet L, Petersen I, Nazareth I, et al: Prevalence of long-term oral glucocorticoid prescriptions in the UK over the past 20 years. *Rheumatology* (*Oxford*) 2011, 50:1982-1990.
- 4. Walsh LJ, Wong CA, Pringle M, et al: Use of oral corticosteroids in the community and the prevention of secondary osteoporosis: a cross sectional study. *Br Med J* 1996:344–346.
- 5. Vanhoof J, Declerck K, Geusens P, et al: Prevalence of rheumatic diseases in a rheumatological outpatient practice. *Ann Rheum Dis* 2002:453–455.
- 6. Salvarani C, F Cantini, L Boiardi, et al: Polymyalgia rheumatica and giant cell arteritis. *Lancet* 2008:234-245.
- 7. Dasgupta B, Borg FA, BSR and BHPR Standards Guidelines and Audit Working Group: BSR and BHPR Guidelines for the management of giant cell arteritis. *Rheumatology (Oxford)* 2010, 49:1594-1597.
- 8. Smeeth L, Cook C, Hall AJ, et al: Incidence of diagnosed polymyalgia rheumatica and temporal arteritis in the United Kingdom, 1990–2001. *Ann Rheum Dis* 2006:1093–1098.
- 9. Herrett E, Gallagher AM, Bhaskaran K, et al: Data Resource Profile: Clinical Practice Research Datalink (CPRD). *International Journal of Epidemiology* 2015, 44:827-836.
- 10. Gonzalez-Gay MA, Vazquez-Rodriguez TR, Lopez-Diaz MJ: Epidemiology of giant cell arteritis and polymyalgia rheumatica. *Arthritis Rheum* 2009, 61:1454-1461.
- 11. Lai LYH, Harris E, West RM, et al: Association between glucocorticoid therapy and incidence of diabetes mellitus in polymyalgia rheumatica and giant cell arteritis: a systematic review and meta-analysis. *RMD Open* 2018, 4:e000521.
- 12. Partington RJ, Muller S, Helliwell T, et al: Incidence, prevalence and treatment burden of polymyalgia rheumatica in the UK over two decades: a population-based study. *Ann Rheum Dis* 2018, 77:1750-1756.
- 13. Proven A, Gabriel SK, Orces C: Glucocorticoid therapy in giant cell arteritis: duration and adverse outcomes. *Arthritis Rheum* 2003:703-708.
- 14. Jover JA, Hernández-García C, Morado IC, et al: Combined treatment of giantcell arteritis with methotrexate and prednisone. a randomized, double-blind, placebo-controlled trial *Annals of internal medicine* 2001, 134:106-114.
- 15. Van der Goes MC, Jacobs JW, Boers M, et al: Patient and rheumatologist perspectives on glucocorticoids: an exercise to improve the implementation of the European League Against Rheumatism (EULAR) recommendations on the management of systemic glucocorticoid therapy in rheumatic diseases. *Ann Rheum Dis* 2010, 69:1015-1021.
- 16. Hwang JL, Weiss RE: Steroid-induced diabetes: a clinical and molecular approach to understanding and treatment. *Diabetes Metab Res Rev* 2014, 30:96-102.
- 17. Clore JN, Thurby-Hay L: Glucocorticoid-induced hyperglycemia. *Endocr Pract* 2009, 15:469–474.
- 18. Gurvitz JH, Bohn RL, Glynn RJ, et al: Glucocorticoids and the risk for initiation of hypoglycaemic therapy. *Arch Int Med* 1994, 154:97-101.
- 19. Morris D: Steroid-induced diabetes and hyperglycaemia. Part 1: mechanisms and risks. *Diabetes & Primary Care* 2018, 20:151–153.

- 20. Tamez-Pérez HE, Quintanilla-Flores DL, Rodríguez-Gutiérrez R, et al: Steroid hyperglycemia: Prevalence, early detection and therapeutic recommendations: A narrative review. *World J Diabetes* 2015, 6:1073-1081.
- 21. Dejaco C, Singh YP, Perel P, et al: 2015 Recommendations for the Management of Polymyalgia Rheumatica: A European League Against Rheumatism American College of Rheumatology Collaborative Initiative. *Arthritis and Rheumatology* 2015, 67:2569-2580.
- 22. Hellmich B, Agueda A, Monti S, et al: 2018 update of the EULAR recommendations for the management of large vessel vasculitis. *Ann Rheum Dis* 2019, Published Online First: 03 July 2019.
- 23. Gonzalez-Gay MA: Giant cell arteritis and polymyalgia rheumatica: two different but often overlapping conditions. *Semin Arthritis Rheum* 2004, 33:289-293.
- 24. Dasgupta B, Borg FA, BSR and BHPR Standards Guidelines and Audit Working Group: BSR and BHPR guidelines for the management of polymyalgia rheumatica. *Rheumatology (Oxford)* 2010, 49:186-190.
- 25. Smith CA, Fidler WJ, Pinals RS: The epidemiology of giant cell arteritis. Report of a ten-year study in Shelby County, Tennessee. *Arthritis Rheum* 1983:1214–1219.
- 26. Salvarani C, Gabriel SE, O'Fallon WM, et al: Epidemiology of polymyalgia rheumatica in Olmsted County, Minnesota, 1970–1991. *Arthritis Rheum* 1995, 38:369-373.
- 27. Gonzalez-Gay MA, Barros S, Lopez-Diaz MJ, et al: Giant cell arteritis: disease patterns of clinical presentation in a series of 240 patients. *Medicine* 2005, 84:269-276.
- 28. Dejaco C, Duftner C, Dasgupta B, et al: Polymyalgia rheumatica and giant cell arteritis: management of two diseases of the elderly. *Aging Health* 2011, 7:633–645.
- 29. Dejaco C, Duftner C, Buttgereit F, et al: The spectrum of giant cell arteritis and polymyalgia rheumatica: revisiting the concept of the disease. *Rheumatology* (*Oxford*) 2017, 56:506-515.
- 30. Weyand CM, Fulbright JW, Evans JM, et al: Corticosteroid requirements in polymyalgia rheumatica. *Arch Intern Med* 1999, 159:577-584.
- 31. Ponte C, Rodrigues AF, O'Neill L, et al: Giant cell arteritis: Current treatment and management. *World journal of clinical cases* 2015, 3:484-494.
- 32. Barraclough K, Mallen CD, Helliwell T, et al: Diagnosis and management of giant cell arteritis. *Br J Gen Pract* 2012, 62:329-330.
- 33. Bruce W: Senile rheumatologic gout. *British Medical Journal* 1888, 2:811–813.
- 34. Barber HS: Myalgic syndrome with constitutional effects: polymyalgia rheumatica. *Annals of the Rheumatic Diseases* 1957, 16:230–237.
- 35. Healey LA: The overlap of rheumatoid arthritis and polymyalgia rheumatica in older patients. *Mediguide to Inflammatory Diseases* 1989:8.
- 36. Doran MF, Crowson CS, O'Fallon WM, et al: Trends in the incidence of polymyalgia rheumatica over a 30 year period in Olmsted County, Minnesota, USA. *Journal of Rheumatology* 2002, 29:1694-1697.
- 37. Lawrence RC, Felson DT, Helmick CG, et al: Estimates of the prevalence of arthritis and other rheumatic conditions in the United States. Part II. *Arthritis Rheum* 2008, 58:26-35.
- 38. Boesen P, Sorensen SF: Giant cell arteritis, temporal arteritis, and polymyalgia rheumatica in a Danish county. A prospective investigation, 1982-1985. *Arthritis Care Research* 1987, 30:294-299.
- 39. Gran JT, Myklebust G, Wilsgaard T, et al: Survival in polymyalgia rheumatica and temporal arteritis: a study of 398 cases and matched population controls. *Rheumatology* 2001, 40:1238-1242.
- 40. Gran JT, Myklebust G: The incidence of polymyalgia rheumatica and temporal arteritis in the county of Aust Agder, south Norway: a prospective study 1987-94. *The Journal of rheumatology* 1997, 24:1739-1743.

- 41. Schaufelberger C, Bengtsson BA, R A: Epidemiology and mortality in 220 patients with polymyalgia rheumatica. *Br J Rheumatol* 1995, 34:261-264.
- 42. Elling P, Olsson AT, Elling H: Synchronous variations of the incidence of temporal arteritis and polymyalgia rheumatica in different regions of Denmark; association with epidemics of Mycoplasma pneumoniae infection. *The Journal of rheumatology* 1996, 23:112-119.
- 43. Gonzalez-Gay MA, Garcia-Porrua C, Vazquez-Caruncho M, et al: The spectrum of polymyalgia rheumatica in northwestern Spain: incidence and analysis of variables associated with relapse in a 10 year study. *The Journal of rheumatology* 1999, 26:1326-1332.
- 44. Salvarani C, Macchioni P, Zizzi F, et al: Epidemiologic and immunogenetic aspects of polymyalgia rheumatica and giant cell arteritis in northern Italy *Arthritis Rheum* 1991, 34:351–356.
- 45. Bird HA, Esselinckx W, Dixon AS, et al: An evaluation of criteria for polymyalgia rheumatica. *Annals of the Rheumatic Diseases* 1979, 38:434–439.
- 46. Jones JG, Hazleman BL: Prognosis and management of polymyalgia rheumatica. *Ann Rheum Dis* 1981, 40:1-5.
- 47. Chuang TY, Hunder GG, Ilstrup DM, et al: Polymyalgia rheumatica: a 10-year epidemiologic and clinical study. *Annals of internal medicine* 1982, 97:672-680.
- 48. Healey LA: Long-term follow-up of polymyalgia rheumatica: evidence for synovitis. *Seminars in Arthritis and Rheumatism* 1984, 13:322–328.
- 49. Aggarwal R, Ringold S, Khanna D, et al: Distinctions between diagnostic and classification criteria? *Arthritis Care Res* 2015, 67:891-897.
- 50. Salvarani C, Cantini F, Boiardi L, et al: Polymyalgia rheumatica and giant-cell arteritis. *N Engl J Med* 2002:261–271.
- 51. Caylor TL, Perkins A: Recognition and management of polymyalgia rheumatica and giant cell arteritis. *American Family Physician* 2013, 88:676-684.
- 52. Mackie SL: Polymyalgia rheumatica: pathogenesis and management. *Clinical Medicine August* 2013, 13:398-400.
- 53. Weyand CM, Goronzy JJ: Giant-cell arteritis and polymyalgia rheumatica. *Annals of internal medicine* 2003, 139:505-515.
- 54. Dasgupta B, Matteson EL, Maradit-Kremers H: Management guidelines and outcome measures in polymyalgia rheumatica (PMR). *Clinical and experimental rheumatology* 2007, 25:130–136.
- 55. Kyle V, Hazleman BL: Treatment of polymyalgia rheumatica and giant cell arteritis. II. Relation between steroid dose and steroid associated side effects. *Annals of the Rheumatic Diseases* 1989, 48:662-666.
- 56. Horton B, Magath T, Brown G: An undescribed form of arteritis of the temporal vessels. *Proc Mayo Clin* 1932:700–701.
- 57. Borchers A, Gershwin M: Giant cell arteritis: A review of classification, pathophysiology, geoepidemiology and treatment. *Autoimmunity Reviews* 2012:A544-A554.
- 58. Jennette JC, Falk RJ, Andrassy K, et al: Nomenclature of systemic vasculitides. Proposal of an international consensus conference. *Arthritis Rheum* 1994:187–192.
- 59. Warrington KJ, Matteson EL: Management guidelines and outcome measures in giant cell arteritis (GCA). *Clinical and experimental rheumatology* 2007, 25:137-141.
- 60. Gonzalez-Gay MA, Martinez-Dubois C, Agudo M, et al: Giant cell arteritis: epidemiology, diagnosis, and management. *Curr Rheumatol Rep* 2010, 12:436–442.
- 61. Crowson CS, Matteson EL, Myasoedova E, et al: The lifetime risk of adult-onset rheumatoid arthritis and other inflammatory autoimmune rheumatic diseases. *Arthritis Rheum* 2011, 63:633-639.
- 62. Crowson CS, Matteson EL: Contemporary prevalence estimates for giant cell arteritis and polymyalgia rheumatica, 2015. *Semin Arthritis Rheum* 2017, 47:253-256.

- 63. Chandran AK, Udayakumar PD, Crowson CS, et al: The incidence of giant cell arteritis in Olmsted County, Minnesota, over a 60-year period 1950-2009. *Scand J Rheumatol* 2015, 44:215-218.
- 64. Abdul-Rahman AM, Molteno AC, Bevin TH: The epidemiology of giant cell arteritis in Otago, New Zealand: a 9-year analysis. *N Z Med J* 2011, 124:44–52.
- 65. Lee JL, Naguwa SM, Cheema GS, et al: The geo-epidemiology of temporal (giant cell) arteritis. *Clin Rev Allergy Immunol* 2008:88–95.
- 66. Franzén P, Sutinen S, von Knorring J, et al: Giant cell arteritis and polymyalgia rheumatica in a region of Finland: an epidemiologic, clinical and pathologic study, 1984-1988. *The Journal of rheumatology* 1992, 19:273-276.
- 67. Haugeberg G, Irgens KA, Thomsen RS: No major differences in incidence of temporal arteritis in northern and western Norway compared with reports from southern Norway. *Scand J Rheumatol* 2003, 32:318–319.
- 68. Baldursson O, Steinsson K, Bjornsson J, et al: Giant cell arteritis in Iceland. An epidemiologic and histopathologic analysis. *Arthritis Rheum* 1994:1007–1012.
- 69. Nordborg C, Johansson H, Petursdottir V, et al: The epidemiology of biopsypositive giant cell arteritis: special reference to changes in the age of the population. *Rheumatology (Oxford)* 2003, 42:549-552.
- 70. Gonzalez-Gay MA, Miranda-Filloy JA, Lopez-Diaz MJ, et al: Giant cell arteritis in northwestern Spain: a 25-year epidemiologic study. *Medicine (Baltimore)* 2007:61–68.
- 71. Barrier J, Pion P, Massari R, et al: [Epidemiologic approach to Horton's disease in the department of Loire-Atlantique. 110 cases in 10 years (1970–1979)]. [Article in French]. La Revue de medecine interne / fondee par la Societe nationale francaise de medecine interne 1982, 3:13–20.
- 72. Chaudhry IA, Shamsi FA, Elzaridi E, et al: Epidemiology of giant-cell arteritis in an Arab population: a 22-year study. *The British journal of ophthalmology* 2007:715–718.
- 73. Kobayashi S, Yano T, Matsumoto Y, et al: Clinical and epidemiologic analysis of giant cell (temporal) arteritis from a nationwide survey in 1998 in Japan: the first government-supported nationwide survey. *Arthritis Rheum* 2003, 49:594–598.
- 74. Reinhold-Keller E, Zeidler A, Gutfleisch J, et al: Giant cell arteritis is more prevalent in urban than in rural populations: results of an epidemiological study of primary systemic vasculitides in Germany. *Rheumatology (Oxford)* 2000:1396–1402.
- 75. Salvarani C, Crowson CS, O'Fallon WM, et al: Reappraisal of the epidemiology of giant cell arteritis in Olmsted County, Minnesota, over a fifty year period. *Arthritis Rheum* 2004:264–268.
- 76. Bas-Lando M, Breuer GS, Berkun Y, et al: The incidence of giant cell arteritis in Jerusalem over a 25-year period: annual and seasonal fluctuations. *Clinical and experimental rheumatology* 2007:S15–17.
- 77. Petri H, Nevitt A, Sarsour K, et al: Incidence of Giant Cell Arteritis and Characteristics of Patients: Data-Driven Analysis of Comorbidities. *Arthritis Care Res* 2015, 67:390-395.
- 78. Ramstead CL, Patel AD: Giant cell arteritis in a neuro-ophthalmology clinic in Saskatoon, 1998-2003. *Can J Ophthalmol* 2007, 42:295-298.
- 79. Dunstan E, Lester SL, Rischmueller M, et al: Epidemiology of biopsy-proven giant cell arteritis in South Australia. *Intern Med J* 2014, 44:32-39.
- 80. Pamuk ON, Dönmez S, Karahan B, et al: Giant cell arteritis and polymyalgia rheumatica in northwestern Turkey: clinical features and epidemiological data. *Clinical and experimental rheumatology* 2009:830–833.
- 81. Vaith P, Warnatz K: Clinical and serological findings of giant-cell arteritis. *The Journal of rheumatology* 2009:124–131.
- 82. Ness T, Auw-Hadrich C, Schmidt D: Temporal arteritis (giant cell arteritis). Clinical picture, histology, and treatment. *Ophthalmologe* 2006:296–301.

- 83. Kale N, Eggenberger E: Diagnosis and management of giant cell arteritis: a review. *Curr Opin Ophthalmol* 2010:417–422.
- 84. Rahman W, Rahman FZ: Giant cell (temporal) arteritis: an overview and update. *Surv Ophthalmol* 2005:415–428.
- 85. Mackie SL, Dejaco C, Appenzeller S, et al: BSR guideline on diagnosis and treatment of giant cell arteritis: full guideline to be published electronically (draft for public consultation) (*Link:* <u>https://wwwrheumatologyorguk/Portals/0/Documents/Guidelines/GCA/Open_Consultation_Full_GCA_Guidelinepdf?ver=2019-07-02-152636-237)(Accessed on 26th November, 2019).</u>
- 86. Ness T, Bley TA, Schmidt WA, et al: The diagnosis and treatment of giant cell arteritis. *Dtsch Arztebl Int* 2013, 110:376–386.
- 87. Généreau T, Lortholary O, Pottier MA, et al: Temporal artery biopsy: a diagnostic tool for systemic necrotizing vasculitis. French Vasculitis Study Group. *Arthritis Rheum* 1999, 42:2674-2681.
- 88. Rubenstein E, Maldini C, Gonzalez-Chiappe S, et al: Sensitivity of temporal artery biopsy in giant cell arteritis: systematic literature review and meta-analysis of clinical data. *Rheumatology* 2019, 58:kez058.051.
- 89. Walvick MD, Walvick MP: Giant cell arteritis: laboratory predictors of a positive temporal artery biopsy. *Ophthalmology* 2011:1201–1204.
- 90. Hunder GG, Bloch DA, Michel BA: The American College of Rheumatology 1990 criteria for the classification of giant cell (temporal) arteritis. *Arthritis Rheum* 1990:1122–1128.
- 91. Rao JK, Allen NB, Pincus T: Limitations of the 1990 American College of Rheumatology classification criteria in the diagnosis of vasculitis. *Annals of internal medicine* 1998, 129:345-352.
- 92. Seeliger B, Sznajd J, Robson JC, et al: Are the 1990 American College of Rheumatology vasculitis classification criteria still valid? *Rheumatology (Oxford)* 2017, 56:1154-1161.
- 93. Conway R: Giant cell arteritis: diagnostic tools, treatment targets, and pathogenic pathways. *PhD dissertation, University College Dublin, School of Medicine* 2017.
- 94. Hayreh SS, Podhajsky PA, Zimmerman B: Ocular manifestations of giant cell arteritis. *Am J Ophthalmol* 1998, 125:509–520.
- 95. Docken W: Treatment of giant cell arteritis. *UpToDate* 2018 (<u>https://wwwuptodatecom/contents/treatment-of-giant-cell-arteritis</u>) (Accessed on 17th June, 2019).
- 96. Harris E, Tiganescu A, Tubeuf S, et al: The prediction and monitoring of toxicity associated with long-term systemic glucocorticoid therapy. *Current Rheumatology Reports* 2015, 17:513.
- 97. Strehl C, Johannes WJ, Bijlsma MDT, et al: Defining conditions where longterm glucocorticoid treatment has an acceptably low level of harm to facilitate implementation of existing recommendations: viewpoints from an EULAR task force. *Ann Rheum Dis* 2016:1-6.
- 98. Villiger PM, Adler S, Kuchen S, et al: Tocilizumab for induction and maintenance of remission in giant cell arteritis: a phase 2, randomised, double-blind, placebo-controlled trial. *Lancet* 2016, 387:1921-1927.
- 99. Stone JH, Klearman M, Collinson N, et al: Trial of tocilizumab in giant-cell arteritis. *N Engl J Med* 2017, 377:1494-1495.
- 100. NICE appraisal committee members: NICE Technology appraisal guidance on tocilizumab for treating giant cell arteritis. 2018.
- 101. Conway R, O'Neill L, O'Flynn E, al e: Ustekinumab for refractory giant cell arteritis: A prospective 52-week trial. *Semin Arthritis Rheum* 2018, 48:523-528.
- 102. Centre Hospitalier Universitaire Dijon: Ustekinumab for the Treatment of Relapse of Refractory Giant Cell Arteritis (ULTRA) ClinicalTrials.gov Identifier: NCT03711448. *ClinicalTrialsgov* 2019.

- 103. Hoffman GS, Cid MC, Rendt-Zagar KE, et al: Infliximab for maintenance of glucocorticosteroid-induced remission of giant cell arteritis: a randomized trial. *Annals of internal medicine* 2007:621–630.
- 104. Seror R, Baron G, Hachulla E, et al: Adalimumab for steroid sparing in patients with giant-cell arteritis: results of a multicentre randomised controlled trial. *Ann Rheum Dis* 2014, 73:2074-2081.
- 105. Langford CA, Cuthbertson D, Ytterberg SR, et al: A randomized, double-blind trial of abatacept (CTLA-4lg) for the treatment of giant cell arteritis. *Arthritis & rheumatology (Hoboken, NJ)* 2017, 69:837-845.
- 106. Spiera RF, Mitnick HJ, Kupersmith M, et al: A prospective, double-blind, randomized, placebo controlled trial of methotrexate in the treatment of giant cell arteritis (GCA). *Clinical and experimental rheumatology* 2001, 19:495-501.
- 107. Hoffman GS, Cid MC, Hellmann DB, et al: A multicenter, randomized, doubleblind, placebo-controlled trial of adjuvant methotrexate treatment for giant cell arteritis. *Arthritis Rheum* 2002, 46:1309-1318.
- 108. Mahr AD, Jover JA, Spiera RF, et al: Adjunctive methotrexate for treatment of giant cell arteritis: an individual patient data meta analysis. *Arthritis Rheum* 2007:2789–2797.
- 109. Mollan SP, Sharrack N, Burdon MA, et al: Aspirin as adjunctive treatment for giant cell arteritis. Cochrane Database of Systematic Reviews. 2014.
- 110. De Silva M, Hazleman BL: Azathioprine in giant cell arteritis/polymyalgia rheumatica: a double-blind study. *Ann Rheum Dis* 1986:136–138.
- 111. Quartuccio L, Maset M, De Maglio G, et al: Role of oral cyclophosphamide in the treatment of giant cell arteritis. *Rheumatology (Oxford)* 2012, 51:1677-1686.
- 112. Henes JC, Mueller M, Pfannenberg C, et al: Cyclophosphamide for large vessel vasculitis: assessment of response by PET/CT. *Clinical and experimental rheumatology* 2011, 29:S43-48.
- 113. de Boysson H, Boutemy J, Creveuil C, et al: Is there a place for cyclophosphamide in the treatment of giant-cell arteritis? A case series and systematic review. *Semin Arthritis Rheum* 2013, 43:105-112.
- 114. Schaufelberger C, Andersson R, Nordborg E: No additive effect of cyclosporin A compared with glucocorticoid treatment alone in giant cell arteritis: results of an open, controlled, randomized study. *Br J Rheumatol* 1998, 37:464-465.
- 115. Schaufelberger C, Möllby H, Uddhammar A, et al: No additional steroid-sparing effect of cyclosporine A in giant cell arteritis. *Scand J Rheumatol* 2006, 35:327-329.
- 116. Diamantopoulos AP, Hetland H, Myklebust G: Leflunomide as a corticosteroidsparing agent in giant cell arteritis and polymyalgia rheumatica: a case series. *Biomed Res Int* 2013:120638.
- 117. Sciascia S, Piras D, Baldovino S, et al: Mycophenolate mofetil as steroidsparing treatment for elderly patients with giant cell arteritis: report of three cases. *Aging Clin Exp Res* 2012, 24:273-277.
- 118. Ly KH, Dalmay F, Gondran G, et al: Steroid-sparing effect and toxicity of dapsone treatment in giant cell arteritis: A single-center, retrospective study of 70 patients. *Medicine (Baltimore)* 2016, 95:e4974.
- 119. Kendall EC: Nobel lecture: The development of cortisone as a therapeutic agent. 1950:Nobelprize.org.
- 120. Benedek TG: History of the development of corticosteroid therapy. *Clinical and experimental rheumatology* 2011, 29:S-5-12.
- 121. Fuller PJ, Young MJ: Mechanisms of mineralocorticoid action. *Hypertension* 2005, 46:1227-1235.
- 122. Gilbert K: 7. Polly Wotsit and the giant dragon: the patient's perspective on PMR and GCA. *Rheumatology (Oxford)* 2014, 53:i4–i5.
- 123. Velden VHvd: Glucocorticoids: mechanisms of action and anti-inflammatory potential in asthma. *Mediators Inflamm* 1998, 7:229–237.

- 124. Rohatagi S, Hochhaus G, Mollmann H, et al: Pharmacokinetic interaction between endogenous cortisol and exogenous corticosteroids. *Pharmazie* 1995, 50:610-613.
- 125. Rosman PM, Benn R, Kay M, et al: Cortisol binding in uremic plasma. I: absence of abnormal cortisol binding to cortisol binding to corticosteroid-binding globulin. *Nephron* 1984, 37:160-165.
- 126. Rosman PM, Benn R, Kay M, et al: Cortisol binding in uremic plasma. II: decreased cortisol binding to albumin. *Nephron* 1984, 37:160-165.
- 127. Frey BM, Seeberger M, Frey FJ: Pharmacokinetics of 3 prednisolone prodrugs. Evidence of therapeutic inequivalence in renal transplant patients with rejection. *Transplantation* 1985, 39:270-274.
- 128. Ogawa K, Sueda K, Matsui N: The effect of cortisol, progesterone, and transcortin on phytohemagglutinin-stimulated human blood mononuclear cells and their interplay. *The Journal of clinical endocrinology and metabolism* 1983, 56:121-126.
- 129. Slaunwhite WR, Lockie GN, Back N, Sandberg AA: Inactivity in vivo of transcortin-bound cortisol. *Science* 1962, 135:1062-1063.
- 130. Bergrem H, Grøttum P, Rugstad HE: Pharmacokinetics and protein binding of prednisolone after oral and intravenous administration. *Eur J Clin Pharmacol* 1983, 24:415-419.
- 131. Tanner A, Bochner F, Caffin J, et al: Dose-dependent prednisolone kinetics. *Clin Pharmacol Ther* 1979, 25:571-578.
- 132. Wald JA, Law RM, Ludwig EA, et al: Evaluation of dose related pharmacokinetics and pharmacodynamics of prednisolone in man. *J Pharmacokinet Biopharm* 1992, 20:567-589.
- 133. Coutinho AE, Chapman KE: The anti-inflammatory and immunosuppressive effects of glucocorticoids, recent developments and mechanistic insights. *Mol Cell Endocrinol* 2011, 335:2-13.
- 134. Perretti M, Ahluwalia A: The microcirculation and inflammation: site of action for glucocorticoids. *Microcirculation* 2000, 7:147-161.
- 135. Buttgereit F, Wehling M, Burmester GR: A new hypothesis of modular glucocorticoid actions. Glucocorticoid treatment of rheumatic diseases revisited. *Arthritis Rheum* 1998, 41:761-767.
- 136. Buttgereit F, Burmester GR, Brand MD: Bioenergetics of immune functions: the power that drives immunity. *Immunology Today* 2000, 21:192-199.
- 137. Buttgereit F, Scheffold A: Rapid glucocorticoid effects on immune cells. *Steroids* 2002, 67:529-534.
- 138. Almawi WY, Melemedjian OK: Molecular mechanisms of glucocorticoid antiproliferative effects: antagonism of transcription factor activity by glucocorticoid receptor. *J Leukoc Biol* 2002, 71:9-15.
- 139. Ishaq M, DeGray G, Mou K, et al: Zap70 signaling pathway mediates glucocorticoid receptor-dependent transcriptional activation: role in the regulation of annexin 1 expression in T cells. *J Immunol* 2007, 179:3851–3858.
- 140. Stahn C, Buttgereit F: Genomic and nongenomic effects of glucocorticoids. *Nat Clin Pract Rheumatol* 2008, 10:525-533.
- 141. In-Ho Song, Ralf Gold, Rainer H Straub, et al: New glucocorticoids on the horizon: repress, don't activate. *The Journal of rheumatology* 2005, 32:1199–1207.
- 142. RM Nissen, KR Yamamoto: The glucocorticoid receptor inhibits NFκB by interfering with serine-2 phosphorylation of the RNA polymerase II carboxy-terminal domain. *Genes Dev* 2000, 14:2314–2329.
- 143. Schäcke H, Rehwinkel H, Asadullah K, et al: Insight into the molecular mechanisms of glucocorticoid receptor action promotes identification of novel ligands with an improved therapeutic index. *Exp Dermatol* 2006, 15:565-573.
- 144. Buttgereit F, da Silva JA, Boers M, et al: Standardised nomenclature for glucocorticoid dosages and glucocorticoid treatment regimens: current

questions and tentative answers in rheumatology. *Ann Rheum Dis* 2002, 61:718-722.

- 145. Falkenstein E, Norman AW, Wehling M: Mannheim classification of nongenomically initiated (rapid) steroid action(s). *The Journal of clinical endocrinology and metabolism* 2000, 85:2072-2075.
- 146. Croxtall JD, Choudhury Q, Flower RJ: Glucocorticoids act within minutes to inhibit recruitment of signalling factors to activated EGF receptors through a receptor-dependent, transcription-independent mechanism. *Br J Pharmacol* 2000, 130:289-298.
- 147. Panettieri RA, Schaafsma D, Amrani Y, et al: Non-genomic effects of glucocorticoids: An updated view. *Trends Pharmacol Sci* 2019, 40:38-49.
- 148. Schacke H, Schottelius A, Docke WD, et al: Dissociation of transactivation from transrepression by a selective glucocorticoid receptor agonist leads to separation of therapeutic effects from side effects. *Proc Natl Acad Sci* 2004, 101:227-232.
- 149. Lipworth BJ: Therapeutic implications of non-genomic glucocorticoid activity. *Lancet* 2000, 356:87–89.
- 150. Koster MJ, Warrington KJ: Giant cell arteritis: pathogenic mechanisms and new potential therapeutic targets. *BMC Rheumatol* 2017, 1:2.
- 151. Dimitrijevic I, Andersson C, Rissler P, et al: Increased tissue endothelin-1 and endothelin-B receptor expression in temporal arteries from patients with giant cell arteritis. *Ophthalmology* 2010, 117:628–636.
- 152. Pirro M, Bocci EB, Di Filippo F, et al: Imbalance between endothelial injury and repair in patients with polymyalgia rheumatica: improvement with corticosteroid treatment. *J Intern Med* 2012, 272:177-184.
- 153. Dasgupta B, Panayi GS, et al: Interleukin-6 in serum of patients with polymyalgia rheumatica and giant cell arteritis. *Br J Rheumatol* 1990, 29:456–458.
- 154. Hernández-Rodríguez J, Segarra M, Vilardell C, et al: Tissue production of proinflammatory cytokines (IL-1beta, TNFalpha and IL-6) correlates with the intensity of the systemic inflammatory response and with corticosteroid requirements in giant-cell arteritis. *Rheumatology (Oxford)* 2004, 43:294–301.
- 155. Weyand CM, Goronzy JJ: Immune mechanisms in medium and large-vessel vasculitis. *Nat Rev Rheumatol* 2013, 9:731–740.
- 156. Weyand CM, Liao YJ, Goronzy JJ: The immunopathology of giant cell arteritis: diagnostic and therapeutic implications. *32* 2012, 3:259–265.
- 157. Deng J, Younge BR, Olshen RA, et al: Th17 and Th1 T-cell responses in giant cell arteritis. *Circulation* 2010, 121:906–915.
- 158. Samson M, Audia S, Fraszczak J, et al: Th1 and Th17 lymphocytes expressing CD161 are implicated in giant cell arteritis and polymyalgia rheumatica pathogenesis. *Arthritis Rheum* 2012, 64:3788-3798.
- 159. Guggino G, Ferrante A, Macaluso F, et al: Pathogenesis of polymyalgia rheumatica. *Reumatismo* 2018, 70:10-17.
- 160. Uddhammar A, Sundqvist KG, Ellis B, et al: Cytokines and adhesion molecules in patients with polymyalgia rheumatica. *Br J Rheumatol* 1998, 37:766-769.
- 161. Alvarez-Rodriguez L, Lopez-Hoyos M, Mata C, et al: Circulating cytokines in active polymyalgia rheumatica. *Annals Rheum Dis* 2010, 69:263-269.
- 162. van der Geest KS, Abdulahad WH, Rutgers A, et al: Serum markers associated with disease activity in giant cell arteritis and polymyalgia rheumatica. *Rheumatology* 2015, 54:1397-1402.
- 163. Roche NE, Fulbright JW, Wagner AD, et al: Correlation of interleukin-6 production and disease activity in polymyalgia rheumatica and giant cell arteritis. *Arthritis Rheum* 1993, 36:1286-1294.
- 164. Pulsatelli L, Boiardi L, Pignotti E, et al: Serum interleukin-6 receptor in polymyalgia rheumatica: a potential marker of relapse/recurrence risk. *Arthritis Rheum* 2008, 59:1147-1154.

- 165. van der Geest KS, Abdulahad WH, Chalan P, et al: Disturbed B cell homeostasis in newly diagnosed giant cell arteritis and polymyalgia rheumatica. *Arthritis & rheumatology (Hoboken, NJ)* 2014, 66:1927-1938.
- 166. Buttgereit F, Straub RH, Wehling M, et al: Glucocorticoids in the treatment of rheumatic diseases: an update on the mechanisms of action. *Arthritis Rheum* 2004, 50:3408-3417.
- 167. Tyrrell JB: Glucocorticoid therapy. In: Felig P, Baxter JD, Frohman LA, eds. *Endocrinology and metabolism 3rd ed New York: McGraw-Hill* 1995:855–882.
- 168. Da Silva JAP, Bijlsma JWJ: Optimizing glucocorticoid therapy in rheumatoid arthritis. *Rheum Dis Clin N Am* 2000, 26:859–880.
- 169. Klippel JH: Systemic lupus erythematosus. *Management In: Klippel JH, Dieppe PA, eds Rheumatology 2nd ed London: Mosby* 1998, 7:7.1–7.8.
- 170. Da Silva JA, Jacobs JW, Kirwan JR, et al: Safety of low dose glucocorticoid treatment in rheumatoid arthritis: published evidence and prospective trial data. *Ann Rheum Dis* 2006, 65:285-293.
- 171. van der Goes MC, Jacobs JW, Bijlsma JW, et al: The value of glucocorticoid cotherapy in different rheumatic diseases - positive and adverse effects. *Arthritis Res Ther* 2014, 16:S2.
- 172. Navarette MG, Brey RL: Neuropsychiatric systemic lupus erythematosus. *Curr Treat Options Neurol* 2000, 2:473–485.
- 173. Langford CA: Management of systemic vasculitis. *Best Pract Res Clin Rheumatology* 2001, 15:281–297.
- 174. Boumpas DT, Austin HA III, Fessler BJ, et al: Systemic lupus erythematosus: emerging concepts. Part I: renal, neuropsychiatric, cardiovascular, pulmonary, and hematologic disease. *Annals of internal medicine* 1995, 112:940–950.
- 175. Jacobs JW, Geenen R, Evers AW, et al: Short term effects of corticosteroid pulse treatment on disease activity and the wellbeing of patients with active rheumatoid arthritis. *Ann Rheum Dis* 2001, 60:61-64.
- 176. Gabriel SE, Sunku J, Salvarani C, et al: Adverse outcomes of anti-inflammatory therapy among patients with polymyalgia rheumatica. *Arthritis Rheum* 1997, 40:1873-1878.
- 177. van Staa TP, Leufkens HG, Abenhaim L, et al: Use of oral corticosteroids in the United Kingdom. *QJM* 2000, 93:105-111.
- 178. Manson SC, Brown RE, Cerulli A, et al: The cumulative burden of oral corticosteroid side effects and the economic implications of steroid use. *Respir Med* 2009, 103:975-994.
- 179. American Diabetes Association: Classification and diagnosis of diabetes. *Diabetes Care* 2018, 41:S13-S27.
- 180. Liu XX, Zhu XM, Miao Q, et al: Hyperglycemia induced by glucocorticoids in nondiabetic patients: a meta-analysis. *Ann Nutr Metab* 2014, 65:324-332.
- 181. Gulliford MC, Charlton J, Latinovic R: Risk of diabetes associated with prescribed glucocorticoids in a large population. *Diabetes Care* 2006, 29:2728–2729.
- 182. Blackburn D, Hux J, Mamdani M: Quantification of the risk of corticosteroidinduced diabetes mellitus among the elderly. *J Gen Intern Med* 2002, 17:717-720.
- 183. Bonaventura A, Montecucco F: Steroid-induced hyperglycemia: An underdiagnosed problem or clinical inertia? A narrative review. *Diabetes Res Clin Pract* 2018, 139:203-220.
- 184. Nuttall FQ, Ngo A, Gannon MC: Regulation of hepatic glucose production and the role of gluconeogenesis in humans: is the rate of gluconeogenesis constant? *Diabetes Metab Res Rev* 2008, 24:438-458.
- 185. Barthel A, Schmoll D: Novel concepts in insulin regulation of hepatic gluconeogenesis. *Am J Physiol Endocrinol Metab* 2003, 285:E685–692.
- 186. Vander Kooi BT, Onuma H, Oeser JK, et al: The glucose-6-phosphatase catalytic subunit gene promoter contains both positive and negative glucocorticoid response elements. *Mol Endocrinol* 2002, 19:3001–3022.

- 187. Vegiopoulos A, Herzig S: Glucocorticoids, metabolism and metabolic diseases. *Mol Cell Endocrinol* 2007, 275:43–61.
- Lofberg E, Gutierrez A, Wernerman J, et al: Effects of high doses of glucocorticoids on free amino acids, ribosomes and protein turnover in human muscle. *Eur J Clin Invest* 2002, 32:345–353.
- 189. Schakman O, Kalista S, Barbe C, et al: Glucocorticoid-induced skeletal muscle atrophy. *Int J Biochem Cell Biol* 2013, 45:2163–2172.
- 190. Kuo T, McQueen A, Chen TC, et al: Regulation of glucose homeostasis by glucocorticoids. *Adv Exp Med Biol* 2015, 872:99–126.
- 191. van Raalte DH, Diamant M, Ouwens DM, et al: Glucocorticoid treatment impairs microvascular function in healthy men in association with its adverse effects on glucose metabolism and blood pressure: a randomised controlled trial. *Diabetologia* 2013, 56:2383–2391.
- 192. Expert Committee on the Diagnosis and Classification of Diabetes Mellitus: Report of the expert committee on the diagnosis and classification of diabetes mellitus. *Diabetes Care* 2003, 26 (Suppl 1) S5-S20.
- 193. Holst JJ: The physiology of glucagon-like peptide 1. *Physiol Rev* 2007, 87:1409-1439.
- 194. Dube PE, Brubaker PL: Nutrient, neural and endocrine control of glucagon-like peptide secretion. *Horm Metab Res* 2004, 36:755–760.
- 195. Holst JJ, Deacon CF, Vilsboll T, et al: Glucagon-like peptide-1, glucose homeostasis and diabetes. *Trends Mol Med* 2008, 14:161–168.
- 196. Nauck M, Stöckmann F, Ebert R, et al: Reduced incretin effect in Type 2 (noninsulin-dependent) diabetes. *Diabetologia* 1986, 29:46-52.
- 197. Perley MJ, Kipnis DM: Plasma insulin responses to oral and intravenous glucose: studies in normal and diabetic subjects. *J Clin Invest* 1967, 46:1954–1962.
- 198. Muscelli E, Mari A, Natali A, et al: Impact of incretin hormones on beta-cell function in subjects with normal or impaired glucose tolerance. *Am J Physiol Endocrinol Metab* 2006, 291:E1144–1150.
- 199. Muscelli E, Mari A, Casolaro A, et al: Separate impact of obesity and glucose tolerance on the incretin effect in normal subjects and type 2 diabetic patients. *Diabetes* 2008, 57:1340–1348.
- 200. Jensen DH, Aaboe K, Henriksen JE, et al: Steroid-induced insulin resistance and impaired glucose tolerance are both associated with a progressive decline of incretin effect in first-degree relatives of patients with type 2 diabetes. *Diabetologia* 2012, 55:1406–1416.
- 201. Boden G, Shulman GI: Free fatty acids in obesity and type 2 diabetes: Defining their role in the development of insulin resistance and beta-cell dysfunction. *Eur J Clin Invest* 2002, 32:14-23.
- 202. Delaunay F, Khan A, Cintra A, et al: Pancreatic beta cells are important targets for the diabetogenic effects of glucocorticoids. *J Clin Invest* 1997, 100:2094–2098.
- 203. Jeong IK, Oh SH, Kim BJ, et al: The effects of dexamethasone on insulin release and biosynthesis are dependent on the dose and duration of treatment. *Diabetes Res Clin Pract* 2001, 51:163-171.
- 204. Larsson H, Ahren B: Insulin resistant subjects lack islet adaptation to short-term dexamethasone-induced reduction in insulin sensitivity. *Diabetologia* 1999, 42:936–943.
- 205. Linssen MM, van Raalte DH, Toonen EJ, et al: Prednisolone-induced beta cell dysfunction is associated with impaired endoplasmic reticulum homeostasis in INS-1E cells. *Cell Signal* 2011, 23:1708–1715.
- 206. Ranta F, Avram D, Berchtold S, et al: Dexamethasone induces cell death in insulin secreting cells, an effect reversed by exendin-4. *Diabetes* 2006, 55:1380–1390.

- 207. van Raalte DH, Kwa KA, van Genugten RE, et al: Islet-cell dysfunction induced by glucocorticoid treatment: potential role for altered sympathovagal balance? *Metab Clin Exp* 2013, 62:568–577.
- 208. Stumvoll M, Goldstein BJ, van Haeften TW: Type 2 diabetes: Principles of pathogenesis and therapy. *Lancet* 2005, 365:1333–1346.
- 209. Taskinen MR, Nikkila EA, Pelkonen R, et al: Plasma lipoproteins, lipolytic enzymes, and very low density lipoprotein triglyceride turnover in Cushing's syndrome. *The Journal of clinical endocrinology and metabolism* 1983, 57:619–626.
- 210. Burt MG, Willenberg VM, Petersons CJ, et al: Screening for diabetes in patients with inflammatory rheumatological disease administered long-term prednisolone: a cross-sectional study. *Rheumatology (Oxford)* 2012, 51:1112-1119.
- 211. D den Uyl, DH van Raalte, Nurmohamed MT, et al: Metabolic effects of highdose prednisolone treatment in early rheumatoid arthritis: balance between diabetogenic effects and inflammation reduction. *Arthritis Rheum* 2012, 64:639-646.
- 212. Fong AC, Cheung NW: The high incidence of steroid-induced hyperglycaemia in hospital. *Diabetes Res Clin Pract* 2013 99:277-280.
- 213. Petersons CJ, Mangelsdorf BL, Jenkins AB, et al: Effects of low-dose prednisolone on hepatic and peripheral insulin sensitivity, insulin secretion, and abdominal adiposity in patients with inflammatory rheumatologic disease. *Diabetes Care* 2013, 36:2822-2829.
- 214. Raul Ariza-Andraca C, Barile-Fabris LA, Frati-Munari AC, et al: Risk factors for steroid diabetes in rheumatic patients. *Arch Med Res* 1998, 29:259–262.
- 215. Rostom S, Mengat M, Lahlou R, et al: Metabolic syndrome in rheumatoid arthritis: case control study. *BMC Musculoskelet Disord* 2013, 26:147.
- 216. Su CC, Chen IeC, Young FN, et al: Risk of diabetes in patients with rheumatoid arthritis: a 12-year retrospective cohort study. *The Journal of rheumatology* 2013, 40:1513–1518.
- 217. Zeng YJ, Zeng FQ, Dai L, et al: Characteristics and risk factors for hyperglycemia in Chinese female patients with systemic lupus erythematosus. *Lupus* 2010, 19:1344-1350.
- 218. Wallace BC, Small K, Brodley CE, et al: Deploying an interactive machine learning system in an evidence-based practice center: abstrackr. *In Proceedings of the ACM International Health Informatics Symposium (IHI)* 2012:819-824.
- 219. Stroup DF, Berlin JA, Morton SC, et al: Meta-analysis of observational studies in epidemiology: a proposal for reporting. Meta-analysis Of Observational Studies in Epidemiology (MOOSE) group. *JAMA* 2000, 283:2008–2012.
- 220. Higgins JPT, Altman DG, Gøtzsche PC, et al: The Cochrane Collaboration's tool for assessing risk of bias in randomised trials. *BMJ* 2011, 343:d5928.
- 221. Higgins JPT, Thomas J, Chandler J, et al: Cochrane handbook for systematic reviews of interventions version 6.0 [updated July 2019]: Tools to assess risk of bias in cohort studies (Last accessed on 2nd March, 2020). Cochrane (Available from http://methodscochraneorg/bias/sites/methodscochraneorgbias/files/public/uplo ads/Tool%20to%20Assess%20Risk%20of%20Bias%20in%20Cohort%20Studie spdf) 2019.
- 222. McHugh ML: Interrater reliability: the kappa statistic. *Biochem Med (Zagreb)* 2012, 22:276-282.
- 223. Riley RD, Higgins JP, Deeks JJ: Interpretation of random effects metaanalyses. *BMJ* 2011, 342:d549.
- 224. Higgins JPT, Thomas J, Chandler J, et al: Identifying and measuring heterogeneity. Cochrane Handbook for Systematic Reviews of Interventions, version 60 (Available from <u>https://handbook-5-</u>

<u>1cochraneorg/chapter 9/9 5 2 identifying and measuring heterogeneityhtm</u>) (Assessed on: 27th February, 2020) 2019.

- 225. R Development Core Team: R: A language and environment for statistical computing. R Foundation for Statistical Computing. Vienna, Austria; 2012.
- 226. Knorring JV: Treatment and Prognosis in Polymyalgia Rheumatica and Temporal Arteritis: A Ten-Year Survey of 53 Patients. *Acta medica Scandinavica* 1979, 205:429-435.
- 227. Godeau P, Aubert L, Guillevin L, et al: [Clinical aspects, outcome and prognosis of Horton's disease. Retrospective study of 47 cases]. *Ann Med Interne (Paris)* 1982, 133:393-400.
- 228. Behn AR, Perera T, Myles AB: Polymyalgia rheumatica and corticosteroids: how much for how long? *Ann Rheum Dis* 1983 42:374-378.
- 229. Gouet D, Marechaud R, Alcalay M, et al: Survival in giant cell arteritis: a 14-year survey of 87 patients. *The Journal of rheumatology* 1985, 12:1209-1210.
- 230. Andersson R, Malmvall BE, Bengtsson BA: Long-term corticosteroid treatment in giant cell arteritis. *Acta medica Scandinavica* 1986, 220:465-469.
- 231. Delecoeuillerie G, Joly P, Cohen-de-Lara A, et al: Polymyalgia rheumatica and temporal arteritis: a retrospective analysis of prognostic features and different corticosteroid regimens (11 year survey of 210 patients). *Ann Rheum Dis* 1988, 47:733-739.
- 232. Nesher G, Sonnenblick M, Friedlander Y: Analysis of steroid related complications and mortality in temporal arteritis: a 15-year survey of 43 patients. *The Journal of rheumatology* 1994, 21:1283-1286.
- 233. Hutchings A, Hollywood J, Lamping DL, et al: Clinical outcomes, quality of life, and diagnostic uncertainty in the first year of polymyalgia rheumatica. *Arthritis Rheum* 2007, 57:803-809.
- 234. Salvarani C, Macchioni P, Manzini C, et al: Infliximab plus prednisone or placebo plus prednisone for the initial treatment of polymyalgia rheumatica: a randomized trial. *Annals of internal medicine* 2007, 146:631-639.
- 235. Cimmino MA, Salvarani C, Macchioni P, et al: Long-term follow-up polymyalgia rheumatica patients treated with methotrexate and steroids. In *Clinical and experimental rheumatology*, vol. 26. pp. 395-400; 2008:395-400.
- 236. Schmidt WA, Moll A, Seifert A, et al: Prognosis of large-vessel giant cell arteritis. *Rheumatology (Oxford)* 2008, 47:1406-1408.
- 237. Dasgupta B, Hutchings A, Hollywood J, et al: Disease activity, quality of life and adverse events in the first two years of polymyalgia rheumatica. *Rheumatology* 2009, 48:i17.
- 238. Khalifa M, Karmani M, Study Group of GCA in Tunisia: Epidemiological and clinical features of giant cell arteritis in Tunisia. *Eur J Intern Med* 2009, 20:208-212.
- 239. Martinez-Lado L, Calviño-Díaz C, Piñeiro A, et al: Relapses and recurrences in giant cell arteritis: a population-based study of patients with biopsy-proven disease from northwestern Spain. *Medicine (Baltimore)* 2011, 90:186-193.
- 240. Mazzantini M, Torre C, Miccoli M, et al: Adverse events during longterm lowdose glucocorticoid treatment of polymyalgia rheumatica: a retrospective study. *J Rheumatol* 2012 39:552-557.
- 241. Alba MA, García-Martínez A, Prieto-González S, et al: Relapses in patients with giant cell arteritis: prevalence, characteristics, and associated clinical findings in a longitudinally followed cohort of 106 patients. *Medicine (Baltimore)* 2014, 93:194-201.
- 242. Seror R, Baron G, Hachulla E, et al: Adalimumab for steroid sparing in patients with giant-cell arteritis: results of a multicentre randomised controlled trial. In *Annals of the rheumatic diseases*, vol. 73. pp. 2074-2081; 2014:2074-2081.
- 243. Muller G, Devilliers H, Besancenot JF, et al: Giant cell arteritis (Horton's disease) in very elderly patients aged 80 years and older: A study of 25 cases. *Geriatrics and Gerontology International* 2016, 16:679-685.

- 244. Carbonella A, Berardi G, Petricca L, et al: Immunosuppressive Therapy (Methotrexate or Cyclophosphamide) in Combination with Corticosteroids in the Treatment of Giant Cell Arteritis: Comparison with Corticosteroids Alone. *Journal of the American Geriatrics Society* 2016, 64:672-674.
- 245. Faurschou M, Ahlstrom MG, Lindhardsen J, et al: Risk of Diabetes Mellitus among Patients Diagnosed with Giant Cell Arteritis or Granulomatosis with Polyangiitis: Comparison with the General Population. *Journal of Rheumatology* 2017, 44:78-83.
- 246. Van Raalte DH, Brands M, van-der-Zijl NJ, et al: Low-dose glucocorticoid treatment affects multiple aspects of intermediary metabolism in healthy humans: a randomised controlled trial. *Diabetologia* 2011, 54:2103-2112.
- 247. Strohmayer EA, Krakoff LR: Glucocorticoids and cardiovascular risk factors. *Endocrinology and Metabolism Clinics of North America* 2011, 40:409-417.
- 248. Weinstein SP, Wilson CM, Pritsker A, et al: Dexamethasone inhibits insulinstimulated recruitment of GLUT4 to the cell surface in rat skeletal muscle. *Metabolism: clinical and experimental* 1998, 47:3-6.
- 249. Ruzzin J, Wagman AS, Jensen J: Glucocorticoid-induced insulin resistance in skeletal muscles: defects in insulin signalling and the effects of a selective glycogen synthase kinase-3 inhibitor. *Diabetologia* 2005, 48:2119-2130.
- 250. Movahedi M, Beauchamp ME, Abrahamowicz M, et al: Risk of Incident Diabetes Mellitus Associated With the Dosage and Duration of Oral Glucocorticoid Therapy in Patients With Rheumatoid Arthritis. *Arthritis & rheumatology (Hoboken, NJ)* 2016, 68:1089-1098.
- 251. Sharma M, Nazareth I, Petersen I: Trends in incidence, prevalence and prescribing in type 2 diabetes mellitus between 2000 and 2013 in primary care: a retrospective cohort study. *BMJ open* 2016, 6:e010210.
- 252. Udayakumar PD, Chandran AK, Crowson CS, et al: Cardiovascular risk and acute coronary syndrome in giant cell arteritis: a population-based retrospective cohort study. *Arthritis Care Res (Hoboken)* 2015, 67:396-402.
- 253. Matthews JL, Gilbert DN, Farris BK, et al: Prevalence of diabetes mellitus in biopsy-positive giant cell arteritis. *Journal of neuro-ophthalmology : the official journal of the North American Neuro-Ophthalmology Society* 2012, 32:202-206.
- 254. Gonzalez-Gay MA, Pineiro A, Gomez-Gigirey A, et al: Influence of traditional risk factors of atherosclerosis in the development of severe ischemic complications in giant cell arteritis. *Medicine (Baltimore)* 2004, 83:342-347.
- 255. Schmidt J, Kermani TA, Muratore F, et al: Statin use in giant cell arteritis: a retrospective study. *The Journal of rheumatology* 2013, 40:910-915.
- 256. Duhaut P, Pinede L, Demolombe-Rague S, et al: Giant cell arteritis and cardiovascular risk factors: a multicenter, prospective case-control study. *Arthritis Rheum* 1998, 41:1960-1965.
- 257. Ungprasert P, Upala S, Sanguankeo A, et al: Patients with giant cell arteritis have a lower prevalence of diabetes mellitus: A systematic review and meta-analysis. *Mod Rheumatol* 2016, 26:410-414.
- 258. Li L, Neogi T, Jick S: Giant cell arteritis and vascular disease-risk factors and outcomes: a cohort study using UK Clinical Practice Research Datalink. *Rheumatology (Oxford)* 2017, 56:753-762.
- 259. Tomasson G, Peloquin C, Mohammad A, et al: Risk for cardiovascular disease early and late after a diagnosis of giant-cell arteritis: a cohort study. *Annals of internal medicine* 2014, 160:73-80.
- 260. Pujades-Rodriguez M, Duyx B, Thomas SL, et al: Associations between polymyalgia rheumatica and giant cell arteritis and 12 cardiovascular diseases. *Heart (British Cardiac Society)* 2016, 102:383-389.
- 261. Robson JC, Kiran A, Maskell J, et al: The relative risk of aortic aneurysm in patients with giant cell arteritis compared with the general population of the UK. *Ann Rheum Dis* 2015, 74:129-135.

- 262. Matteson EL, Buttgereit F, Dejaco C, et al: Glucocorticoids for Management of Polymyalgia Rheumatica and Giant Cell Arteritis. *Rheumatic diseases clinics of North America* 2016, 42:75-90.
- 263. Wilson JC, Sarsour K, Collinson N, et al: Incidence of outcomes potentially associated with corticosteroid therapy in patients with giant cell arteritis. *Semin Arthritis Rheum* 2016.
- 264. Broder MS, Sarsour K, Chang E, et al: Corticosteroid-related adverse events in patients with giant cell arteritis: A claims-based analysis. *Semin Arthritis Rheum* 2016, 46:246-252.
- 265. Hoes JN, van der Goes MC, van Raalte DH, et al: Glucose tolerance, insulin sensitivity and β -cell function in patients with rheumatoid arthritis treated with or without low-to-medium dose glucocorticoids. *Ann Rheum Dis* 2011 70:1887-1894.
- 266. Health and Social Care Information Centre: Attribution Data Set GP-Registered Populations Scaled to ONS Population Estimates – 2011-2012 (Link: <u>http://wwwhscicgovuk/catalogue/PUB05054</u> (Accessed on 2 August, 2017).
- 267. Hospital Episode Statistics (Link: <u>http://content.digital.nhs.uk/article/4966/Hospital-Episode-Statistics-HES</u>) (Accessed on 2 August 2017).
- 268. Office for National Statistics: Mortality Statistics: Metadata. *Cardiff, UK: ONS* 2014.
- 269. Index of Multiple Deprivation and Townsend Scores (Link: <u>https://census.ukdataservice.ac.uk/get-data/related/deprivation</u>) (Accessed on 2 August, 2017).
- 270. CPRD observational data (Link: <u>https://www.cprd.com/ObservationalData/CodedData.asp</u>) (Accessed on 2 August 2017).
- 271. Jick H, Jick SS, Derby LE: Validation of information recorded on general practitioner based computerised data resource in the United Kingdom. *BMJ* 1991, 302:766-768.
- 272. Khan NF, Harrison SE, Rose PW: Validity of diagnostic coding within the General Practice Research Database: a systematic review. *Br J Gen Pract* 2010, 60:e128-136
- 273. NHS Digital: Linked HES-ONS mortality data (Link: <u>http://content.digital.nhs.uk/article/2677/Linked-HES-ONS-mortality-data</u>) (Accessed on 29 September, 2017).
- 274. Department for Communities and Local Government: The English Index of Multiple Deprivation (IMD) 2015 Guidance. 2015
- 275. English indices of deprivation 2019. (Link: <u>https://wwwgovuk/government/statistics/english-indices-of-deprivation-2019</u>) (Accessed on 10th December, 2019).
- 276. Office of National Statistics, "2011 Census: Population and household estimates for small areas in England and Wales, March 2011" (Link: https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigratio n/populationestimates/bulletins/2011censuspopulationandhouseholdestimatesf orsmallareasinenglandandwales/2012-11-23) (Accessed on 2 December, 2019).
- 277. Medicines & Healthcare Products Regulatory Agency: CPRD Linked Data (Link: <u>https://www.cprd.com/home/</u>) (Accessed on 29 September, 2017).
- 278. Benchimol EI, Smeeth L, Guttmann A, et al: The REporting of studies Conducted using Observational Routinely-collected health Data (RECORD) Statement. *PLoS Med* 2015, 12:e1001885.
- 279. Paskins Z, Whittle R, Sultan AA, et al: Risk of fracture among patients with polymyalgia rheumatica and giant cell arteritis: a population-based study. *BMC Med* 2018, 16:4.

- BNF NICE Guidelines Equivalent anti-inflammatory doses of corticosteroids (Link: <u>https://bnf.nice.org.uk/treatment-summary/glucocorticoid-therapy.html</u>) (Accessed on 23 January, 2018).
- Wilson JC, Sarsour K, Collinson N, et al: Serious adverse effects associated with glucocorticoid therapy in patients with giant cell arteritis (GCA): A nested case-control analysis. Semin Arthritis Rheum 2016, 46:819-827.
- 282. Bhaskaran K, Dos-Santos-Silva, Leon DA, et al: Association of BMI with overall and cause-specific mortality: a population-based cohort study of 3.6 million adults in the UK. *Lancet Diabetes Endocrinol* 2018, 6:944-953.
- Bhaskaran K, Forbes HJ, Douglas I, et al: Representativeness and optimal use of body mass index (BMI) in the UK Clinical Practice Research Datalink (CPRD). *BMJ open* 2013, 13:9.
- 284. Marston L, Carpenter JR, Walters KR, et al: Issues in multiple imputation of missing data for large general practice clinical databases. *Pharmacoepidemiol Drug Saf* 2010, 19:618–626.
- 285. Cox DR: Regression Models and Life Tables. *Journal of the Royal Statistical Society* 1972, 34:187–220.
- 286. Royston P, Parmar MK: Flexible parametric proportional-hazards and proportional-odds models for censored survival data, with application to prognostic modelling and estimation of treatment effects. *Stat Med* 2002, 21:2175-2197.
- 287. Bellavia A: A percentile approach to time-to-event outcomes. *Thesis for doctoral degree* 2015.
- 288. Breslow NE, Lubin JH, Marek P, et al: Multiplicative models and cohort analysis. *ournal of the American Statistical Society* 1983, 78:1-12.
- 289. Thomas DC: Models for exposure-time-response relationships with applications to cancer epidemiology. *Annual Reviews of Public Health* 1988, 9:451–482.
- 290. Vacek PM: Assessing the effect of intensity when exposure varies over time. *Statistics in Medicine* 1997, 16:505–513.
- 291. Abrahamowicz M, Bartlett G, Tamblyn R, et al: Modeling cumulative dose and exposure duration provided insights regarding the associations between benzodiazepines and injuries. *Journal of Clinical Epidemiology* 2006, 59:393–403.
- 292. Sylvestre MP, Abrahamowicz M: Flexible modeling of the cumulative effects of time-dependent exposures on the hazard. *Stat Med* 2009:3437–3453.
- 293. Lambert PC, Royston P: Further development of flexible parametric models for survival analysis. *Statistics in Medicine* 2002, 21:2175–2197.
- 294. Dixon WG, Abrahamowicz M, Beauchamp ME, et al: Immediate and delayed impact of oral glucocorticoid therapy on risk of serious infection in older patients with rheumatoid arthritis: a nested case-control analysis. *Ann Rheum Dis* 2012, 71:1128-1133.
- 295. Young J, Xiao Y, Moodie EE, et al: Effect of cumulating exposure to abacavir on the risk of cardiovascular disease events in patients from the Swiss HIV Cohort Study. *J Acquir Immune Defic Syndr* 2015, 69:413-421.
- 296. Health Survey for England 2017: Adult and child overweight and obesity. *NHS Digital (LInk: <u>https://filesdigitalnhsuk/EF/AB0F0C/HSE17-Adult-Child-BMI-rep-v2pdf</u>) (Assessed on: 5th December, 2019).*
- 297. Matcho A, Ryan P, Fife D, et al: Fidelity assessment of a clinical practice research datalink conversion to the OMOP common data model. *Drug Saf* 2014, 37:945-959.
- 298. Shbeeb I, Challah D, Raheel S, et al: Comparable rates of glucocorticoidassociated adverse events in patients with polymyalgia rheumatica and comorbidities in the general population. *Arthritis Care Res* 2018, 70:643-647.
- 299. Ghosh P, Borg FA, B D: Current understanding and management of giant cell arteritis and polymyalgia rheumatica. *Expert Rev Clin Immunol* 2010, 6:913–928.

- 300. Derendorf H, Hochhaus G, Meibohm B, et al: Pharmacokinetics and pharmacodynamics of inhaled corticosteroids. *J Allergy Clin Immunol* 1998, 101:S440-446.
- 301. Meigs JB, Cupples LA, Wilson PW: Parental transmission of type 2 diabetes: the Framingham Offspring Study. *Diabetes* 2000, 49:2201-2207.
- 302. Klein BE, Klein R, Moss SE, et al: Parental history of diabetes in a populationbased study. *Diabetes Care* 1996, 19:827-830.
- 303. Suchindran S, Vana AM, Shaffer RA, et al: Racial differences in the interaction between family history and risk factors associated with diabetes in the National Health and Nutritional Examination Survey, 1999–2004. *Genet Med* 2009, 11:542-547.
- 304. InterAct Consortium, Scott RA, Langenberg C, et al: The link between family history and risk of type 2 diabetes is not explained by anthropometric, lifestyle or genetic risk factors: the EPIC-InterAct study. *Diabetologia* 2013, 56:60-69.
- 305. Kral BG, Becker DM, Yanek LR, et al: The relationship of family history and risk of type 2 diabetes differs by ancestry. *Diabetes Metab* 2019, 45:261-267.
- 306. Wainwright NW, Surtees PG: Places, people and their physical and mental functional health. *J Epidemiol Community Health* 2004, 58:333-339.
- 307. Hayward RA, Rathod T, Muller S, et al: Association of polymyalgia rheumatica with socioeconomic status in primary care: a cross-sectional observational study. *Arthritis Care Res (Hoboken)* 2014, 66:956-960.
- 308. Barraclough K, Liddell WG, du Toit J, et al: Polymyalgia rheumatica in primary care: a cohort study of the diagnostic criteria and outcome. *Fam Pract* 2008, 25:328-333.
- 309. Helliwell T: Polymyalgia rheumatica in primary care: an exploration of the challenges of diagnosis and management using survey and qualitative methods (PhD thesis). *Research Institute of Primary Care Sciences, Keele University* 2016.
- 310. Helliwell T, Muller S, Hider SL, et al: Challenges of diagnosing and managing polymyalgia rheumatica: a multi-methods study in UK general practice. *Br J Gen Pract* 2018, 68:e783-e793.
- 311. Kremers HM, Reinalda MS, Crowson CS, et al: Relapse in a population based cohort of patients with polymyalgia rheumatica. *The Journal of rheumatology* 2005, 32:65-73.
- 312. Kyle V, Hazleman BL: The clinical and laboratory course of polymyalgia rheumatica/giant cell arteritis after the first two months of treatment. *Ann Rheum Dis* 1993, 52:847-850.
- 313. Quick V, Kirwan JR: Our approach to the diagnosis and treatment of polymyalgia rheumatica and giant cell (temporal) arteritis. *J R Coll Physicians Edinb* 2012, 42:341-349.
- 314. Gale S, Wilson JC, Chia J, et al: Risk associated with cumulative oral glucocorticoid use in patients with giant cell arteritis in real-world databases from the USA and UK. *Rheumatol Ther* 2018, 5:327-340.
- 315. Lai KH, Topaz M, Goss FR, et al: Automated misspelling detection and correction in clinical free-text records. *J Biomed Inform* 2015, 55:188–195.
- 316. Patrick J, Nguyen D: Automated Proof Reading of Clinical Notes. *PACLIC* 2011:303-312.
- 317. Rahm E, Hong HD: Data Cleaning: Problems and Current Approaches. *IEEE Data Eng Bull* 2000, 23:3-13.
- 318. Tolentino HD, Matters MD, Walop W, et al: A UMLS-based spell checker for natural language processing in vaccine safety. *BMC Med Inform Decis Mak* 2017, 12:3.
- 319. Hunder GG, Sheps SG, Allen GL, et al: Daily and alternate-day glucocorticosteroid regimens in treatment of giant cell arteritis: comparison in a prospective study. *Annals of internal medicine* 1975:613-618.

- 320. Herrett E, Thomas SL, Schoonen WM, et al: Validation and validity of diagnoses in the General Practice Research Database: a systematic review. *Br J Clin Pharmacol* 2010, 69:4–14.
- 321. Holden SH, Barnett AH, Peters JR, et al: The incidence of type 2 diabetes in the United Kingdom from 1991 to 2010. *Diabetes Obes Metab* 2013, 15:844-852.
- 322. Prior JA , Ranjbar H , Belcher J , et al: Diagnostic delay for giant cell arteritis a systematic review and meta-analysis. *BMC Med* 2017, 15:120.
- 323. Ezeonyeji AN, Borg FA, Dasgupta B: Delays in recognition and management of giant cell arteritis: results from a retrospective audit. *Clin Rheumatol* 2011, 30:259–262.
- 324. Kennedy S: Polymyalgia rheumatica and giant cell arteritis: an in-depth look at diagnosis and treatment. *J Am Acad Nurse Pract* 2012, 24:277–285.
- 325. Chakravarty K, Elgabani SH, Scott DG, et al: A district audit on the management of polymyalgia rheumatica and giant cell arteritis. *Br J Rheumatol* 1994, 33:152–156.
- 326. World Health Organization: Definition and diagnosis of diabetes mellitus and intermediate hyperglycemia: Report of a WHO/IDF consultation. 2006.
- 327. VA Fonseca: Defining and characterizing the progression of type 2 diabetes. *Diabetes Care* 2009, 32:S151–S156.
- 328. Nichols GA, Hillier TA, Brown JB, et al: Progression from newly acquired impaired fasting glucose to type 2 diabetes. *Diabetes Care* 2007, 30:228-233.
- 329. Meigs JB, Muller DC, Nathan DM, et al: Baltimore Longitudinal Study of Aging. The natural history of progression from normal glucose tolerance to type 2 diabetes in the Baltimore Longitudinal Study of Aging. *Diabetes* 2003, 52:1475–1484.
- 330. Ferrannini E, Nannipieri M, Williams K, et al: Mode of onset of type 2 diabetes from normal or impaired glucose tolerance. *Diabetes* 2004, 53:160–165.
- 331. Weyer C, Tataranni PA, Bogardus C, et al: Insulin resistance and insulin secretory dysfunction are independent predictors of worsening of glucose tolerance during each stage of type 2 diabetes development. *Diabetes Care* 2001, 24:89–94.
- 332. Festa A, Williams K, D'Agostino R Jr, et al: The natural course of beta-cell function in nondiabetic and diabetic individuals: the Insulin Resistance Atherosclerosis Study. *Diabetes* 2006, 55:1114–1120.
- 333. NHS Digital: National Diabetes Audit, 2015-2016: Report 1: Care Processess and Treatment Targets. 2017.
- 334. Centers for Disease Control and Prevention (CDC): National Diabetes Statistics Report. 2017.
- 335. Tate AR, Dungey S, Glew S, et al: Quality of recording of diabetes in the UK: how does the GP's method of coding clinical data affect incidence estimates? Cross-sectional study using the CPRD database. *BMJ open* 2017, 7:e012905.
- 336. Maruthur NM: The growing prevalence of type 2 diabetes: increased incidence or improved survival? . *Curr Diab Rep* 2013, 13:786–794.
- 337. Suissa S: Immortal time bias in pharmaco-epidemiology. *Am J Epidemiol* 2008, 167:492-499.
- 338. European Network of Centres for Pharmacoepidemiology and Pharmacovigilence (ENCePP) Guide on Methodological Standards in Pharmacoepidemiology - Section 4.2.2.5. Unmeasured confounding (*Link:* <u>http://wwwenceppeu/standards and guidances/documents/ENCePPGuideofM</u> <u>ethStandardsinPE_Rev5pdf</u>) (Assessed on 1st December, 2019).
- 339. European Network of Centres for Pharmacoepidemiology and Pharmacovigilence (ENCePP) Guide on Methodological Standards in Pharmacoepidemiology - Section 4.2.2 Bias and confounding (Link: <u>http://wwwenceppeu/standards and guidances/documents/ENCePPGuideofM</u> <u>ethStandardsinPE_Rev5pdf</u>) (Assessed on 1st December, 2019).
- 340. Han PY, Duffull SB, Kirkpatrick CM, et al: Dosing in obesity: a simple solution to a big problem. *Clin Pharmacol Ther* 2007, 82:505-508.

- 341. Barras M, Legg A: Drug dosing in obese adults. *Aust Prescr* 2017, 40:189-193.
- 342. Duru N, van der Goes MC, Jacobs JW, et al: EULAR evidence-based and consensus-based recommendations on the management of medium to high-dose glucocorticoid therapy in rheumatic diseases. *Ann Rheum Dis* 2013, 72:1905–1913.
- 343. Joint British Diabetes Societies for Inpatient Care (JBDS-IP): Management of hyperglycaemia and steroid (glucocorticoid) therapy. 2014, JBDS 08.
- 344. American Diabetes Association: Diabetes in older adults. *Diabetes Care* 2012, 35:2650-2664.
- 345. Lipska KJ, Bailey CJ, Inzucchi SE: Use of metformin in the setting of mild-tomoderate renal insufficiency. *Diabetes Care* 2011, 34:1431–1437.
- 346. National Institute for Health and Clinical Excellence: Type 2 diabetes in adults: management. *NICE Guidelines* 2015.
- 347. Dashora UK, Taylor R: Maintaining glycaemic control during high-dose prednisolone administration for hyperemesis gravidarum in Type 1 diabetes. *Diabetic Medicine* 2004, 21:298-299.
- 348. Kassirer JP: Incorporating patients' preferences into medical decisions. *N Engl J Med* 1994, 330:1895–1896.
- 349. Barry MJ, Edgman-Levitan S: Shared decision making—pinnacle of patientcentered care. *N Engl J Med* 2012, 366:780–781.
- 350. Montori VM, Gafni A, Charles C: A shared treatment decision-making approach between patients with chronic conditions and their clinicians: the case of diabetes. *Health Expect* 2006, 9:25–36.
- 351. Stone JH, Tuckwell K, Dimonaco S, et al: Trial of Tocilizumab in giant cell arteritis. *N Engl J Med* 2017, 377:317-328.
- 352. National Institute for Health and Clinical Excellence (NICE): Single technology appraisal on Tolizumab for treating giant cell arteritis (Committee Papers). 2018.
- 353. Greenstone MA, Shaw AB: Alternate day corticosteroid causes alternate day hyperglycemia. *Postgrad Med J* 1987, 63:761.
- 354. Garber AJ: Obesity and type 2 diabetes: which patients are at risk? *Diabetes Obes Metab* 2012, 14:399-408.
- 355. Flegal KM, Carroll MD, Kit BK: Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. *JAMA* 2012, 307:491-497.
- 356. Kodama S, Horikawa C, Fujihara K, et al: Comparisons of the strength of associations with future type 2 diabetes risk among anthropometric obesity indicators, including waist-to-height ratio: a meta-analysis. *Am J Epidemiol* 2012, 176:959-969.
- 357. Kahn SE, Hull RL, Utzschneider KM: Mechanisms linking obesity to insulin resistance and type 2 diabetes. *Nature* 2006, 444:840-846.
- 358. Felber JP, Golay A, Jéquier E, et al: The metabolic consequences of long-term human obesity. *Int J Obes* 1988, 12:377-389.

APPENDICES

APPENDIX 1: 'MeSH' Search Terms

Population

- 1. polymyalgia rheu*
- 2. giant cell art*
- 3. temporal art*
- 4. large vessel vasculiti*

Intervention/ Exposure

- 5. glucocortico*
- 6. corticostero*
- 7. steroid*
- 8. predniso*

Outcomes

9. diabet* 10.impaired glucose tolerance 11.glucose 12.hyperglyc* 13.HbA1C 14.glycated h* 15.#13 or #14 16.HOMA 17.obes* 18.overweight 19.BMI 20.body mass index* 21.#19 or #20 22.waist* 23.hip-waist ratio 24.hip waist ratio 25.#22 or #23 or #24 26. #1 or #2 or #3 or #4 27. #5 or #6 or #7 or #8 28. #9 or #10 or #11 or #12 or #15 or #16 or #17 or #18 or #21 or #25

Final Search Terms

#26 and #27 and #28

APPENDIX 2: MOOSE Guidelines for meta-analyses and

systematic reviews of observational studies

Sections	Criteria
	Problem definition
	Hypothesis statement
Reporting of	 Description of study outcome(s)
background	 Type of exposure or intervention used
	Type of study designs used
	Study population
	 Qualifications of searchers (e.g. librarians and investigators)
	• Search strategy, including time period included in the synthesis and
	Keywords
	Efforts to include all available studies, including contact with authors Detables and registering as each ad
Departing of	Databases and registries searched Operational sectors and registries and registries and registries and registries are set of the sectors and registries are set of the sectors and registries are set of the sectors are sectors are set of the sec
Reporting of	Search software used, name and version, including special features
strategy	articles)
	 List of citations located and those excluded including justification
	 Method of addressing articles published in languages other than
	English
	 Methods of handling abstracts and unpublished studies
	Description of any contact with authors
	• Description of relevance or appropriateness of studies assembled
Reporting of	for assessing the hypothesis to be tested
	• Rationale for the selection and coding of data (e.g. sound clinical
	principles or convenience)
	Documentation of how data were classified and coded (e.g. multiple
	raters, blinding, and interrater reliability)
	 Assessment of contounding (e.g. comparability of cases and controls in studies where appropriate)
	• Assessment of study quality including blinding of quality assessors:
methods	stratification or regression on possible predictors of study results
	Assessment of heterogeneity
	 Description of statistical methods (e.g. complete description of fixed
	or random effects models, justification of whether the chosen
	models account for predictors of study results, dose-response
	models, or cumulative meta-analysis) in sufficient detail to be
	replicated
	Provision of appropriate tables and graphics
Reporting of	• Graphic summarizing individual study estimates and overall
	estimate Table while a description information for a set of other included
results	I able giving descriptive information for each study included
	 Results of sensitivity testing (e.g. subgroup analysis) Indication of statistical uncortainty of findings
	Output to statistical uncertainty of Indings Output tative assessment of bias (a.g. publication bias)
Reporting of	 Quantitative assessment of bias (E.y. publication bias) Justification for evolution (e.g. evolution of non-English longuage)
discussion	citations)
	Assessment of quality of included studies

APPENDIX 3: ISAC Protocol

Safety and associated costs of glucocorticoid therapy for the treatment of chronic inflammatory diseases

Date of ISAC Approval:

28/07/2016

Lay Summary:

Steroids are a group of medications that are widely used to treat many different diseases of long duration, such as asthma, and their prolonged use has increased over time in the UK despite their side effects. Side effects can develop within days (e.g. mental illness) or after prolonged use (e.g. high blood pressure). Side effects can have a major impact on quality of life, independence and wellbeing, particularly as people get older, and lead to many hidden costs to the individual and the healthcare system. For many conditions there are no approved alternatives to steroids and it is important we obtain accurate information about the side effects of treatment and the full costs of these to the individual and the NHS. This information is very helpful in building a strong case to support the introduction of new treatment, which have been shown to be effective in clinical trials, into the NHS. With CPRD data we intend to develop new methods that allow clinicians and policy makers to balance risks and benefits of steroid treatment in patients with common long-lasting diseases. This will be achieved first by describing the side effects.

Technical Summary:

The aim of this research is to develop tools that enable clinicians and policy decision makers to balance harms and benefits of glucocorticoids for the treatment of chronic inflammatory diseases. This will be achieved through; 1) description of glucocorticoid toxicity profiles and its predictors; and 2) development of treatment stratification tools for clinical evaluation to accurately identify patients at increased risk of developing toxicity. This research will enable informed personalised adaptation of treatment for patients identified with high risk of toxicity (e.g. earlier dose reduction or use of glucocorticoid-sparing therapies), improvement of clinical guidelines and quality of care and equity in health care provision, ultimately improving long-term patient outcomes. To estimate toxicity rates and describe toxicity patterns we will expand our phenotyping work to identify patients with common chronic inflammatory diseases and adverse events. The description of common patterns of toxicity, overall and in patients with different underlying diseases and age groups, and the exploration of the relative importance of duration of medication and dose will inform the construction or isk prediction models to enable patient risk stratification according to risk of toxicity development and the future development of decision cost-models to guide policy and clinical decision making.

Health Outcomes to be Measured:

Glucocorticoid toxicity All-cause hospitalisation Hospitalisation for toxicity Death for toxicity All-cause mortality

Collaborators:

Dr Mar Pujades Rodriguez - Chief Investigator - University Of Leeds Ann Morgan - Researcher - University Of Leeds Paul David Baxter - Researcher - University Of Leeds Paul Stewart - Researcher - University Of Leeds

APPENDIX 4: R Codes

Appendix 4.1. Cohort

DETERMINE START OF FOLLOW-UP DATE

Determine the latest date of the following criteria # Criteria 1: Diagnosis of PMR and/or GCA (Variable = first_diagnosis)

Criteria 2: Start of CPRD: 1/1/1998 Cohort\$CPRDStart_DailyD <-as.Date("1998-01-01")

Criteria 3: UTS date + 1 year Cohort\$NewUTS <- ymd(Cohort\$uts) %m+% years(1)</pre>

Criteria 4: GP Registration date + 1 year Cohort\$NewRegDate <- ymd(Cohort\$current_registration_date) %m+% years(1)</pre>

Criteria 5: At least 18 years old (Set birth year to 07/01/YYYY format) Cohort\$month <- "7" Cohort\$day <- "1" Cohort\$DOB <- ISOdate(Cohort\$birth_year, Cohort\$month, Cohort\$day) Cohort\$DOB <- strptime(Cohort\$DOB, format = "%Y-%m-%d")</pre>

Calculate baseline age during study entry / age at diagnosis
Cohort\$Age_days <- difftime(Cohort\$study_entry_date, Cohort\$DOB, units =
c("days"))
Cohort\$Age_years <- Cohort\$Age_days/365.25</pre>

Check if any patients below 18yo (None!)
Below18 <- Cohort [which(Cohort\$Age_years < 18),]</pre>

Aggregate by ID and find latest date Cohort2 <- Cohort %>% mutate(LatestDate = pmax(first_diagnosis, CPRDStart_DailyD, NewUTS, NewRegDate, na.rm = TRUE))

Note (1): All the data was checked twice Note (2): I compared the study entry date constructed from my cohort with the given study entry date, they were identical.

DETERMINE BASELINE DM STATUS

File 1: Anti-DM Medication # Aggregate by ID and find earliest Rx date AntiDM_Earliest <- setDT(AntiDM)[order(eventdate), head(.SD, 1L), by="patid"]</p>

File 2: DM Diagnosis
Create subset of patients with any kind of DM - H/O, T1, T2, secondary DM, Unspecified, gestational, steroid induced
AnyDM <- subset(DM_Dx, DM_Dx\$X0..H.O.AII!="NULL" |</p>
DM_Dx\$X1.Diab.1.AII!="NULL" | DM_Dx\$X2.Diab.2.AII!="NULL" |
DM_Dx\$X3.2ry.diab.AII!="NULL" | DM_Dx\$X4.Diab.unspec.AII!="NULL" |
DM_Dx\$X5.Gestation.AII!="NULL" | DM_Dx\$X6.Steroid.related.AII!="NULL")

Determine earliest DM diagnosis date

AnyDM <- AnyDM %>% mutate(Earliest_DMDx = pmin(AnyDM\$X0..H.O.All, AnyDM\$X1.Diab.1.All, AnyDM\$X2.Diab.2.All, AnyDM\$X3.2ry.diab.All, AnyDM\$X4.Diab.unspec.All, AnyDM\$X5.Gestation.All, AnyDM\$X6.Steroid.related.All))

Merge files (Cohort, Anti-DM medication & DM diagnosis) # Stage 1: Merge cohort with those who were on antiDM drugs CPRD1 <- merge(Cohort, AntiDM_Earliest, by=c("patid"), all.x=TRUE)</pre>

Stage 2: Merge cohort+those on meds with those who had a DM diagnosis CPRD2 <- merge(CPRD1, AnyDM, by=c("patid"), all.x=TRUE)

Select earliest of the following dates (DM diagnosis or Anti-DM prescription)
Earliest_DMDx = earliest date of DM diagnosis
eventdate = earliest prescription date for antiDM medications
CPRD2 <- CPRD2 %>% mutate(EarliestDM_Dx_Rx = pmin(CPRD2\$Earliest_DMDx,
CPRD2\$eventdate))

Populate the "EarliestDM_Dx_Rx" rows with missing values using the "Earliest_DMDx" values (as some patients have a DM diagnosis but did not have any DM prescription) CPRD2\$Earliest_DMDx[CPRD2\$Earliest_DMDx == "NULL"] = NA CPRD2\$Earliest_DM_Dx_Rx_Final <- CPRD2\$EarliestDM_Dx_Rx CPRD2\$Earliest_DM_Dx_Rx_Final <- ifelse (!is.na(CPRD2\$Earliest_DM_Dx_Rx_Final), CPRD2\$Earliest_DM_Dx_Rx_Final, CPRD2\$Earliest_DMDx)

Determine DM status prior to GCA/PMR diagnosis CPRD2\$DM <- "NoDM" CPRD2\$DM[(CPRD2\$Earliest_DM_Dx_Rx_Final >= CPRD2\$first_diagnosis)] <- "DM" CPRD2\$DM[(CPRD2\$Earliest_DM_Dx_Rx_Final < CPRD2\$first_diagnosis)] <-"Pre_existing_DM"

Exclude patients with pre-existing DM CPRD3 <- subset(CPRD2, DM!="Pre_existing_DM")</pre>

Exclude patients with T1DM # Definition of T1DM = Have a diagnosis date of Type1 DM + only on Insulin & not on any other oral hyperglycemic agents CPRD3\$T1DM <- "Not_T1DM" CPRD3\$T1DM[(CPRD3\$X1.Diab.1.All!="NULL") & (CPRD3\$insulin=="1") & (CPRD3\$antidiab=="NULL")] <- "T1DM" CPRD3_NoT1DM <- subset(CPRD3, T1DM!="T1DM")</pre>

DETERMINE DEATH DATE

Combine death date from ONS & HES (priority given to ONS) CPRD3_NoT1DM\$death_date <- CPRD3_NoT1DM\$ONS_death_date CPRD3_NoT1DM\$death_date <- ifelse (CPRD3_NoT1DM\$ONS_death_date == "NULL", CPRD3_NoT1DM\$CPRD_death_date, CPRD3_NoT1DM\$ONS_death_date)

Categorize into 4 groups CPRD3_NoT1DM\$Event_First [(CPRD3_NoT1DM\$death_date!="NULL") & (CPRD3_NoT1DM\$DM=="DM")] <- "DM_Died" CPRD3_NoT1DM\$Event_First [(CPRD3_NoT1DM\$death_date!="NULL") & (CPRD3_NoT1DM\$DM=="NoDM")] <- "NoDM_Died"</pre> CPRD3_NoT1DM\$Event_First [(CPRD3_NoT1DM\$death_date=="NULL") & (CPRD3_NoT1DM\$DM=="DM")] <- "DM_Alive" CPRD3_NoT1DM\$Event_First [(CPRD3_NoT1DM\$death_date=="NULL") & (CPRD3_NoT1DM\$DM=="NoDM")] <- "NoDM_Alive"

DETERMINE END OF FOLLOW UP DATE

Determine the earliest date of the following variables: # 1: Latest contact (latest date where the GP submitted the data) # 2: Transfer out date / Deregistration date (patient left the GP) # 3: Death # 4: DM CPRD3_NoT1DM <- CPRD3_NoT1DM %>% mutate(End_of_FU = pmin(latest_contact, transfer_out_date, death_date, Earliest_DM_Dx_Rx_Final, na.rm = TRUE))

Calculate FU duration (from study entry date to end of FU) CPRD3_NoT1DM\$FU_Days <- difftime (CPRD3_NoT1DM\$End_of_FU, CPRD3_NoT1DM\$study_entry_date, units=c("days"))

Exclude patients with negative follow-up dates CPRD_Final <- subset(CPRD3_NoT1DM, FU_Days >= 0)

Replace those with "0" FU days as 0.1 CPRD_Final\$FU_Days [CPRD_Final\$FU_Days <1] <- 0.1 CPRD_Final\$FU_Years <- CPRD_Final\$FU_Days/365.25</pre>

RESTRICT TO NEWLY DIAGNOSED PMR/GCA PATIENTS

New_Cohort <- subset(CPRD_Final, first_diagnosis>=study_entry_date)

Note: All the data was checked at every stage, especially the "NULL" & "NA" values

Appendix 4.2. Covariates

DETERMINE USE OF NON-ORAL GC DURING STUDY FU

Merge cohort & Rx data Covariates <- merge (New_Cohort, Covariate, by="patid")</pre>

Create Y/N variables for all non-oral GC Covariates\$Non_OralGC_Use <- "Yes" Covariates\$Non_OralGC_Use[(Covariates\$articgc=="NULL" & Covariates\$imgc=="NULL" & Covariates\$rectgc=="NULL" & Covariates\$topgc=="NULL" & Covariates\$inhgc=="NULL" & Covariates\$nasgc=="NULL" & Covariates\$unkgc=="NULL")] <- "No" Covariates\$Non_OralGC_Use[(Covariates\$fludrocortisone=="1")] <- "No"</pre>

Change from long to wide format Unique_Reshape <- reshape(Unique, idvar = "patid", v.names = "Non_OralGC_Use", timevar = "Non_OralGC_Use", direction = "wide")

Unique_Reshape\$NonOralGC_Final <- "No" Unique_Reshape\$NonOralGC_Final[(Unique_Reshape\$Non_OralGC_Use.Yes=="Yes")] <- "Yes" Unique_Reshape\$NonOralGC_Final[(Unique_Reshape\$Non_OralGC_Use.Yes=="Yes" " & Unique_Reshape\$Non_OralGC_Use.No=="No")] <- "Yes"

FAMILY HISTORY OF DM

Merge cohort & family history file
DM <- merge (New_Cohort, DM_Dx, by="patid")</pre>

Create Y/N variables DM\$FamilyHx <- "Yes" DM\$FamilyHx[(DM\$X.2..F.H.CPRD=="NULL" & DM\$X.2..F.H.HES=="NULL" & DM\$X.2..F.H.ONS=="NULL" & DM\$X.2..F.H.All=="NULL"] <- "No"

<u>BMI</u>

Impute missing data for height (using mean)
BMI_Mean <- aggregate (height.m.~patid, BMI_Weight, mean, na.rm=TRUE)</pre>

Calculate BMI BMI_Merged\$Calculated_BMI <- BMI_Merged\$weight.kg. / (BMI_Merged\$Mean_height * BMI_Merged\$Mean_height)

Categorize BMI as either baseline, prebaseline or postbaseline (by comparing event date vs. study entry date) BMI\$DateDiff <- difftime (BMI\$eventdate, BMI\$study_entry_date, units = "days") BMI\$DateDiff Year <- BMI\$DateDiff/365.25</pre>

Baseline = BMI recorded ON or within 1 year prior to entry

Pre-baseline = Average of all recorded values recorded during the period between 2-5 years BEFORE study entry

Post-baseline = Average of all recorded values recorded within 1 year AFTER study entry

BMI\$BMI_Cat <- "NR"

BMI\$BMI_Cat[(BMI\$DateDiff_Year > 0) & (BMI\$DateDiff_Year <= 1)] <- "BMI_Post" BMI\$BMI_Cat[(BMI\$DateDiff_Year > -1) & (BMI\$DateDiff_Year <= 0)] <-"BMI_Baseline" BMI\$BMI_Cat[(BMI\$DateDiff_Year > -5) & (BMI\$DateDiff_Year <= -1)] <- "BMI_Pre"

Subset baseline data only
BMI_Baseline <- subset(BMI, BMI_Cat=="BMI_Baseline")</pre>

Find mean of BMI for each patient at baseline (n=8276) BMI_Final <- aggregate (BMI_Final~patid+BMI_Cat, BMI_Baseline, mean, na.rm=TRUE)

<u>SMOKING</u>

Baseline = Smoking status recorded ON or within 1 year prior to entry (If have evidence of ex-smoker - change!)

Pre-baseline = All smoking status codes recorded BEFORE the year prior to entry and use the higher level of exposure

Post-baseline = All smoking status codes recorded in the year AFTER the date of entry and use the higher level of exposure

For patients w/o data at baseline, use last observation carried forward or backwards

Categorize smoking as either baseline, prebaseline or postbaseline (by comparing event date vs. study entry date)

Smoking\$DateDiff <- difftime (Smoking\$eventdate, Smoking\$study_entry_date, units = "days")

Smoking\$DateDiff_Year <- Smoking\$DateDiff/365.25

SmokingSmoking_Cat <- "Unknown"</pre>

Smoking\$Smoking_Cat[(Smoking\$DateDiff_Year > 0) & (Smoking\$DateDiff_Year <= 1)] <- "SmokingStatus_Post"

Smoking\$Smoking_Cat[(Smoking\$DateDiff_Year > -1) & (Smoking\$DateDiff_Year <= 0)] <- "SmokingStatus_Baseline"

Smoking\$Smoking_Cat[(Smoking\$DateDiff_Year > -2) & (Smoking\$DateDiff_Year <= -1)] <- "SmokingStatus_Pre"

Determine highest level of exposure (2 examples shown) Smoking\$Smoking[(Smoking\$SmokingStatus=="Non smoker (1)" & Smoking\$Smoking_Cat=="SmokingStatus_Baseline")] <- "NonSmoker_Baseline" Smoking\$Smoking[(Smoking\$SmokingStatus=="Ex smoker (2)" & Smoking\$Smoking_Cat=="SmokingStatus_Baseline")] <- "ExSmoker_Baseline"</pre>

Note: Checks were performed throughout the process

Appendix 4.3. Prescription Database

DETERMINE DOSING FREQUENCY

Merge cohort & Rx data Cohort_Rx <- merge (Rx, New_Cohort, by="patid")</pre>

Select Rxs between study entry date to end of FU
CPRD <- subset (Cohort_Rx, prescription_date >= study_entry_date &
prescription_date <= End_of_FU)</pre>

Create subset for oral GCs (exclude all non oral GCs)
Rx <- subset (CPRD, !orgc=="NULL")</pre>

Extract information on frequency of use from "dosage instructions" variable (free text) # Check how many unique dosage instructions (n=3016) – all these were screened manually

Check.directions <- unique(Rx \$dose)

Categorize dosing frequency (2 examples shown for each category)
Rx[grep ("NOW", Rx\$dose), "Directions"] <- "Stat"
Rx[grep ("STAT", Rx\$dose), " Directions "] <- "Stat"</pre>

Rx[grep ("^AD", Rx\$dose), " Directions "] <- "As directed" Rx[grep ("^AS RECOMMENDED", Directions \$dose), "Rx"] <- "As directed"

Rx[grep ("/DAY", Rx\$dose), " Directions "] <- "OD" Rx[grep ("2 TAB DAILY", Rx\$dose), " Directions "] <- "OD"

Rx[grep ("EVERY OTHER DAY", Rx\$dose), " Directions "] <- "EOD" Rx[grep ("EOD", Rx\$dose), " Directions "] <- "EOD"

Rx[grep ("PRN", Rx\$dose), " Directions "] <- "PRN" Rx[grep ("AS REQD", Rx\$dose), " Directions "] <- "PRN"

Rx[grep ("A WEEK", Rx\$dose), " Directions "] <- "Weekly" Rx[grep ("WEEKLY", Rx\$dose), " Directions "] <- "Weekly"

Rx[grep ("1EVERY3DAYS", Rx\$dose), " Directions "] <- "2x/wk" Rx[grep ("TWICE/WEEK", Rx\$dose), " Directions "] <- "2x/wk" Rx[grep ("1 MANE 1 EVENING", Rx\$dose), " Directions "] <- "BD" Rx[grep ("1 MANE 2 NOCTE", Rx\$dose), " Directions "] <- "BD"

Rx[grep ("TID", Rx\$dose), " Directions "] <- "TDS" Rx\$Rx [(Rx\$dose=="ONE EVERY EIGHT HOURS")] <- "TDS"

Rx[grep ("THREE TIMES A WEEK", Rx\$dose), " Directions "] <- "3x/wk" Rx[grep ("THREE TIMES PER WEEK", Rx\$dose), " Directions "] <- "3x/wk"

Rx[grep ("FOUR TIMES", Rx\$dose), " Directions "] <- "4x/d" Rx[grep ("QID", Rx\$dose), " Directions "] <- "4x/d"

Rx[grep ("FIVE TIMES", Rx\$dose), " Directions "] <- "5x/d" Rx[grep ("5X DAILY", Rx\$dose), " Directions "] <- "5x/d" Rx[grep ("REDUCING", Rx\$dose), " Directions "] <- "Taper" Rx[grep ("54321", Rx\$dose), " Directions "] <- "Taper" Rx\$Rx [(Rx\$dose=="100MG")] <- "Unknown" Rx\$Rx [(Rx\$dose=="AFTER FOOD")] <- "Unknown"

Change frequency to "numeric" version (e.g. OD = 1) Rx[grep ("Stat", Rx\$Rx), "Frequency"] <- "1" Rx[grep ("PRN", Rx\$Rx), "Frequency"] <- "1" Rx[grep ("As directed", Rx\$Rx), "Frequency"] <- "0" Rx[grep ("Unknown", Rx\$Rx), "Frequency"] <- "0" Rx[grep ("OD", Rx\$Rx), "Frequency"] <- "1" Rx[grep ("EOD", Rx\$Rx), "Frequency"] <- "0.5" Rx[grep ("Weekly", Rx\$Rx), "Frequency"] <- "0.143" Rx[grep ("2x/wk", Rx\$Rx), "Frequency"] <- "0.286" Rx[grep ("3x/wk", Rx\$Rx), "Frequency"] <- "0.429" Rx[grep ("BD", Rx\$Rx), "Frequency"] <- "2" Rx[grep ("TDS", Rx\$Rx), "Frequency"] <- "3" Rx[grep ("4x/d", Rx\$Rx), "Frequency"] <- "4" Rx[grep ("5x/d", Rx\$Rx), "Frequency"] <- "5" Rx[grep ("6x/d", Rx\$Rx), "Frequency"] <- "6" Rx[grep ("Taper", Rx\$Rx), "Frequency"] <- "1"

Note: Checks were done on 18/1/18 & 24/6/18 & 21/9/19

DETERMINE NUMBER OF TABS PER DOSE

File 1: OD / EOD # Create subset for OD & EOD dosing OD <- subset (Rx, Directions == "EOD" | Directions == "OD")

Check number of unique instructions (n=2138)
OD_Unique <- unique(OD[c("dose", "daily_dose", "Tabs_dose")])</pre>

Conversion of MG to number of tabs OD\$Tabs_dose [(OD\$dose=="10 MG DAILY" | OD\$dose=="10 MG EVERY DAY") & (OD\$strength=="2.5mg")] <- "4"</pre>

Check all "ML" related dosing – make sure that they were all per ML OD\$Tabs_dose [(OD\$dose=="THREE 5ML SPOONSFUL TO BE TAKEN IN THE MORNING")] <- "3" #5ml = 1 tab (5mg) Note: The same process was related for all other files - BD, TDS, As directed etc # Combine all individual files A1 <- rbind (OD, BD) A2 <- rbind (A1, TDS) A3 <- rbind (A2, MT) A4 <- rbind (A3, SPU) A5 <- rbind (A4, AD) A6 <- rbind (A5, WK) Rx_Final <- rbind (A6, Taper)</pre>

Note: The length of file was checked to make sure that the total tallied the original Rx file

DETERMINE PREDNISOLONE EQUIVALENT DOSE

Total unique GC products = 68
Products <- unique (Rx\$product_name)</pre>

Categorize PED (2 examples shown)
Rx[grep ("Deltacortril 2.5mg gastro-resistant tablets", Rx\$product_name),
"predn_equiv"] <- "2.5"
Rx\$predn_equiv [(Rx\$product_name=="Dexamethasone 2mg/5ml oral solution")] <"2.667" #per ML</pre>

Note: Checks were done on all 68 products on 18/1/18 & 24/6/18

DETERMINE NUMBER OF TABS PER DAY

Method 1: Tabs per day = Quantity prescribed / duration in CPRD+imputed from "directions for use" variable Data\$Tabs_day_CPRD <- Data\$quantity_prescribed/Data\$duration</p>

Method 2: Tabs per day = Number of tabs per dose * frequency (for Rxs with quantifiable instructions) Data\$Tabs_day <- Data\$Tabs_dose*Data\$Frequency</pre>

Method 3: Tabs per day = Quantity prescribed / Difference in 2 consecutive dates # Order in ascending order by dates & patid, then determine duration based on 2 consecutive dates Data <- orderBy(~patid+prescription_date, data=Data) Data <- Data %>% group_by (patid) %>% mutate (Duration_Dates = c(NA, round(diff.difftime(prescription_date, units="days")))) %>% select(everything())

Data\$Tabs_day_Dates <- Data\$quantity_prescribed/Data\$Duration_Dates

Merge ALL THREE tabs per day variables
Priority 1: Tabs_day_CPRD
Priority 2: Tabs_day
Priority 3: Tabs_day_Dates

Merge first 2 & then check if there are any infinite, undefined or missing values!
Data\$Tabs_day_Final1 <- Data\$Tabs_day_CPRD
Data\$Tabs_day_Final1 <- ifelse ((Data\$Tabs_day_CPRD == "Inf" |
Data\$Tabs_day_CPRD == "NaN" | is.na(Data\$Tabs_day_CPRD)), Data\$Tabs_day_CPRD,
Data\$Tabs_day_CPRD)</pre>

Check <- Data[c("Tabs_day_Final1", "Tabs_day_CPRD", "Tabs_day")] sum(is.na(Data\$Tabs_day_Final1)) #0 sum(is.infinite(Data\$Tabs_day_Final1)) #0 sum(is.nan(Data\$Tabs_day_Final1)) #0

Merge 1, 2 & 3
Data\$Tabs_day_Final <- Data\$Tabs_day_Final1
Data\$Tabs_day_Final <- ifelse ((Data\$Tabs_day_Final1 == "0"),
Data\$Tabs_day_Dates, Data\$Tabs_day_Final1)
Check <- Data[c("Tabs_day_CPRD", "Tabs_day", "Tabs_day_Dates",
"Tabs_day_Final")]
sum(is.na(Data\$Tabs_day_Final)) #11306
sum(is.infinite(Data\$Tabs_day_Final)) #69812
sum(is.nan(Data\$Tabs_day_Final)) #17</pre>

Change all the missing and infinity data to zero
Data\$Tabs_day_Final[Data\$Tabs_day_Final == "Inf" | Data\$Tabs_day_Final == "NaN" | is.na(Data\$Tabs_day_Final)] <- '0' sum(is.na(Data\$Tabs_day_Final)) #0 sum(is.infinite(Data\$Tabs_day_Final)) #0 sum(is.nan(Data\$Tabs_day_Final)) #0

DETERMINE DURATION

3 ways of calculating duration

- # "duration" given in CPRD (+imputed from directions for use variable)
- # "Duration" calculated: quantity prescribed / number of tabs per day
- # "Duration_dates" calculated: difference between 2 consecutive prescription dates

Merge ALL THREE duration variables
Priority 1: duration (Given in CPRD + imputed)

Priority 2: Duration
Data\$Duration_Calc <- Data\$quantity_prescribed/Data\$Tabs_day
Merge first 2
Data\$Dur <- Data\$duration
Data\$Dur <- ifelse ((Data\$duration == "0"), Data\$Duration_Calc, Data\$duration)</pre>

Priority 3: Duration_dates (Merge 1, 2 & 3)
Data\$Dur_Dur <- Data\$Dur
Data\$Dur_Dur <- ifelse ((Data\$Dur == "0" | Data\$Dur == "Inf" | Data\$Dur == "NaN" |
is.na(Data\$Dur)), Data\$Duration_Dates, Data\$Dur)</pre>

MEDIAN DURATION # Calculate total duration for each patient by disease CumDuration <- aggregate(Data\$Dur_Dur, by=list(Data\$patid, Data\$disease), sum, na.rm=TRUE) summary (Data\$Dur_Dur)

For all missing duration - impute with median of 28 days
Data\$Duration_Final <- Data\$Dur_Dur
Data\$Duration_Final[Data\$Duration_Final == "0" | is.na(Data\$Duration_Final)] <- '28'</pre>

Replace those with > 90 days duration with the max of 90 days Data\$Duration_Final [Data\$Duration_Final > 90] <- 90</pre>

ADDRESSING MISSING DATA FOR NUMBER OF TABS PER DAY

Data\$Tabs_day_New <- Data\$quantity_prescribed/Data\$Duration_Final

Replace all zero with this new tabs/day
Data\$Tabs_day_Last <- Data\$Tabs_day_Final
Data\$Tabs_day_Last <- ifelse ((Data\$Tabs_day_Final == "0" |
is.na(Data\$Tabs_day_Final)), Data\$Tabs_day_New, Data\$Tabs_day_Final)</pre>

Assume number of tabs=1 if not stated Data\$Tabs_day_Assumption <- Data\$Tabs_day_Last Data\$Tabs_day_Assumption[Data\$Tabs_day_Assumption == "0" | is.na(Data\$Tabs_day_Assumption)] <- '1'</pre>

DETERMINE DAILY DOSE

Calculate total daily dose by patient (mg) - based on max daily dose cut off points # Formula: Tabs_day_Final * PED

Data\$DailyDose <- Data\$Tabs_day_Assumption*Data\$predn_equiv

Replace those with > 120mg/day with 120mg
Data\$DailyDose [Data\$DailyDose > 120] <- 120</pre>

GC STARTING DOSE, DAILY DOSE, CUM DOSE & DURATION

PMR
Calculate cumulative dose by patient
PMR\$CumDose <- PMR\$DailyDose*PMR\$Duration_Final
Sum_PMR <- PMR %>%
group_by(patid) %>%
summarise(sum=sum(CumDose))

Calculate cumulative duration by patient
Dur_PMR <- PMR %>%
group_by(patid) %>%
summarise(sum=sum(Duration_Final))

Calculate average daily dose by patient
DailyDose <- merge (Sum_PMR, Dur_PMR, by="patid")
DailyDose \$DD <- DD\$CumDose/DD\$CumDuration</pre>

Calculate first dose Earliest <- setDT(PMR)[order(prescription_date), head(.SD, 1L), by="patid"]</pre>

Note (1): Repeat process for PMR patients with DM & no DM Note (2): Repeat process for GCA & GCA/PMR patients, then for DM & no DM subgroups

Appendix 4.4. Preparing Loops to Calculate Daily Doses

LOOP - DAILY DOSE

PMR only
Use aggregate to calculate the total dose taken during follow up
PMR2\$total.px.dose <- with (PMR2, Duration*DailyDose)
num.pts <- length(unique(PMR2\$patid)))
temp <- rep(NA, num.pts)
pt.df <- as.data.frame(temp)
pt.df\$pt.aggregated.dose <- aggregate(PMR2\$total.px.dose, by = list(PMR2\$patid),
sum)</pre>

Look at total dose summed from monthly doses # Manipulate dates (Using Dx dates) PMR2\$dx.date <- as.Date(PMR2\$first_diagnosis) PMR2\$dx.day <- as.numeric(PMR2\$dx.date) # Default start date is 01-01-1970</pre>

Manipulate dates (Using Rx dates)
PMR2\$px.date <- as.Date(PMR2\$prescription_date)
PMR2\$px.day <- as.numeric(PMR2\$px.date) # Default start date is 01-01-1970
last.px.taken <- max(PMR2\$px.day + PMR2\$Duration) # Take the last Rx and plus the
duration of that Rx (75 days)
num.days <- last.px.taken + 720 # Add dummy "0" for all patients who had less than 2
years of FU/Rx</pre>

```
# Manipulate dates (Using DM_Dx dates)
PMR2$dm.date <- as.Date(PMR2$DM_Dx)
PMR2$dm.day <- as.numeric(PMR2$dm.date) # Default start date is 01-01-1970</pre>
```

Create empty array (by repeating 0 for the number of patients x number of days)
num.pts <- length(unique(PMR2\$patid))
some.zeroes <- rep(0, num.pts*num.days)
dose.matrix <- matrix(some.zeroes, nrow=num.pts, ncol=num.days, byrow=TRUE)
dim(dose.matrix) # 19204, 18247</pre>

```
# Create a vector with patid within
pts <- unique(PMR2$patid)</pre>
```

```
# Use loops to fill dose matrix
num.PMR <- length(PMR2$prescription_date)
for(i in 1:num.PMR) {
  for(j in 1:PMR2$Duration[i]) {
    dose.matrix[which(pts==PMR2$patid[i]), PMR2$px.day[i]+j-1] <-
   dose.matrix[which(pts==PMR2$patid[i]),
  PMR2$px.day[i]+j-1] + PMR2$DailyDose[i]
  }
```

Change the matrix to data frame format dose.df <- data.frame(dose.matrix)</pre>

}

Create another file for diagnosis of PMR / PMR
diags <- unique(data.frame(cbind(PMR2\$patid, PMR2\$dx.day)))</pre>

Adjust to start on diagnosis day adjusted.mx <- matrix (rep(0, num.pts*720), nrow=num.pts, ncol=720, byrow=T) dim(adjusted.mx)

```
# Populate matrix with diagnosis day
for(i in 1:num.pts) {
    for (j in diags$dx.day[i]:(diags$dx.day[i]+719)) {
        adjusted.mx[i,j-diags$dx.day[i]+1] <- dose.matrix[i, j]
    }
    adj.df <- data.frame(adjusted.mx)</pre>
```

```
# Aggregate daily dose by 30 days
PMR_30days<- do.call(cbind, by(t(adj.df), (seq(ncol(adj.df)) - 1) %/% 30, FUN =
colSums))
dim (PMR_30days)
PMR_30days <- data.frame(PMR_30days)</pre>
```

```
# Rename months
```

```
names(PMR_30days) <- c("Month00", "Month01", "Month02", "Month03", "Month04",
"Month05", "Month06", "Month07", "Month08", "Month09",
"Month10", "Month11", "Month12", "Month13", "Month14",
"Month15", "Month16", "Month17", "Month18", "Month19",
"Month20", "Month21", "Month22", "Month23")
```

Calculate total dose over 2 years

pt.df\$two.years <- with(PMR_30days, Month00 + Month01 + Month02 + Month03

- + Month04 + Month05 + Month06 + Month07 + Month08 + Month09
 - + Month10 + Month11 + Month12 + Month13 + Month14 + Month15
 - + Month16 + Month17 + Month18 + Month19 + Month20 + Month21
 - + Month22 + Month23)

Incorporate patid
PMR_30days\$patid <- diags\$patid</pre>

Create another file for diagnosis of DM
DM <- unique(data.frame(cbind(PMR2\$patid, PMR2\$dm.day, PMR2\$DM)))</pre>

Merge PMR/PMR diagnosis & DM files DMDx_Dx <- merge(diags, DM, by=c("patid"))</pre>

Merge all files Combo <- merge(DMDx_Dx, PMR_30days, by=c("patid"), all.y = TRUE) names (Combo)

Note: Repeat process for GCA & PMR/GCA cohort

171

Appendix 4.5. Format Data For Survival Analysis

PREPARE FORMAT FOR SURVIVAL ANALYSIS

PMR
Change from wide to long format
PMR_Long <- reshape(PMR, direction="long", varying=1:24, idvar='patid',
timevar="month", v.names="totaldose")</pre>

Add start & stop (beginning & end of interval) variables PMR_Long\$Stop <- PMR_Long\$month PMR_Long\$Start <- PMR_Long\$Stop - 1

Add "Event" column
Diff_30 = DM date - Dx date, then divided by 30
PMR_Long\$Event <- "0"
PMR_Long\$Event [PMR_Long\$DM=="1" & (PMR_Long\$Diff_30 >= PMR_Long\$Start
& PMR_Long\$Diff_30 <= PMR_Long\$Stop)] <- "1"</pre>

Sort by patid and month
PMR_Long <- orderBy (~ + patid + month, data = PMR_Long)</pre>

For each patient - determine the "max month" / end of FU
PMR_Long <- PMR_Long %>%
group_by(patid, DM) %>%
filter(cumsum(lag(Event == "1", default = FALSE)) < 1)</pre>

Check WCE input (Data are in the right format for WCE estimation)
checkWCE(WCE_PMR_Final, id = "patid", event = "Event", start = "Start", stop =
"Stop", expos = "totaldose")

Note: Repeat process for GCA & PMR/GCA cohort

Appendix 4.6. Kaplan-Meier Survival Analysis

KAPLAN-MEIER ANALYSIS

Stratify by disease KM[grep ("DM", KM\$DM), "Diabetes"] <- 1 KM[grep ("NoDM", KM\$DM), "Diabetes"] <- 0

fit <- survfit(Surv(KM\$FU_Years, KM\$Diabetes) ~ KM\$Disease, data = KM) ggsurvplot(fit, data = KM) # Basic plot

ggsurvplot(fit, data = KM, risk.table = TRUE, pval = TRUE, conf.int = FALSE, xlim = c(0,2), xlab = "Follow-up Duration (years), ylim = c(0.8,1), pval.coord = c(0.8, 0.9), break.time.by = 0.5, ggtheme = theme_light(), legend.title = "Disease", legend.labs = c("PMR", "GCA and PMR/GCA"), risk.table.y.text.col = T, risk.table.y.text = FALSE)

Note: Repeat for gender & age (by tertiles, quartiles & quintiles)

Appendix 4.7. Extended Time-Varying Cox for Cumulative Dose

TIME VARYING COX - CUMULATIVE DOSE

PMR

Mod1 <- transformBy("patid", Cum.dose = cumsum(totaldose), data = PMR) cum.fit <- coxph(Surv(Start, Stop, DM1) ~ FamilyDM + NonOralGC + I(Cum.dose/10000) + I((Cum.dose)*(Cum.dose)/100000000), data = Mod1) cum.fit cox.zph(cum.fit) summary (cum.fit)

Note: Repeat for each individual covariate; as well as the GCA & PMR/GCA cohort

Appendix 4.8. Rolling Cumulative Dose

ROLLING CUMULATIVE DOSE ANALYSIS

```
## PMR ##
# Month 1
patients <- unique(PMR$patid)
rolling1 <- vector()
for(i in patients) {
    doses4i <- subset(PMR, patid==i)
    n <- length(doses4i$month)
    temp <- cumsum(doses4i$totaldose)
    if (n>1) {
      temp[1:n] <- rollapply(doses4i$totaldose,1,sum)
    } # end of i loop
    rolling1 <- c(rolling1,temp)
} # end of j loop</pre>
```

PMR\$rolling1 <- rolling1

Note: Repeat for months 2-24

```
# Time-varying Cox for rolling cumulative data
# Rolling - Month 1
cum.fit <- coxph(Surv(Start, Stop, DM.status) ~ FamilyDM + NonOralGC +
I(rolling1/1000) + I(rolling1*rolling1/1000000), data = PMR_Rolling)
cum.fit
cox.zph(cum.fit)
summary (cum.fit)
```

Note: Repeat for months 2-24

Concordance for month 2 is the highest # Calculation of HR in Excel using coefficient for month 2 with (PMR_Rolling, tapply (totaldose,month,summary))

```
Low Dose (month 1) = (LDose*coeff) + (LDose*Coeff(quadratic)* Coeff(quadratic))
High Dose (month 1) = (HDose*coeff) + (HDose*Coeff(quadratic)* Coeff(quadratic))
HR = exp (difference between high dose & low dose)
```

Note (1): Repeat for all doses from month 2-24 Note (2): Repeat process for GCA & PMR/GCA cohort

Appendix 4.9. Weighted Cumulative Dose

WCE ANALYSIS

Lowest BIC at month 10 # Compare doses over the most recent 10 months between patients on high dose vs. low dose with (PMR, tapply (totaldose, month, summary)) LQ10 <- c(0.00, 0.00, 0.00, 0.00, 3.12, 19.21, 36.69, 60.0, 98.15, 263.3) UQ10 <- c(210.0, 225.00, 250.0, 274.1, 300.00, 324.97, 367.50, 422.3, 500.00, 714.3) HR.WCE(wce.fit, UQ10, LQ10, allres = TRUE)

Note: Repeat process for GCA & PMR/GCA cohort